ABSTRACT

Title of Dissertation: A CRITICAL EVALUATION OF A SCHOOL

SYSTEM'S EFFORT TO DEVELOP AND

IMPLEMENT A "GROW YOUR OWN" PRINCIPAL

PREPARATION PROGRAM

Heath Morrison, Doctor of Philosophy, 2005

Dissertation directed by: Professor Carol Parham

Department of Education Policy and Leadership

Professor Emeritus James Dudley

Department of Education Policy and Leadership

This study is a formative evaluation of one school system's efforts to develop and implement a "grow your own" principal preparation program to address principal candidate quality and quantity. When over half of the school districts across the country have reported a shortage of qualified principal candidates and it is estimated that 40% of the nation's principals will retire in the near future, more school systems can be expected to attempt to implement such programs as a solution to this problem.

Six research questions were created for this study, and quantitative and qualitative methods were used to collect and analyze data. The research design was based on Daniel Stufflebeam's conceptual model for program evaluation, CIPP. Data were gathered to address the four components of this model—context, input, process, and product. A survey instrument wase- mailed to the 111 program participants from the school system's Aspiring Leaders Program and design team members. The final

sample yielded 95 responses—a return rate of 86%. Descriptive and inferential statistics were used. Data were collected and analyzed through personal interviews with system leaders, and focus group interviews were conducted with the design team and program participants.

The data provided insight into the various aspects of the school system's creation of a leadership program. Important elements and resources required in design and implementation included the ISLLC Standards. An emphasis was placed on making the program "hands-on" and practical in nature. The resource investment, both human and fiscal, was considerable. Based on survey results, the program was well received by participants across all cohorts and leadership skill areas. Program participants who have advanced to the principalship reported the training was excellent in assisting their preparation. These findings and conclusions are reported in Chapters IV and V.

This study was a formative evaluation of one school system's effort to prepare principals and therefore has limited generalizability. The results do raise significant issues with implications for policy and procedures. It is expected that this research will assist the designated school system's efforts to improve its principal preparation program, and provide insight for other school systems considering developing such programs.

A CRITICAL EVALUATION OF A SCHOOL SYSTEM'S EFFORT TO DEVELOP AND IMPLEMENT A "GROW YOUR OWN" PRINCIPAL PREPARATION PROGRAM

by

Heath Morrison

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park in partial fulfillment of the requirements for the degree of Doctor of Philosophy

2005

Advisory Committee:

Professor Carol Parham, Co-Chairperson/Advisor Professor Emeritus James Dudley, Co-Chairperson/Advisor Professor Gilbert R. Austin Professor George Marx Professor Thomas Weible © Copyright by

Heath Morrison 2005

DEDICATION

To Dr. James Dudley, whose generosity and mentorship have been immeasurable...

To my wonderful children, Samantha and Zachary, who are my continual inspiration ...

And to my wife, Jennifer Morrison, whose faith, support, and love helped make this dream a reality.

ACKNOWLEDGMENTS

My deepest appreciation to my advisor, Carol Parham, for agreeing to accept me as her first doctoral candidate. I am extremely grateful for her encouragement and excellent advice as my advisor and for her leadership as my chairperson.

I am also indebted to my other advisor and co-chairperson, Dr. Dudley for teaching me so much about the unselfishness of a caring educator.

To the other members of my committee—Dr. Gilbert Austin, Dr. George Marx, and Dr. Thomas Weible—I offer my sincere gratitude for providing feedback and support for my efforts throughout the entire process. I extend a special note of appreciation to Dr. Austin for assisting me with the data analysis of this study.

This dissertation could not have been completed without the assistance of so many individuals within my school system. I thank Jim Richmond, Ron Cunningham, and Dr. John Cox, who have all mentored me as an educational leader. I extend my sincerest gratitude to all of the individuals who took their valuable time to participate in the study, particularly my fellow principals. I was blessed to have the support of my school family in completing this work, especially my secretaries and vice principals. I am also blessed to have my students who continually reinforce the choice of education as my profession.

Finally, I want to acknowledge my family, who have never failed to give me the strength and persistence to complete this dissertation. I thank my parents, Howard and Barbara Morrison, for their sacrifices that have allowed me to endeavor to be a life-long learner. I thank my children, Samantha and Zachary, who always make me want to strive to be the best father and role model I can become. And I thank my loving wife, Jennifer, without whom this project could not have been completed. Thank you for believing in me and this dream.

TABLE OF CONTENTS

List of Tab	les
Chapter I	INTRODUCTION TO THE STUDY
	Importance of the Principal
	Issues of Quantity and Quality With the Principalship
	Principal Quantity Issues
	Principal Quality Issues
	Principal Preparation Concerns
	Concerns Related to Traditional Programs
	Concerns for Quality Intern Experiences
	Statement of the Problem
	One School System's Efforts to Create a "Grow Your Own"
	Principal Preparation Program
	Program Design
	Program Elements
	Research Problem
	Research Design
	Research Questions
	Conceptual Framework
	Definitions
	Limitations
	Delimitations
	Assumptions
	Significance
	Organization of Study
Chapter II	REVIEW OF THE LITERATURE
	The Evolving Role of the Principalship
	Forces of Change Impacting the Principalship
	Changes in the Principalship—Implications for Preparation
	Concerns With Traditional Principal Preparation Programs
	Trends in Effective Principal Preparation Programs
	Standards for Principals
	Leadership Training Programs
	Relationship of Literature Review to This Study
Chapter III	RESEARCH METHODOLOGY
	Restatement of Purpose
	Program Evaluation
	The CIPP Evaluation Model

	Application of CIPP Model to Study Research Questions	5
	Location of the Study	5
	Research Methods	5
	Data Collection Techniques	5
	Data Analysis	6
	Limitations	6
	Validity of Data	6
	Focus Group Interviews	6
	Summary	6
Chapter IV	FINDINGS	7
	Introduction	7
	Reorganization of the School System with an Emphasis on	7
	the Principalship	7
	A Critical Shortage of Qualified Principal Candidates to Meet Anticipated Needs	7
	Concerns Over the Changing Role of the Principal in an	
	Era of Accountability	8
	Concerns Over Traditional University Preparation of Principals	8
	Addressing New Licensure Requirements for Principals	8
	Addressing Minority Achievement and a Lack of Minority	
	Candidates for the Principalship	8
	Make the Program Unique to the School System	8
	Making the Program Hands-On and Practical	8
	Framing the Program Around the ISLLC Standards and	
	Identified System Themes of Leadership	8
	Making the Program Free of Charge, Inclusive, and	
	Available for College Credit	9
	Human Resources	g
	Financial Resources	ģ
	Program Sources and the Process of Designing the	
	ProgramProgram	ç
	Finalizing the Process and Initiating the Program	10
	Recruiting Participants	10
	Summary of Quantitative Data from the Survey Instrument	12
	Program Benefits—A Clear Understanding of the	12
	Principalship, Especially in the Area of Accountability	12
	A Greater Sense of Local Preparedness and Desire for	
	Becoming a Principal in the Designated School System	12
	The Establishment of a Strong Network of Colleagues	
	to Provide Support in New Positions	12
	Needed Program Improvements—A Greater Emphasis	
	on Time Management, Communication, and Data	
	Analysis	12

	The Need for a Stronger Partnership with the College Involved with the Program and Support for Advanced Graduate Level Work
	Completing All Phases of the Designated Program,
	Including the Project and Intern Components
	Separate Training for Aspiring Principals from Training
	for Other Leadership Positions
	Summary
	~,
Chapter V	SUMMARY, CONCLUSIONS, AND
-	RECOMMENDATIONS
	Introduction
	Summary of the Study
	Summary of Findings for the Research Questions
	Findings Supported by Review of Research
	Findings Supported by the Research Study
	Summary of Quantitative Survey Findings
	Summary of the Qualitative Data Findings
	Conclusions
	Conclusion #1
	Conclusion #2
	Conclusion #3
	Conclusion #4
	Conclusion #4
	Recommendations For Practice
	Recommendation #1
	Brief Rationale
	Recommendation #2
	Brief Rationale
	Recommendation #3
	Brief Rationale
	Recommendation #4
	Brief Rationale
	Recommendation #5
	Brief Rationale
	Recommendation #6
	Brief Rationale
	Recommendations for Further Research
	Recommendation #1
	Brief Rationale
	Recommendation #2
	Briaf Pationala

	Recommendation #3	154
	Brief Rationale	154
Perso	onal Reflection on This Study	154
Appendix A	The Survey	156
Appendix B	ISLLC Standards	179
Appendix C	Request to do Research and Interview Protocols	181
Appendix D	Second Request Letter	210
Appendix E	Leadership Grant	212
Appendix F	School System Leadership Themes and ISLLC Standards	216
References		218

LIST OF TABLES

1.	Research Questions and Alignment with the Ten CIPP Components Checklist	55
2.	Aspiring Principal Program Participants	56
3.	Data Analysis Chart	64
4.	Reorganization of School System Executive Staff	75
5.	Number of Principals Leaving the Position, 1990-2000	77
6.	Current Principal Tenure	78
7.	Number of Candidates Applying for Administrative Positions Before and After the Principal Preparation Program	79
8.	Minority Principals	84
9.	Variety of Local School Stakeholders Who Have Presented During the Leadership Program	86
10.	Participation of Principals in the Program, 2001	88
11.	Match of System Themes and the ISLLC Standards	90
12.	Composition of Leadership Advisory Committee	94
13.	Grant Application	97
14.	Financial Breakdown for Summer Institute	98
15.	Reasons That Motivated Respondents to Pursue the County Leadership Program	104
16.	Correlation Coefficients for Domains One, Two, and Three	108
17.	Number and Percentage of Respondents by Gender, Age, and Race	109
18.	One-Way Analysis of Variance of Differences Among Cohorts One, Two, and Three on Domains One, Two, and Three	110

19.	One-Way Analysis of Variance of Differences Among Cohorts One, Two, and Three on Summary Questions for Domains One, Two, and Three	111
20.	Independent t-Test of the Total Cohorts' Responses versus the Design Team's Responses to Domains One, Two, and Three	112
21.	Independent t-Test of the Total Cohorts' Responses versus the Design Team's Responses to Summary Questions	113
22.	Independent t-Test Comparing Males' and Females' Responses for Domains One, Two, and Three	114
23.	One-Way Analysis of Variance of Differences Based on Age for Domains One, Two, and Three	115
24.	Independent t-Test Comparing African Americans and Caucasians for Domains One, Two, and Three	116
25.	Independent -Test Comparing Elementary versus Middle and High School Respondents on Domains One, Two, and Three	117
26.	Independent t-Test Comparing Applicants and Non-Applicants for Principalships in Domains One, Two, and Three	118
27.	Independent t-Test Comparing Applicants and Non-Applicants for Other Leadership Positions in Domains One, Two, and Three.	119
28.	One-Way Analysis of Variance of Differences Based on Respondents' Thoughts About Applying for Future Leadership Positions in Domains One, Two, and Three	120
29.	Program Participants Who Have Advanced in Leadership Positions	124
30.	Gender and Race of Program Participants Who Have Advanced in Leadership	125

CHAPTER I

INTRODUCTION TO THE STUDY

Importance of the Principal

Good schools have good principals. This theme resonates throughout literature about effective schools. In a study on high-achieving schools, the authors concluded that the schools they visited "succeeded to a large degree because of the alert, consistent, resourceful, and sustaining energy of the principals, and that leaders such as these are essential ingredients of successful schools" (Educational Research Service, 2000, p.10).

The evolution of the principalship has mirrored the increased expectations on public schools. Since the 1950s, the principal's role has grown to include "financial manager, skillful negotiator, manager of human resources, source of legal knowledge, and human relations expert" (Hessel & Holloway, 2002, p.12). With the publication of *A Nation at Risk* (National Commission on Excellence in Education [NCEE], 1983), the public demanded an increased policy focus on improving schools and student achievement through systemic educational reform. While certainly not a study on the principalship, *A Nation at Risk* drew attention to the need to have effective leaders for school principals (NCEE, 1983).

Research on effective schools consistently stresses how absolutely vital the principal is to a school's ongoing success. Many studies (Barker, 1997; Edmond, 1979; Goodlad, 1955; J. Murphy, 2001) have claimed for decades that good principals are the most important key to school reform. At least one study suggests, "The principal holds the most crucial leadership position in a school district" (Dosdall,

2001, p.26). In a report entitled, "Exploring the Principal's Contribution to School Effectiveness: 1980-1995," Hallinger and Heck (1998) claim principals have a huge effect on school effectiveness and student achievement. They state, "Principals influence school performance by shaping school goals, direction, structure, and organizational and social networks" (p.6).

The National Association of Elementary School Principals (NAESP) has documented more than twenty nationwide statistical studies (ERS, 2000, p. 25) demonstrating the dramatic effect of the principal on school and student performance. The overwhelming conclusion of these reports was that active instructional leadership by principals is the best indication of higher student achievement (Council of Chief State School Officials [CCSO], 1996; Edmonds, 1979, 1982; Lashway, 2003).

At the secondary level, much of the research recognizes how difficult it is for one individual to provide instructional leadership to the entire school community (Lezotte, 1991; Sergiovanni, 1996). Increasingly, it is expected that principals broaden the concept of instructional leadership to create a team or community of leaders with teachers and staff.

More than anything else, excellent principals create a collective sense of responsibility for school improvement by not only promoting the belief that students can learn, but also by creating initiatives to ensure that all students do learn. As Richard Riley, former Secretary of the U.S. Department of Education (1984), stated, "The principalship is a position that is absolutely critical to educational change and improvement. A good principal can create a climate that can foster excellence in teaching and learning while an ineffective one can quickly thwart the progress of the most dedicated performers" (ERS, 2000, p.15).

Issues of Quantity and Quality With the Principalship

Principal Quantity Issues

Concern about a looming principal shortage has been well documented in recent professional literature (Barker, 1997; Donaldson, 2001; Jordan, 1994). The U.S. Department of Labor estimates that 40% of the United States' 93,200 principals are nearing retirement and that the nation's need for additional school leaders will "increase 10-15% through 2005 to accommodate the growing student population" (Tracy & Weaver, 2000, p.2). The same report indicates that over half of the nation's school districts are facing immediate principal shortages.

In 1997, the National Association of Elementary School Principals (NAESP) and the National Association of Secondary School Principals (NASSP) asked Educational Research Service (ERS) to survey superintendents nationwide about the possibility of a looming principal shortage. This survey, published in 1998, showed that in half of the United States' school districts, superintendents cited a shortage of qualified candidates for principalship vacancies. The study cited a huge wave of principal retirement shortages in all regions and at all three levels—elementary, middle, and high—as the major cause for the shortage, which is expected to worsen over the current decade (Lovely, 2004, p.1).

According to the same report, fewer principals were planning to remain in the position until retirement. NASSP reported a 50% turnover rate in principalships nationally during the 1990s and predicts another 60% turnover by 2010 (Whitaker, 2001, p.3). "The attrition rate stood at 48% for the decade from 1988 to 1998 and could reach as high as 60% for the next decade. We are already seeing a pattern of principals opting to retire at the earliest possible date" (Ferrandino, 2001, p.2).

Principal Quality Issues

Not only are the nation's applicants for principal positions decreasing, there is growing concern that the quality of applicants is also declining (Bell, 2001; ERS, 1998; ERS, 2000; IEL, 2000). The Institute for Educational Leadership (2000) suggests that aspiring principals' backgrounds and experiences are grounded in the traditional managerial role of the principal rather than as an instructional leader who is accountable for student learning. Many superintendents and personnel experts have commented on the candidates' lack of experience for principal positions, as well as "a lack of knowledge and skills in the areas of instruction and assessment" (Whitaker, 2001, p.6). In a survey conducted in 2000, 98% of California superintendents reported shortages of qualified principals (Orozco, 2001, p.2). Charles Stein, a Director of Human Resources in New Jersey, stated, "A few years ago, a principal vacancy meant over 100 good candidates. Now we might have twenty with half of them recycled, meaning they have previously been rejected by other districts" (as cited in Brockett, 1999, p.2).

Because of the crucial role principals play in educational reform, there is a growing concern that schools led by underqualified or inexperienced principals will not be able to implement the many national and state improvement programs currently mandated for public schools. "No school that I know reaches its potential without a principal who knows how to provoke the best in others, who has a deep teaching sense of the mood and culture of that particular community at each particular moment in time, and who has the courage, integrity, and skill to act" (Anderson, 1995, p.422).

In a survey of school superintendents, there was a common concern that applicants for principalships are simply not as experienced in teaching and/or in administration as in previous years (Kerrins, 2001, p.3). Half of the 400 superintendents interviewed reported a shortage of qualified candidates to fill their

principal vacancies. The survey found "large concerns with principal candidates showing a lack of competency in the areas of planning, organizing, and evaluating staff development consistent with curriculum and instructional needs" (Kerrins, 2001, p.4). Gene Bottoms and Kathy O'Neill (2001) of the Southern Regional Education Board wrote:

Personnel shortages in education never last long. We can be sure that school boards will find someone to fill every principal vacancy. The real "emergency" we face is the prospect that unless we identify and train school leaders who have a deep knowledge about how to improve the instructional functions of a school, we will do little to resolve spotty leadership, low advancing schools, and under-served students.

(Bottoms & O'Neill, 2001, p. 7)

School systems cannot be satisfied to simply "fill principal slots." They must look to institutions and programs designed to prepare future principals to ensure an adequate supply of quality candidates.

Principal Preparation Concerns

Concerns Related to Traditional Programs

As school systems prepare to face a shortage of qualified applicants for their administrative positions, the traditional method of preparing principals has come under increased scrutiny. Educators and policy makers have articulated growing concerns over university programs that stress leadership theory, but generally offer limited opportunities to apply knowledge and theory into practice (Anderson, 1991; Lauder, 2000; Smith & Piele, 1997). Common criticisms range from classes being taught by professors who have not entered a school in years to candidates studying "theory and policy in classes that were poorly related to one another from a curriculum that was

fractured and rarely included reflective, hands-on experience that administrators themselves assert is a fundamental component of principal programs" (National Association of State Boards of Education (NASBE), 1999, p.10).

Concerns for Quality Intern Experiences

School systems may have been comfortable with the idea that the role of assistant principal would adequately provide the practical experiences university-based principal preparation programs lacked. However, there is increasing evidence to suggest that the role of assistant principal fails to provide the training and background needed for the instructional leadership experience of today's principals. Hartzell, Williams and Nelson (1995) stated, "The nature of the assistant principalship and the skills required to be successful as an assistant principal are oriented much more toward management than toward leadership, a condition that does not promote the development of visionary leadership in its occupants" (Hartzell, et al. as cited in Batenhorst, 2002, p.1). School systems requiring a large number of qualified candidates to become instructional leaders may be coming to the conclusion that "traditional preparation programs, teaching experience, coursework, practicum, and even service as a vice principal are not successful in adequately preparing aspiring administrators for the principalship" (Cotton, 2003, p. 15).

Statement of the Problem

In an effort to recruit candidates and prepare them for the challenges of the principalship, many school districts around the country are creating leadership programs specifically for aspiring principals. Rather than relying solely on colleges and universities to provide future principals, school systems are coming to the realization they will have to supplement such programs with training of their own.

These principal programs provide an opportunity to not only better prepare individuals

for the principalship, but also entice a larger number of candidates to the recruitment pool. Even though 47% of the nation's teachers have master's degrees, a number in administration, many choose not to consider principalships (Groff, 2001, p.6). The challenge is to entice those who are qualified to explore leadership positions, especially principalships. One way to entice them is through high-quality leadership preparation programs.

Many school districts have responded to concerns about principal candidate quantity and quality by developing leadership within their own organizations and instituting "grow your own" principal preparation programs. Gerald Tirozzi (2001), Executive Director of the NASSP, in a recent edition of *Phi Delta Kappan* stated, "Unlike the common practice of corporations and the military where there are systematic and continuing initiatives to grow and develop a management cadre that can take on greater and greater responsibilities and succeed at each step along the way, education makes no such careful investment of resources in its future leaders" (p. 4). Tirozzi asks the question: "Can school systems design individualized programs where the content and implementation reflect the needs of both the system and program participants?" (p.4).

In its work, "The Principal, Keystone of a High-Achieving School," Educational Research Service (2000) cites systems such as Jefferson County Public Schools District (Kentucky), Huron Valley School District (Michigan), and Fort Wayne School (Indiana) as examples of "locally designed 'grow your own' principal candidate programs" (p.50). "Grow your own" programs are often characterized as "hands-on and experiential" and provide a school system with a great way to gather information about future principal candidates (ERS, 2000, p.53).

"Grow your own" principal preparation programs are not designed to supplant existing university programs. Instead, they are a means for a school system to provide

additional training and support for aspiring principals. In many instances, school systems have elected to work with institutions of higher education in the formulation and implementation of programs to "close the gap." In some cases, they have decided to combine with other school districts to pool resources (Douegh, 1992, p.13). In the state in which this study was conducted, twelve of the seventeen counties that have programs are working in collaboration with one or more institutions of higher learning and a number of the smaller districts have combined resources to develop a unified program (State Survey, 2004, p. 1).

While there appears to be a growing number of "grow your own" principal preparation programs across the country, current research about them is limited. Few studies have been done to examine if they have avoided some of the noted concerns of university-based programs, or if they are an effective means of addressing the growing concerns of a shortage of qualified candidates for future principalships.

One School System's Efforts To Create A "Grow Your Own" Principal Preparation Program

In 1999, the superintendent of the school system being studied convened a committee of various stakeholders including central office personnel, principals, vice principals, and teachers to address leadership preparation issues, including the critical shortage of qualified principal applicants. A large turnover in the system's principals (13 of 31 in the previous four years) combined with a limited pool of applicants was a growing source of concern for the district and superintendent.

After a lengthy process, the Aspiring Principal Preparation Program was created, and 99 potential leaders enrolled over a three-year period, many of whom have completed the first three program phases. The stated goal and main objective of this school district's principal preparation program is to produce a cadre of potential

principals who are better prepared for the challenges of the position than the current supply of principal candidates. Having gone through the program, participants should be better prepared to face the issues and challenges of a principalship in a time when schools and school systems are being challenged as never before to educate every child to the fullest extent of his/her ability, regardless of ethnicity, race, social economic status, and/or innate abilities. While the target audience for the program is current vice principals, administrative assistants, and teachers in educational administrative courses, it is open to anyone with interest in the principalship.

Program Design

Program Elements

Based on a review of current research about principal preparation programs and available standards, it was decided that the Interstate School Leaders Licensure Consortium (ISLLC) standards should serve as a foundation upon which to build the program. The program structure was to utilize current best practices and research on leadership and the principalship while applying a practical, "hands-on" approach to help program participants decide if the principalship was a realistic and personally desirable goal.

The Aspiring Principal Program in the school system studied had five phases in its initial design:

A five-day summer institute required for individuals who wanted to
participate in the larger program. The institute objective aligned with
the ISSLC standards and had a focus on vision, accountability,
instructional leadership, political decision-making, and legal issues in
education.

- 2. Seminars throughout the year where participants had to attend six out of the nine seminars to complete phase two of the program. The seminars were self-selected. A participant could choose to attend more than six. Participants enrolled for credit in a university associated with the program were required to attend all sessions.
- Projects to benefit the school system that were self-selected by participants, but had prior approval. A list of suggested projects was developed.
- 4. A mock interview accompanied with a professional portfolio to provide all participants the opportunity for a diagnostic interview experience.
- 5. An internship program designed to benefit current vice principals as well as aspiring principals.

Key central office administrators were used as keynote speakers. University professors from around the state as well as staff from the State Department of Education were utilized as consultants. Current principals were invited to collaborate and discuss issues of the program through presentations and question-and-answer sessions. The program was designed in collaboration with local colleges, allowing for participants to receive college credits toward a master's in educational administration and/or a certificate in advanced studies.

Research Problem

The research focus of this study is the development of a critical evaluation of one school system's efforts to develop and implement a "grow your own" program that will provide a continuing supply of highly qualified candidates to fill anticipated needs. The study serves two purposes:

- To determine the extent to which this system's efforts to design and implement a program that supplements university preparation have succeeded in meeting program goals (product), efficient use of system resources (input), and appropriate selection of program activities (process). The context in which this program was developed was also examined.
- 2. To determine what can be learned from these efforts that might be beneficial to this system as it attempts to improve the program, and what can be learned that might prove useful to other systems coping with the need to address principal shortages.

Research Design

The research design for this study is a formative evaluation of the aspiring leadership program in one school system in a mid-Atlantic state. The study is formative rather than summative as the data analysis results will be used to guide decision-making relative to improving the program. The research for this study begins with an extensive literature review of the most current information related to principal preparation programs, the changing role of the principal, and program evaluation tools used in a model created by Daniel Stufflebeam.

Both quantitative and qualitative methods were used for this study. For this effort, a comprehensive survey instrument was e-mailed to each of the 99 participants from the first three cohort groups of the Aspiring Leadership Program as well as the 12 members of the original design team. Focus group interviews with members of the design team who provided suggestions for creation of the program and often acted as facilitators were conducted. Also, focus group interviews were conducted with

program participants as well as with individuals who have attained principalships in the designated county.

Additional research was conducted in the school district. Research efforts included site visitations, comprehensive interviews with program administrators, participants, and presenters. Program documents and internal evaluation efforts will also be examined. Program resources, local as well as state and national, will be assessed as to source, purpose, and utilization.

Research Questions

Six research questions were framed in terms of the evaluation format being employed. The questions were stated so as to examine initial efforts to conceptualize the program as well as on-going efforts to make program adjustments based on participant and program designer feedback. In this sense, the evaluation is formative in nature and is designed to further inform the program design committee of areas where further changes may be warranted. The six questions are as follows:

- 1. What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program?
- 2. Which program elements were created specifically to address those forces impacting the principals?
- 3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program?
- 4. What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants?

- 5. What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role?
- 6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement?

Conceptual Framework

The conceptual framework for this study derives, in part, from scholarly works examining the importance of the evolving role of the principal and the history of principal preparation programs. It also derives from an examination of the limited, but growing, research on "grow your own" principal preparation programs.

As the examination of one such program is the focus of this study, it is necessary to choose an appropriate conceptual framework on which to base this study. Due to the fact that the program selected for study is still in a state of development, the researcher elected to utilize Daniel Stufflebeam's Content, Input, Process, and Product Model (CIPP) both to assess current program effects and to guide decision-making for future steps. According to Stufflebeam (2002), change is ongoing, and programs and processes employed to cope with change must constantly adapt to keep pace with ongoing developments (p.14).

Stufflebeam describes the CIPP Evaluation Model as "a comprehensive framework for guiding evaluation of programs, projects, personnel, products, institutions, and systems" (Stufflebeam, 2002, p.5). While the CIPP Evaluation Model was originally created to provide information in a systematic, proactive way for

decision-making, it is appealing because it has been proven to be beneficial in assisting educators after the fact to account for their decisions and actions (Stufflebeam, 1971a, p.1). Use of the CIPP Model allowed the researcher to determine the degree to which the designated county's Aspiring Leaders Program meets its originally stated objectives as well as in ascertaining its overall effectiveness, sustainability, and perhaps, transportability.

The research design employed both quantitative and qualitative methodologies.

The research design is based on Daniel Stufflebeam's conceptual model for program evaluation. Stufflebeam advocates a comprehensive evaluation model that examines four dimensions of program development and implementation:

- Context The context component of the CIPP model deals with the issues and forces of change which prompted the school system to consider the need for an aspiring principal leadership program.
- Input The focus on input variables examines the central idea and ideals that guided developers as well as the accounting of the resources expended by the school system to create and implement the program.
- 3. Process The process component of the CIPP model seeks to answer questions related to programs and processes as well as the various efforts related to program revision over the first three years.
- 4. Product An examination of the product segment of the CIPP model, will allow the researcher to assess the impact of the program on participant knowledge and skills related to the principalship as viewed both by members of the program design team and the program participants.

Definitions

The following terms are defined for the purpose of the study:

- 1. <u>Principal</u>: The chief operating officer of a school charged with instructional and managerial responsibilities.
- 2. <u>"Grow Your Own" Principal Preparation Programs:</u> Programs established and operated by local school systems to supplement and enhance the preparation provided by colleges and universities. While these programs may be instituted in partnership with a local university, they are conducted by the school district.
- 3. <u>Stufflebeam CIPP Evaluation Model:</u> An evaluation model that focuses on the core ideas of content, input, process, and product. Stufflebeam defines his model as being either formative (guiding program improvement), or substantive (evaluating program effectiveness), or both.

Limitations

- The study is limited to one school system's efforts to develop and implement a leadership development program for principal candidates.
 It is designed to meet the needs of a single school system and may not be suitable for generalization to school systems whose needs vary from the ones for which the program was designed.
- 2. The number of individuals who have moved through all phases of the program and been appointed to the principalship is quite small, and the length of time they have served as principals is also limited. Therefore, it is somewhat early to pronounce these efforts as successful or unsuccessful based on performance evaluations.
- 3. It should also be noted that the researcher was part of the initial design team and has been involved with the program in its implementation

phases. Therefore, there is a concern for the potential of "researcher bias." Utilizing multiple data sources and methods of collection should help to limit such bias.

Delimitations

- 1. The evaluation design is purposefully delimited to be of a formative nature as opposed to a summative one. This design is based on the need to accumulate data that will permit redesign and implementation changes as the program evolves into a more established format.
- 2. No effort was made to screen program candidates. The design team felt this initial effort should provide participant self-selection. Selfselection has provided for a much wider range of talent than would have occurred had specific criteria for program admission been applied.

Assumptions

For the purpose of this dissertation, the researcher has noted the following assumptions: Program design team members will desire an accurate assessment of the conceptual plan and of its implementation and will respond readily about its perceived strengths and weaknesses. Program participants will be candid about their reactions to the program. At the same time, many of the participants may desire to seek administrative positions in the school system being studied and may have some reluctance to identify negative aspects of the program. Care will be taken to guard the confidentiality of all data so as to protect interviewees and survey respondents in an effort to reduce this threat.

Significance

The significance of this study is to further the research on a local school system's attempt to create its own principal preparation program to supplement university programs. At this time, there is limited research about these types of programs, so this study should provide insight into the challenges of creating and implementing a program as well as potentially adding to the available research about them.

This study seeks to identify how a school system planned, developed, and implemented a principal preparation program. Using the CIPP model to evaluate this school system's efforts to create a "grow your own" principal preparation program has provided feedback, suggestions, and recommendations to the school system for program improvement.

Because of the critical role principals play in the success of a school, a potential national principal shortage is a concern to all school systems. Since creating principal preparation programs appears to be a solution to this potential crisis that many school districts are employing, examining one school system's experiences with such an endeavor may provide useful information to other programs. While this school system's program has not been identified as a model "grow your own" principal preparation program, documenting issues of its design, implementation, and initial effectiveness could assist other school systems in the design and implementation of their own programs.

Organization of Study

This study of one school system's efforts to design and implement a principal preparation program consists of five chapters. Chapter I presents an overview, which addresses the importance of this study as well as the conceptual framework for the program and the design for a formative evaluation of it.

Chapter II reviews the changing role of the principal in an era of high stakes accountability and the causes and potential effects of the growing principal shortage, and examines concerns and issues surrounding current principal preparation programs and efforts to improve them.

Chapter III restates the problem being studied in light of what the literature review revealed about principal preparation programs and examines Stufflebeam's CIPP model. It focuses on the population to be studied while providing support for limiting the study to one school system. It contains a clear identification of the sources of information utilized (questionnaires, surveys, interviews, etc.) as well as a detailed section on data collection and analysis procedures.

Chapter IV presents the findings of the study. Included is a restatement of the problem and research questions along with a summary of the data collection. A focused presentation of findings organized around the research question and subquestions is provided.

Chapter V discusses conclusions reached from the findings in Chapter IV.

Implications for changes in current practices or policy are suggested, along with recommendations for additional research in the area of principal preparation programs.

CHAPTER II

REVIEW OF THE LITERATURE

At a time when school systems across the country are experiencing potential administrative applicant quality and quantity issues, a number have chosen to become directly involved in the identification and preparation of their future principals. The purpose of this study is to examine how one school system chose to deal with this issue by creating a "grow your own" principal preparation program. This study is a formative evaluation that utilizes Stufflebeam's CIPP model in an effort to assess the program in its initial stages of development and seek information to guide program improvement.

This review of research is designed to investigate the literature related to administrative preparation programs with special focus on those that reflect "grow your own" principal preparation programs by school systems. The growing concerns with the traditional university model of principal preparation will be explored and several recent aspiring principal programs that have received national recognition will be highlighted. Indicators of reform, such as the ISLLC standards, will be addressed in relation to their use in current principal preparation programs.

The Evolving Role of the Principalship

A common theme from the literature review was not only how much the role of the principal has evolved over time, but how it has changed as society has changed (Glanz, 1998; Hessel & Holloway, 2002). The principalship has "been influenced and shaped by a variety of historical forces. The history of school leadership is clearly a

history of the interaction of broad social and intellectual moments within American society" (Hessel & Holloway, 2002, p.11).

The early roots of the principal emerged in the late 1800s as principal teachers were created to attend to school improvement functions as attendance and school cleanliness (Glanz, 1998). This role changed toward the late 1800s as a "teacher of teachers," assessing educators in the art of teaching and attending to other management issues (Glanz, 1998). Glanz notes another shift in the principal at the turn of the last century during the "Era of Economy and Efficiency" as the position was patterned after "the administrator of business and industrial management view of school organization" (Glanz, 1998, p.41). As society became increasingly bureaucratized, so did the role of principal, taking on an increased level of administrative responsibilities.

One of the greatest periods of change in the principalship came in the era from the 1950s to the end of the 1970s. As the United States entered into a Cold War with the Soviet Union, traveled into space, engaged in a very controversial war in Vietnam, and faced turbulent times at home regarding race relations, the principalship continued to change. In her work, "Integration of the ISLLC Leadership Standards into Rhode Island Principal Preparation Programs," Lisa Harpin (2003) documents researcher Phillip Hallinger's views of the changing role of the principal:

During the 1950s, the principal was viewed as an administrator who managed the school. With the urbanization of the schools in the 1960s, the principal's role changed to that of bureaucrat forced to carry out the policies instituted by government agencies. When federal efforts focused on curriculum during the 1960s and early 1970s, the principal became known as a "change agent' and responsibilities increased (p.36).

With the publication of *A Nation at Risk* in 1983, the role of the principal changed again as reaction to this publication spurred renewed interest in public education from Congress, state legislatures, and local school boards. Principals were now responsible "for creating the conditions that would support increased student achievement" (Harpin, 2003, p.36). Hessel and Holloway (2002) define this period of the 1980s as "The Age of Reform," where the focus of the principalship was to provide instructional leadership. However, while policy makers cited the need for principals to embrace this new focus on instructional leadership, the position still required principals to be a "financial manager, skillful negotiator, manager of human resources, source of legal knowledge, and human relations expert" (p.12).

The 1990s to the present have been labeled as "The Standards Movement" or "The Age of Accountability." External stakeholders such as parents, business leaders, politicians, and community members began to impact school reform. The ISLLC Standards, adopted in 1986, are examples of administrative standards created to define and improve the skills of school leaders. Principals were now asked to lead schools that were judged by the results students achieved. Ron Thomas (2004), in his work, "Personalizing Schools in the Age of Accountability," noted several implications of these shifts for educational leaders:

- 1. Schols are no longer judged by the processes educators engage in but by the results the students achieve.
- 2. Schools are no longer recognized for universal access to education but for universal proficiency in learning (p.1).

As the public expects more from schools and higher accountability for results, the principal has become more important than ever. Several studies have reported that principal quality is the best indicator of high student achievement (Lashway, 2003;

Richardson, 1989). In a report entitled, "Effective Schools Research: A Guide to School Improvement," effective schools at all levels had principals who:

- were assertive in their instructional role,
- were goal- and task-oriented,
- were well-organized and demonstrated skill in delegating responsibility to others.
- conveyed high expectations for students and staff,
- had policies that were well-defined and well-communicated,
- made frequent classroom visits,
- were highly visible and accessible to students and staff,
- provided strong support to the teaching staff, and
- were adept at parent and community relations. (ERS, 2000, p.6)

Forces of Change Impacting The Principalship

Principals lead schools and schools do not exist in a static environment. Over the past several decades, a number of forces have risen to change the face of public education and these forces have had a dramatic effect on the principal's role. These forces have not only created a need for a greater number of principals, but also a pool of qualified candidates who can meet the needs of a rapidly changing school population. The number of school-aged children has swelled dramatically, increasing beyond the ability of school systems to build new schools and keep class sizes down. It is estimated that 6,000 new schools must be built by 2006 to handle a 10% increase in K-12 students from 1998 (U.S. Dept. of Education, 2001, p.99). This increased enrollment has been accompanied by a growing percentage of ethnic and linguistic student diversity, presenting schools with new challenges. Not only will Hispanics and Asians account for 61% of the nation's population growth by 2025, but also a

significant portion of these students will enter schools speaking a language other than English (Tirozzi, 2001, p.4). The number of students with special needs has also risen, as has the level of support they require to be educated. Parents and community members have pushed for new laws and policies to protect the interests of these students to ensure they are being treated equitably in the nation's schools.

Principals have had to lead schools at a time of fluctuating budgets, growing school violence, changes in government policy and law, and a continued focus on standards and accountability. Paul Houston, Executive Director of the American Association of School Administrators, states, "There are no solutions envisioned by a dissatisfied public, but there are demands placed on school leaders to fix society's ills. The context then becomes a public who insists on quick, neat, simple solutions for problems that extend far beyond the scope of any one person, even the principal" (Houston, cited in Yerkes, 1998, p.12).

Until recently, the standards movement was focused on students and teachers. The need for similar standards for educational leaders is a consistent finding of research on school leadership (Lauder, 2000; J. Murphy, 2001). Principals and school leaders today must operate in what educational research often calls the "standards movement," which will continue to be true with the nation's "No Child Left Behind" (NCLB) Act of 2001 (NAESP, 2004, p.1). The restructuring and accountability of schools calls for educational leaders to not only demonstrate results but to show progress and success for all students in the educational process. Keedy and Grandy (1999) stated, "If there were ever a time for innovative, aggressive leadership in our schools, the time is now" (p. 2).

As the forces impacting the principalship have continued to grow, states have begun to adopt new standards for certification and licensure requiring aspiring leaders to go well beyond the traditional administrative degree programs in educational

administration. The Interstate School Leaders Licensure Consortium (ISLLC) Standards were created to develop a framework for redefining school leadership and instructional improvement (see Appendix B). While the Standards were not written specifically for principals, their reference to "school administrator" implies principals or similar educational leaders. The Standards reflect research conducted by a multitude of professional organizations and have been endorsed by such organizations as the National Policy Board for Educational Administration (NPBEA), the National Council for the Accreditation of Teacher Education (NCATE), the Chief Council of State School Officers (CCSSO), National Association of Elementary School Principals (NAESP), and National Association of Secondary School Principals (NASSP) (CCSSO, 1996).

During this time period, "principals already accustomed to being the instructional leaders of the schools now have a whole new set of federally mandated responsibilities and requirements telling them just what 'instructional leadership' means" (NCLB, 2004, p.3). Principals now have to be able to make data-driven decisions to improve instruction and face possible corrective action if schools fail to meet mandated accountability measures.

As schools have entered into a new century, principals are impacted by the No Child Left Behind Act of 2001 (NCLB). NCLB reflects President Bush's stated plan to ensure all children are educated to their full potential by 2014. Key provisions of NCLB include:

- setting higher educational standards;
- annual testing of children to measure progress toward achieving the higher standards;
- analysis of the test data annually to ensure that students are progressing;
 and

rewards (and penalties) aimed at schools where students make (or do not make) "adequate yearly progress" (AYP). (ERS, 2003, p.1)

While all of these forces raise the level of expectation for future school leaders, they do not help school systems with the issue of finding an adequate supply of quality candidates for present and future principalships. Instead, all of these forces have combined to affect both the quantity and quality of the principal applicant pool, yet offer limited advice and support as to how this might be accomplished.

Changes in the Principalship—Implications for Preparation

A constant theme in professional literature reflects the crucial and evolving role of the principal in educational reform (Byron, 1999; Cohan, 1998; Downs, 1999; Lashway, 2003). As the role of the principal has changed significantly in recent years, then it is logical to assume principal preparation programs must also change to keep pace.

As public schools have been asked to take on more and more responsibility, the expectations on principals to meet these additional demands have increased. "Expectations for the principalship have steadily expanded since the reforms of the early 1980s, always adding to, and never subtracting from, the job description" (Copland, 2001, p.4). Many of the authors in the literature question if the job has grown too large for one individual, and if the sheer magnitude of the responsibilities has caused people to leave the position and new applicants to shy away. The days when principals were judged only by such factors as the condition of their buildings, satisfaction of teachers with student discipline, and staying within their budgets are gone. Administrative programs must help future principals lead and change schools, not merely maintain them.

While there appears to be wide-ranging agreement on how complex the principal's job has become, the literature indicates that effective leaders are the ones who can constantly multi-task while giving highest priority to instructional leadership (Trail, 2000, p.2). Because public schools operate in an environment of new standards and accountability, many authors echo the following suggestion, "If school systems want to concentrate on improving student performance, there must be a focus on the principal and the role of instructional leadership" (Croghan, 1984, p.24). This shift from a managerial to an instructional leader is one principal preparation programs must address. According to Hoyle, English, and Steffy, "This involves establishing and communicating a school vision, determining priorities by considering students and staff needs, evaluating school climate, and assessing student achievement data" (Hoyle, et. al, as cited by Harpin, 2003, p.2).

An important theme from the literature review raises the question of whether the evolution of the principal as the instructional leader has, in fact, contributed greatly to the current principal shortage. This question is particularly compelling as "more and more school districts judge a principal's effectiveness as an instructional leader based on the school results on high-stakes tests" (Anderson, 1995, p.22). While there are many factors cited in the literature, such as inadequate remuneration and time fragmentation, to help explain the current principal shortage, job-related stress relating to high stakes accountability is also a growing concern (Ferrandino, 2001, p.7). "The literature is quite clear that principals are under constant stress that manifests itself emotionally, cognitively, and physically. School reform efforts have had a direct impact on the stress felt by principals....accountability pressures and the overall changed nature of the role of the principal have compounded the problem of finding individuals to fill principalships" (Whitaker, 2001, p.4). It is clear that if principals, they must

provide participants with the skills and knowledge to handle new accountability measures.

In an article entitled "The Lose-Lose Leadership Hunt," Gordon A. Donaldson, a professor of education at the University of Maine, expresses concern over how school districts are linking principals' contracts and professional reputations to "high stakes testing and outcomes-based promotion." He writes,

These are hostile conditions for leadership at the school level. The prevailing view asserts that learning and teaching can be "turned around" by the "executive officer" of the school—that schools can operate the way a trim ship or a high-performing business purportedly does. If policy makers at all levels of government wish to attract our very best teachers into school leadership, they will need to recognize how this view of schooling neither supports effective teaching and learning nor promises to reward leaders for taking on the leadership of a school. (Donaldson, 2001a, p.42)

In his article, "The Myth of the Superprincipal," Michael Copland (2001) states.

Two decades into the current age of school reform, one can argue that we have reached the point where aggregate expectations for the principalship are so exorbitant that they exceed the limits of what might reasonably be expected from one person. (p.4)

He argues that it is the overwhelming expectations that currently deter those who previously aspired to principalships.

For aspiring leaders to be ready to meet the ever-changing demands of the principalship, there must be consistent standards in principal preparation programs. The National Association of State Boards of Education Study Group (1999) stated,

"Systemic reform of the principalship begins with ensuring that all principals have the knowledge, skills, and experience they need in order to create an environment that helps teachers teach and students learn" (p.9).

Concerns With Traditional Principal Preparation Programs

As school districts around the country report the quality and quantity of principal candidates are declining, "a broad consensus has emerged in educational policy circles that raising the quality of school leadership is essential" (Keller, 2000, p.2). Multiple studies indicate it is imperative that an adequate supply of principals be available and these principals be well prepared to meet the needs of schools and challenges posed by high-stakes testing and other forces impacting the principalship (Fenwick, 2001; Hargraves & Fuller, 1998; Quinn, 2002; Tirozzi, 2001).

While the demand for quality principal candidates has grown, the supply has not kept pace. Ann Lauder (2000), Coordinator of Leadership Assessment and Development at the Center for Excellence in Leadership at the Education Service Center, wrote,

Districts are finding it progressively difficult to persuade teacher-leaders to consider or prepare for the principalship. Many teacher-leaders have one or more advanced degrees and are reluctant to pursue yet another traditional degree. Disappointment in traditional and theory-based preparation programs, coupled with the public demand for increased expectations on the principalship, has produced a world of new and redesigned principal preparation programs. (p.25)

Until recently, school districts across the country have been content to allow colleges and universities to train future principals. The traditional preparational approach followed for individuals aspiring to become principals has remained very

consistent and relatively simple. First, an educator would enroll in a graduate program after a successful teaching career. The aspiring administrator would then receive a master's degree and "complete the requirements for an entry-level administration certificate" (Daresh, 1990, p.1). While the university was expected to provide the theoretical background for the potential principal, the practical experience was usually gained by attaining a vice principalship.

While this relationship was long regarded as "a near perfect marriage, wherein theory and practice have been united in the creation of ideal preservice preparation" (Daresh, 1990, p.2), this perception has changed dramatically. The "traditional approach" often left schools complaining of ineffective administrators who were not prepared to address their responsibilities and beginning administrators expressing a lack of confidence "that they are ready to step out of the classroom and into the main office for the first time" (Daresh, 1990, p.3).

Much of the current research on principal preparation programs addresses university-based courses and curriculum and cites serious concerns over their effectiveness. "Preparation programs in educational administration have been locked into modes of thinking and structures of practice that have been overtaken by changes in the environment" (Lumsden, 1992, p.2). In a comprehensive study of principal preparation programs, Joseph Murray (1996) asserts that "Changes need to be made in the content and pedagogy of preparation programs" (p.15).

With the changes in school principals' responsibilities accompanying various educational reform efforts, policy makers have recommended that schools and districts truly examine the type of training aspiring leaders are receiving in preparation for principal positions. According to Karyn DeAngelis (1997) of the National Center for Educational Statistics, "Not only must school systems examine what types of programs (internships, practicums, and courses of study) are available, they must ask

the fundamental question of what information is provided in these programs" (DeAngelis, 1997, p.10; Quinn, 2002).

There is increasing evidence to suggest that most college and university programs for the preparation of educational administrators have limited effectiveness. Dating back to the 1960s, Hemphill (1962) indicated that years of formal preparation were weakly correlated with ratings of effectiveness (p.11). Gross and Harriott (1965) showed "that a number of graduate courses were actually negatively related to leadership skills" (p.13). In 1996, the National Association of Secondary School Principals (NASSP) issued a report suggesting that preparatory programs had been overtaken by environmental changes and "were resolved into modes of thinking and structures of practice" that fail to keep up with forces impacting the principalship (Keil & Czernak, 2003, p.1). James Lyons, Chairman of the Educational Administrative Department of the University of North Carolina, wrote, "The novice principal seldom understands the full scope of the issues, expectations, and scope of responsibility that accompany the position" (Lyons, 1992, p.4). The National Commission on Excellence in Education Administration released a report in 1988 calling for immediate reform of principal preparation programs. This report "was the impetus for national and state boards of education and legislatures to recognize the critical need for educational reform of principal preparation programs" (Mercado, 2002, p.14). Joseph Murphy, Chairperson of the Department of Educational Leadership at Vanderbilt University, determined a number of problems associated with current principal preparation programs:

- Non-competitive entrance requirements and little effort to weed out poor candidates;
- An ill-defined knowledge base with few standards;
- Minimal academic rigor;

- A fragmented curriculum;
- Lack of connection to the world of practice;
- Uninspired instructional methods;
- Poorly regarded faculty;
- Lack of diversity in students and faculty;
- Reliance on an academic rather than professional model. (Murphy, as cited in NASBE, 1999, p.10)

Annie C. Lewis, an independent education researcher and writer, has voiced many concerns about the "dismal state" of administrator preparation programs at the university level. She states, "The classes were very theoretical, very textbook-based, and it was very easy to sleep through them" (Lewis, as cited by Keller, 2000, p.2).

Interestingly, many of the calls for change to the existing principal preparation programs have come from university professors in the field of administrative education. G. Cawelti (1982), in his article, "Training for Effective School Administrators," indicates, "Critics of the current university-based programs assert that the traditional pedagogy focuses on school administration, financial preparations, labor negotiations, school law, and facilities planning" (p.14). He states, "Improved university preparation programs and more effective human resource development for practicing administrators are more important than ever. We must now think through what the curriculum for school administrators should be and what pedagogy will best assist administrators to better cope if not lead" (p.14). Increasingly, there is a demand for university classes to "address the skills needed to lead successful schools such as the use of data to improve instruction and the use of effective teaching and learning situations, or to make decisions about aligning courses, classroom assignments, and students' work to standards" (Bottoms & O'Neill, 2001, p.23).

As the landscape has changed for the principalship, there has been a call for new designs and recommendations for principal preparation programs. One of the most common charges leveled against such programs is, "They are unbalanced; students are saturated with educational theory when enrolled in graduate studies, but receive scant experience in the types of professional challenges they are likely to encounter in the real world" (Lumsden, 1992, p.2).

Realizing that aspiring principals may be obtaining degrees with a great deal of theoretical background but little practical experience, a number of school districts are taking a more proactive role in the preparation of their future leaders.

Trends In Effective Principal Preparation Programs

While the call for reform on university-based principal preparation programs seems to be a constant theme in the literature, so is the call for school districts to supplement higher education preparation with programs of their own. A report sponsored by the National Association of State Boards of Education (1999) suggested, "Demands for fundamental improvements in principal preparation and support are linked directly to pressures for educational reform" (p.11). In an era of unprecedented accountability, school districts are beginning to realize that they must share in the responsibilities of identifying and preparing future leaders, or they may well be left with an insufficient quantity of quality principal candidates.

One of the most recent solutions to the principal shortage is "for school districts to grow their own principals" (Keil & Czernick, 2003, p.2). While there is an extremely limited amount of research about "grow your own" principal preparation programs, what is available reflects many common themes and ideas. Fink and Resnick (2001) discuss local principal preparation programs that "create a culture of learning that focuses on accountability for student achievement" (Keil & Czernick,

2003, p.2). Leslie Fenwick (2001), a professor of educational policy at Clark Atlantic University, suggests school systems "seek university-district collaboration in 'grow your own' programs that encourage diversification of leadership at district and school levels" (p.4). School systems are being called upon to take a proactive approach to grow their own principals. Donaldson (2001) stated, "Leaders cannot fully develop the capacity to lead anywhere but in the action of the school. University courses, leadership academies, and professional conferences can be helpful, but we can only learn the interpersonal and intrapersonal lessons of leadership by leading in a highly supportive and reflective environment" (p.3).

In his article, "Leadership 101," John Newsom (2001) reflects that "'grow your own' programs typically have three major features: classroom work, where aspiring principals learn from experts in the field; a full-time internship, where they work alongside veteran school administrators; and a network, where program graduates can share information with colleagues and get feedback from mentors" (p. 14). He suggests that candidates who go through all three phases successfully are very likely to find administrative positions quickly.

While there are great differences in how school systems approach their "grow your own" models, the current research identifies several key components of successful principal preparation programs:

- 1. Field-based exposure with mentors
- 2. Substantial internship experiences
- 3. Cohort groups and study groups
- 4. Focus on leading, not management
- 5. Participant requirements to create a professional portfolio. (Bottoms & O'Neill, 2001; Marshak, 2003; Murphy, 2001)

Daresh's (1997) work (cited in Mercado, 2002, p.4) recommends a number of practices, addressing both content and delivery, that lead to successful educational leadership preparation programs. The content recommendations are:

- 1. Pre-service programs with an emphasis on the development of reflective skills.
- 2. Preparation programs that help principals acquire skills as moral and ethical leaders.
- 3. Principles of adult learning that guide the development of preparation programs.
- 4. Curricula that are coherent, integrative, and sequenced in a logical fashion.
- 5. Great emphasis placed on teaching and learning processes in schools.The delivery recommendations are:
 - 1. Opportunities for more clinical learning must be made available to aspiring principals.
 - 2. Experienced administrators should serve as mentors to future leaders.
 - 3. Aspiring principals proceed toward their goals in cohorts.
 - 4. Authentic assessment techniques must be used to track student progress.
 - 5. Pre-service preparation is viewed as only part of a bigger picture of professional development. (p.4)

Trying to keep up with the ever-changing role of the principal, many school systems have initiated or dramatically redesigned their principal preparation programs to appeal to and attract those educators who may be the next generation of school leaders. Ann Lauder (2000), in her article entitled "The New Look for Principal Preparation Programs," writes, "The major changes include emphasis on the

effectiveness and efficiency of the preparation program, advocacy for application of adult learning methodology, focus on the participants' skills and the results produced, and support for curricular choice based on distinguishing individual needs" (p.24).

Effective principal preparation programs must bridge the gap between what is taught in university-based programs and what skills and knowledge principals need to become true instructional leaders of the schools. A great deal of current research asserts that there is too little emphasis placed on curriculum, teaching and learning, and student achievement in college-based programs (Bottoms & O'Neill, 2001, p.23). By creating their own programs to supplement what potential administrators may have received in university-based programs, school districts can personalize their efforts to meet their particular needs. Samuel Sava (1990) of the NAESP states,

While there are many similarities in America's public schools, it is also true that these schools are different from one another as are the communities they serve. What is considered effective leadership in one school may be less effective in another. Universities must face this dilemma of how to provide principal preparation experiences that will directly relate to any, let alone all, of the communities in which aspiring principals may work. (p. 25)

Creating their own distinct programs allows school districts to personalize their programs in ways that colleges and universities cannot. Ultimately, the most effective principal preparation programs will be those that "find ways in which universities and school districts work collaboratively and as partners" (Daresh, 1992, p.15). This focus on content and collaboration between school systems and universities is one of the most common components discussed in Williamson and Hudson's work (2004), "The Good, the Bad, the Ugly: Internships in Principal Preparation." They assert quality programs for principal preparation include:

- a sense of purpose and vision developed collaboratively with students,
 local school personnel, and practicing school leaders,
- a knowledge base that incorporates the knowledge and skills required for contemporary school leadership,
- a recognition that the primary function of schools is to improve the quality of teaching and student learning,
- a more thoughtful and purposeful inclusion of clinical activities including school-based internships (p.3).

Another common theme emerging about principal preparation programs is the need for participants to engage in reflective practice in cohort groups. Stein (1998) advocates situation learning in classrooms where adult learners engage in simulated group activities, group discussions, and critical reflection where they can verbalize knowledge gained and engage in problem-solving approaches with experts in the field (Browne-Ferrigno, 2001, p.21).

Increasingly, there is a call for principal preparation programs to initiate internships. In her article, "Preparing School Leaders: Case Study Summary and Implications," Tricia Browne-Ferrigno (2001) asserts that aspiring school leaders need to conceptualize the principalship through interaction and socialization with current principals. She states, "Research shows that socialization through clinical practicums and internships increases role clarification and technical expertise, changes role conceptions about the principalship, and develops leadership skills and professional behaviors" (p.29).

Standards For Principals

As school systems have developed their own preparation programs, it is to be expected that the content and pedagogy should be a clear point of focus. With the

concept of school leadership evolving from a mere managerial focus to a greater instructional focus, the definition of the qualities that principals should demonstrate has also evolved. Policy makers have begun to agree "to ensure excellence among all principals, states need to provide a clear picture of an effective principal, continue in a set of standards, and require principals to be evaluated regularly according to the knowledge, skills, and dispositions defined in the standards" (NASBE, 1999, p.13). Richard Schmick (1989), a researcher at Vanderbilt University, argues that principal preparation programs must be built on a solid knowledge base and teach aspiring leaders to "think strategically about situations involving instructional leadership and school improvement" (p.18). Clearly, the absence of uniform licensure standards and a common foundation of knowledge and skills negatively impact the preparation of principals (Thomson, 1993, p.22).

In response to a void in universal standards for school leaders, a set of skills, knowledge, and performance levels of qualified educational leaders was developed by the Interstate School Leaders Licensure Consortium (ISLLC). To date, these standards have been adopted as official licensing standards in 34 states (Berman, 2002, p.4). The ISLLC Standards are now being integrated into principal preparation programs at colleges and universities around the country.

These standards "set out to develop a powerful framework for redefining school leadership and to connect that framework to strategies for improving educational leadership throughout the nation" (Hessel & Holloway, 2002, p.5). The ISLLC Standards may well set the foundation for a definition of what school systems desire in a well-qualified principal. The ISLLC standards call for competency in facilitating vision, cultural and instructional programs, management of organization, collaboration with and engagement of the community, ethics and integrity, and understanding the public.

In their work, "A Framework for School Leaders: Linking the ISLLC Standards to Practice," Hessel and Holloway (2002) note that a number of school districts and states are framing their preparation programs on the ISLLC standards. "Considerable energy is being devoted to linking ISLLC to the reform of formal training programs for school administrators. These efforts run the gamut from individual universities, to cooperative cross-county/university activities, to whole-state initiatives" (p.8).

Throughout the ISLLC standards literature is a recurring theme of their basis and grounding in research. Developed by ISLLC members, the standards are based on:

- A thorough analysis of what is known about effective educational leadership at the school and district levels.
- A comprehensive examination of the best thinking about the types of leadership that will be required for future schools.
- Synthesis of the thoughtful work on administrator standards developed by national organizations, professional associations, and reform commissions.
- In-depth discussions of leadership and administrative standards by leaders within the twenty-four states involved in ISLLC. (CCSSO, 2001, p.2)

While the general concern is that principal preparation programs should differ to meet the needs of an individual school district, this is an age of accountability in education. Utilizing the ISLLC standards as a knowledge base for such programs may provide quality control for principal preparation programs. The empirical knowledge base from which the standards are derived relies on the research linking educational leadership and productive schools. In fact, the decision of the ISLLC members to

focus on standards resulted from "a realization that a thorough review of current literature revealed a major void in the area of educational administration; a set of common standards was conspicuously absent" (Hessel & Holloway, 2002, p.4).

The standards' focus on student learning is consistent with the current research on the changing role of the principal. As changes in society have affected public education, there has been a continual evolution of the principalship. The state in which this study took place adopted the ISLLC Standards for licensure assessment and wrote in a report on the principalship:

The ISLLC Standards reflect the centrality of student learning; acknowledge the changing role of the school leader; recognize the collaborative nature of school leadership; are ambitious, improving the quality of the profession; inform performance-based systems of principal assessment and evaluation; are integrated and coherent; and are predicated on access, opportunity, and empowerment for all staff. (State Task Force, 2000, p. 13)

However, simply using the ISLLC standards does not ensure that a school district has created a satisfactory principal preparation program. While there has been a tremendous growth in such programs in the state in which this study was conducted over the past few years, there have been limited time and effort devoted to evaluation and effectiveness. In her work, "Integration of ISLLC Standards into Rhode Island Principal Preparation Programs," Lisa Harpin (2003) notes several researchers who are not in full support of the ISLLC Standards. Achilles and Price (2001) question "whether or not those prepared in using the standards will actually know and be prepared to do anything of substance to impact student outcomes" (p.11). Even in principal preparation programs that have been recognized as models of effectiveness,

the direct use of the ISLLC Standards has not always been obvious or tied to direct student outcomes.

Leadership Training Programs

While the advent of "grow your own" principal preparation programs has been recent, several aspiring leader programs have been cited in the research. Few studies truly reflect "grow your own" principal preparation programs, meaning a school system developing and implementing a program over which it has total control. These models all demonstrate collaboration between school systems and universities and have engaged in some form of evaluation. One such study was authored by Tricia Browne-Ferrignio (2001), entitled "Preparing School Leaders: Case Study Summary and Implications." While the study was not intended to serve as an evaluation of a particular program of principal preparation, this exploratory study did seek to "understand the nature of changes and the processes through which they occurred, and to link participants' professional growth to their readiness to assume school leadership positions" (p.1). This case study spanned one calendar year and focused on the experiences of 18 educational practitioners participating in a leadership preparation program collaboratively sponsored by an urban university in a western state and several local school systems. Data were collected through several instruments in real time as the participants were actively engaged in the program and triangulated through three different methods. Data collection included pre- and post-surveys, open-ended questionnaires, and a series of semi-structured interviews.

The program was sequenced around four learning domains: educational leadership, school environment, supervision of curriculum and instruction, and school improvement. Each domain was then connected to concurrent field experiences.

Browne-Ferrigno (2001) reported her findings in this case study through the identification of four themes: (1) career aspirations, (2) leadership experience, (3) role conceptualization and socialization, and (4) learning in a cohort. Some of the more prominent findings of this study were "that experiential learning must be the core element of principal preparation to ensure needed skill development and socialization into the community practice" (p.3). She also noted that career counseling is needed for aspiring principals, especially women, to ease the transition into administrative positions (p.38).

Another study utilized formative evaluation data to guide program design. Authors Bradshaw, Perreault, McDowelle, and Bell (1997) wrote "Evaluating the Results of Innovative Practices in Educational Leadership Programs" detailing aspects of a principal preparation program implemented by East Carolina University and local school systems. This program was implemented because of changes the North Carolina Legislature mandated in principal preparation programs. Based on the evaluative data gathered collaboratively with program participants and instructors, several innovative components of the program were highlighted. First, this program included an extended internship for program participants. Designated as "Principal Fellows," participants "quickly assumed responsibilities in substantive areas (such as student discipline and teacher evaluation) while others were assigned 'token' responsibilities (such as directing a small project to support school volunteers) and struggled to be viewed as significant contributors to the leadership effort in their schools" (p.4). Despite differences in assignments, most Principal Fellows viewed the experience as satisfactory in surveys and interviews.

The study also highlighted the use of "the Springfield Simulation," a program sponsored by the National Association of Secondary School Principals (NASSP) emphasizing skill development in six areas: problem analysis, judgment,

organizational ability, decisiveness, leadership, and sensitivity. Principal Fellows focused on two of these skills and were put though simulation activities aligned with North Carolina leadership performance domains.

The Leadership Portfolio was also seen as a program strength and was framed around the ten North Carolina leadership domains. The authors proposed the use of the portfolio, stating, "The development of a culminating portfolio allowed Fellows to integrate common assignments and field experiences, and it established a process for continuing professional development" (Bradshaw et al., 1997, p.3).

Several areas of improvement were noted by the authors. First, there was a recognition of the need to align the program more closely to the ISLLC standards. The authors stated, "We have a responsibility to prepare students in a way that will facilitate their success on the examinations at the same time as we try to take what is best from the state standards and the NASSP simulations that have been so helpful to our students" (Bradshaw et al., 1997, p.14). Finally, there was a concern on keeping the program's focus on problem-based learning instead of deviating toward more theoretical-based situations. Trying to keep the practical and the theoretical balance of the program was extremely important to everyone involved.

This program has received national attention because of its positive results. Since 1995, its first year, the Principal Fellows Program has graduated 442 administrators, placing principals and assistant principals in about 20% of the state's schools (Newson, 2001, p.32).

Another study that utilized formative evaluation tools was Ludmille Pueblitz Mercado's "The Secondary Principalship Academy: A Critical Ethnography of the Houston Independent School District and the University of Houston's Innovative Principal Preparation Program." Mercado (2002) stated that "documenting the states of planning, development, implementation, and program effectiveness of the

Secondary Principalship Academy, as well as identifying the perceptions of the participants, was the intent of this study" (p.8). Mercado conducted his critical ethnographic research utilizing interviews, field notes, and program observations.

The Secondary Principalship Academy was founded to recruit 20 aspiring administrators who would engage in a 24-month study developing leadership skills through four types of experiences: problem-based learning, action research projects, field experiences through a secondary campus-lead internship, and topic seminars taught by school system personnel. Participants were to develop a portfolio to highlight their accomplishments.

Findings of this study concluded that an internship year was the program's cornerstone. This occurred during the second year of program participation, and individuals were removed from the schools to assume a full-time paid internship position with a mentor principal from the school district. Program participants were placed in cohort groups to support each other and learn from their experiences.

An unexpected finding of this program was the "inflexible attitudes of district principals and district superintendents toward the participation in this program" (Mercado, 2002, p.8). While these interactions with mentors were not hostile, career administrators seem to be locked into traditional views of these administrative duties over instructional leadership.

While these programs were not truly "grow your own" preparation programs as defined in this study because they all employed more university planning and implementation than direct school system development, they serve as models of principal preparation programs that have been nationally recognized and evaluated. These research models did not state the evaluative tool used to determine each preparation program's effectiveness; however, for the purpose of this study, it was necessary to select and describe the evaluation tool used.

Relationship of Literature Review to This Study

The review of literature establishes that as the role of the principalship has evolved in recent years, methods of principal preparation have come into question. As the principalship has grown from a predominantly managerial role to one of instructional leader in an era of increased accountability, many researchers have criticized traditional university-based training models as insufficient or obsolete. Studies suggest that many programs rely too much on theory and do not provide the practical experiences needed to ensure a quality pool of future principals.

As the need for quality principals continues to grow as more current administrators reach retirement age, it is logical that school systems take an active role in principal preparation by creating "grow your own" programs. The literature research suggests that such programs must be framed around a set of standards, such as ISLLC, that clearly articulate the knowledge, skills, and indicators successful school leaders must possess. Other common characteristics of quality principal preparation programs include partnerships with universities, hands-on activities that deal with practical educational issues, and incorporation of some form of internship.

The central purpose of this study was to examine the effects of one school system's efforts to create and implement a "grow your own" principal preparation program. This research effort attempts to answer six questions—two questions deal with program control, one with program inputs (resources), one with process, and two with product. The six questions are as follows:

- 1. What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program?
- 2. Which program elements were created specifically to address those forces impacting the principals?

- 3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program?
- 4. What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants?
- 5. What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role?
- 6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement?

After reviewing several models of evaluation, the CIPP model, designed by Daniel Stufflebeam, was selected to provide the structure for this formative evaluation study. The research questions were aligned with Stufflebeam's core components of content, input, process, and product, which will be outlined in Chapter III.

CHAPTER III

RESEARCH METHODOLOGY

Restatement of Purpose

The purpose of this study was to examine one school system's efforts to design and implement a "grow your own" principal preparation program to address issues deriving from a shortage of qualified principal candidates. This chapter presents the procedures used in collection and analysis of data and information essential to answer the research questions raised in Chapter I.

The methodology for this study, which includes both quantitative and qualitative procedures, is presented. A description of the population, the instrumentation, the variables, and statistical analysis is discussed. This chapter begins with a review of the literature on program evaluation. An overview of the Stufflebeam CIPP Model, along with an explanation of its use in this study, is addressed.

Program Evaluation

Although there is a growing body of literature citing changes and modifications to leadership development programs (Kelley & Peterson, 2000; Milstein & Kruger, 1997), "research about the effectiveness of program designs is limited" (Murphy, 1993, as cited by Browne-Ferrigno, 2001, p.25). While a number of school systems have created "grow your own" principal preparation programs over the past several years, decisions on how to evaluate these programs have often been left unanswered. Of the 17 school systems with principal preparation programs in the

state in which this study was conducted, only two documented any type of formal program evaluation methods or designs (State Survey, 2004, p.2).

Nonetheless, there is a good deal of information about program evaluation during the last quarter century. Michael Scriven (as cited by Fitzpatrick, 2004) was the first to distinguish between formative and summative evaluations, which serve as the foundations of most evaluation models. As defined in *Program Education:**Alternative Approaches and Practical Guidelines*, the primary purpose of formative evaluation is to "provide information for program improvement..." and provide "information to judge the merit or worth of a part of the program" (p.16). As such, formative evaluation tends to look at aspects of the program implementation as well as tentative outcome data. Summative evaluations "are concerned with providing information to some decisions or assist in making judgments about program adoption, continuation, or expansion" (Fitzpatrick, 2004, p.17). Summative evaluations, while somewhat concerned with operations data, are primarily outcome focused.

Much of the literature on program evaluation describes a number of models and concerns about the subject. In his work entitled, *Hard-won Lessons on Program Evaluation*, Michael Scriven (1992) suggests five main approaches or views in program evaluations "in order of their period of dominance in evaluation" (p.3).

These major perspectives include:

- The "decision support view" based on the work of individuals such as Ralph Tyler and Daniel Stufflebeam that purported the idea of program evaluation as part of rational program management.
- 2. The "relativistic" approach where the client's viewpoints were used as a framework without commitment by the evaluator on their value.

- The "rich description" approach where evaluation is done as an ethnographic or journalistic enterprise and the evaluators report without making evaluative statements or judgments.
- 4. The "social process" school notable for its rejection of the importance of evaluations to support decisions about programs, and the substitution of understanding social programs as a basis for evaluating them.
- The "constructivist" approach which rejects evaluation as a search for quality, merit, worth, etc. in favor of the idea that it is negotiation.(p.2)

In his work entitled, *Assumptions Underlying Evaluation Models*, Ernest House presents a history and outline of a number of major evaluation models. House notes, "The major elements in understanding the models are their ethics, their epistemology, and their political ramifications" (House, as cited in Madaus et al., 1978, p.47). Several of the models cited by House include:

- Systems Analysis, whose major proponent, A.M. Rivlin, examined
 quantified variables and cost benefit analysis to determine if expected
 efforts are achieved efficiently. The major audiences for this model are
 economists and managers.
- 2. Behavioral Objectives, whose major proponents, Ralph Tyler and W.J. Popham, examined the objectives of a program with quantified outcome variables. Behaviors are measured by norm-referenced or criterion-referenced tests. Managers and psychologists are the major audiences of this model.
- Decision-making, with Stufflebeam arguing that evaluation is structured by the decisions that are made. The evaluations supply information based on the decisions using questionnaires and interview

- surveys. Administrators are the major audience targeted by this model. CIPP falls into this category.
- Goal-free, with Scriven serving as the major proponent. He argues the need to reduce bias in evaluation by searching for all possible outcomes. Consumers are the major audience for this model. (Madaus, 1978)

While there are many types of evaluation models from which to choose, looking for a model that is informative to decision-makers led the researcher of this study to select the CIPP model created by Daniel Stufflebeam.

The CIPP Evaluation Model

As the focus of the study was to examine one school system's experience of designing and implementing a principal preparation program, choosing an effective evaluation model was critical. This study used Stufflebeam's model to examine the content, input, process, and product components of one school system's experience in leadership development.

Realizing that educational administrators perform many evaluative tasks, it is reasonable to assume they should be guided by a sound, clear conceptualization of evaluation. Daniel Stufflebeam's CIPP model is one such conceptual framework.

The CIPP model is a comprehensive approach to evaluation. Originally introduced over 35 years ago, Stufflebeam's model stresses the need for process as well as product evaluation. The definition of evaluation, which is basic to the understanding of the CIPP model, is "the process of delineating, obtaining, and providing useful information for judging decision alternatives" (Stufflebeam, 1971, p.4). Using this definition, Stufflebeam (1971) asserts that evaluation is conceived of

as a systematic, continuing three-step process of delineating questions, obtaining relevant information, and providing information to decision makers (p.15).

A basic tenet of the CIPP model is that evaluation should serve as a tool for decision making. According to the CIPP model, "There are four kinds of decisions—planning, structuring, implementing, and recycling—which respectively are served by context, input, process, and product evaluations" (Stufflebeam, 1971, p.5). In general, these four parts of evaluation respectively ask the following: What needs to be done? (context) How should it be done? (input) Is it being done? (process) Did it succeed? (product) (Stufflebeam, 2002, p.2).

The question of "Did it succeed?" is particularly important to the CIPP Evaluation Model. Stufflebeam asserts that this part of the checklist examines "Were the beneficiaries rewarded? Were their needs met? Were the gains for the beneficiaries sustained? Did the processes that produced these gains prove transportable and adaptable for effective use in other settings?" (Stufflebeam, 2002, p.1).

What made the CIPP model appealing for this study was that it provided a focused evaluation tool to review an organizer's decisions, assess a program's history, and disseminate the effective services provided to targeted individuals. Using the CIPP model, a final synthesis report can be expected to "pull together evaluation findings to inform the full range of audiences about what was attempted, done, and accomplished, what lessons were learned, and, the bottom line, assessment of the program" (Stufflebeam, 2002, p.9).

Since it was first introduced in 1966, the CIPP model has undergone several revisions. Currently in its fifth installment, the model now includes a ten-component checklist to "break out product evaluation into four subparts in order to help assist and

assess a program's long-term viability" (Stufflebeam, 2002, p.2). The ten-component checklist for the fifth installment of the CIPP model is as follows:

- 1. Contractual agreements
- 2. Context evaluation
- 3. Input evaluation
- 4. Process evaluation
- 5. Impact evaluation
- 6. Effectiveness evaluation
- 7. Transportability evaluation
- 8. Sustainability evaluation
- 9. Meta-evaluation
- 10. The final synthesis report. (Stufflebeam, 2002, p.1)

Utilizing the CIPP model allowed the evaluator to not only determine the merit and significance of the program under consideration, but also allowed for careful reflection of lessons learned. The components of the CIPP model provided the framework for this study. The history of the program being studied and key decisions about its implementation and resources used were addressed using the checklist under context, input, and process. Stufflebeam (2002) suggests, "The CIPP model's main theme is that evaluation's most important purpose is not to prove, but to improve" (p.2). The overall structure of the CIPP model and its expressed goal of aiding in program improvement are what made it an ideal tool to be used in this study.

While most of what is written about the CIPP model is positive in the literature review, Stufflebeam's work does have some detractors. Scriven (1992) suggests that "the CIPP model was a little overgeneralized in that it claimed all program evaluation was orientated to decision support" (p.3). He argues the CIPP's expressed purpose regarding evaluation, to improve rather than prove, is adequate for formative

evaluation, but narrows and fragments the overall concept of evaluation. He further contends that the CIPP model "seems about the most complicated and confusing way of analyzing the practical procedures of evaluation...and it's certainly the most complicated one I've ever seen" (Scriven, 1971, p.36). Not only does he suggest that for the CIPP model to be effective for school personnel, an intensive amount of inservice training is needed, but also he concludes that the benefit will not be commensurate with the cost of such training (Scriven, 1971, p.36).

While William B. Michael (1971) in his article, "Educational Evaluation and Decision Making," views the CIPP model much more favorably than Scriven, he too raises issues with some parts of the model. He notes there must be a concentrated and focused effort throughout the stages of context, input, process, and product evaluation to identify "explicit and implicit inferences regarding possible cause and effect relationships" (p.27). He also suggests that greater development of methodology for setting value systems is needed. Finally, he raises a concern that external validity issues may be more likely than internal validity issues, explaining, "Threats to external validity may be due most often to a lack of randomization or to the lack of the evaluator to assume a position of power and influence which he might assume in evaluation studies involving decisions about a multimillion dollar educational enterprise" (p.29). Michael concludes his paper by stating the CIPP model is likely the most comprehensive conceptualization of evaluation currently available.

While the CIPP model does have some detractors, others have found it an extremely useful model to evaluate a program or product. Les Goodwin used Stufflebeam's CIPP model for his work entitled, *A Proposed Model for Educational Accountability*. Goodwin (1975) stated that the CIPP model not only provided a framework for "conceptualizing the decision-making which is required in an accountability program," but also "focused attention on the associated information

required for making those decisions" (p.51). Goodwin praised the CIPP model for establishing how content, input, process, and product worked together systematically and demonstrated a relationship to the decision-making process in educational accountability. The author also noted "utilization of this model also permitted focusing on only those factors which were necessary for the comprehension of the total accountability system, thus limiting the study to its most significant aspects" (Goodwin, 1975, p.51). Using the CIPP model, Goodwin developed a model for educational accountability for school officials that provided direction and focus as they engaged in examining their ongoing accountability programs.

The choice of an appropriate evaluation model was essential to this study. The CIPP model is logically organized and provided for an opportunity to examine the history of the program as well as assess its overall effectiveness. Finally, Stufflebeam's position that CIPP's goal is to improve rather than prove a particular point of view provided an opportunity to assist the system in this study in building a better program.

Application of CIPP Model to Study Research Questions

The six research questions for this study were framed around the four components of the CIPP model of evaluation. The first two questions reflected context; question three reflects input; question four reflects process; questions five and six reflect product.

This study emphasizes both qualitative and quantitative methods. In their work, *Education Research: An Introduction*, Gall, Gall, and Borg (2003) suggest that the research design should be driven by the existing knowledge in the field of study and the research questions. Because this study was a formative evaluation of one school system's efforts to design and implement an aspiring principal program, the use

of quantitative and qualitative methods was appealing. Fitzpatrick, Sanders, and Worthen (2004) stress the value of both methods in evaluative work stating, "Too often we fail to encompass methodological flexibility, unthinkingly adopting a single-minded perspective that can answer only questions stemming from that perspective's discipline" (p.64). The use of the CIPP model allowed for a wide range of methods to collect data in examining this school system's principal preparation program.

Table 1 summarizes the components of the CIPP model ad anticipated methodologies associated with the research questions. Details regarding design, methodology, and procedures are provided throughout the remainder of the chapter.

Location of the Study

The study was conducted in a school system within a mid-Atlantic state. The school system ranks in the middle of the state's school districts in terms of population size, but has recently been identified as one of the fastest growing in the state. It has both suburban and rural characteristics as the population has grown increasingly transient. The majority of new growth is centered in a development zone located in the northeastern section of the county. There are currently 31 schools in the system, with a new science and technology high school opening in the fall of 2005. The school system has a student population of approximately 25,000 with a racial breakdown of approximately 55% Caucasian, 38% African-American, 3% Hispanic, 3% Asian, and 1% other.

Table 1

Research Questions and Alignment with the Ten CIPP Components Checklist

Research Questions	10-Component Checklist of CIPP		
1. What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program? (context)	Context Evaluation		
2. Which program elements were created specifically to address those forces impacting the principals? (context)	Context Evaluation		
3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program? (input)	Contractual Agreements Input Evaluation		
4. What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants? (process)	Process Evaluation		
5. What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role? (product)	Impact Evaluation Effectiveness Evaluation Transportability Evaluation		
6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement? (product)	Effectiveness Evaluation Sustainability Evaluation Meta-evaluation		

The subjects studied and surveyed were participants from the first three cohorts of the program. After the third year of the program, 99 participants were available for the study, which included teachers, instructional leaders, guidance counselors, and vice principals. Table 2 provides some descriptive information on the program participants.

Table 2
Aspiring Principal Program Participants

	Position				Race			
	Vice Principal	Team Leader	Guidance Counselor	Teacher	White	Black	Hispanic	Other
Cohort 1	16	4	0	27	32	13	2	0
Cohort 2	3	1	1	20	17	8	0	0
Cohort 3	0	1	0	26	19	7	0	1
Total	19	6	1	73	68	28	5	1

Members of the original design team were also surveyed for this study. These twelve individuals included central office staff, principals, vice principals, and teachers.

Research Methods

A great deal of data was gathered for this study using both qualitative and quantitative methods. Data was gathered to address the four components of the CIPP Model as defined by six research questions.

According to Gall, Gall, and Borg (2003), qualitative research can be used to discover themes, patterns, and relationships in sample populations. One of the guiding principles of qualitative research centers on purposeful selection of informants, participants, and documents (Creswell, 1994). Most of the research questions in this study can best be answered through in-depth interviews and a thorough review of documents. The researcher selected candidates to be interviewed based on their knowledge and experience with the program. Personal interviews were conducted with key Central Office personnel, such as the superintendent, associate superintendent, assistant superintendent of instruction, assistant superintendent for human resources, and staff development coordinator. Focus group interviews were

conducted with members of the program design team, program participants, and program participants who have attained principalships.

Quantitative methods for this study were primarily centered on the use of a survey of program participants. Surveys tend to measure attitudes and behaviors and "constitute one of the most important data collection tools available in evaluation" (Fitzpatrick, Sanders, & Worthen, 2004, p.341). Survey results allowed for addressing how effective program participants thought the principal preparation program was as well as determining areas of strength and need for improvement.

For the purposes of this study, the researcher selected an instrument around the proposed research questions. The questionnaire addressed the perceptions of program participants about initial reactions to its effectiveness. The survey was administered to the 99 participants of the first three cohorts of the school system's Aspiring Leaders Program. It was also administered to the twelve original members of the Design Committee. While a fourth cohort had begun the leadership training at the time the survey was administered, they were not included in the study as they had not completed the entire program.

Data Collection Techniques

The research questions for this study were framed around the CIPP model and required both qualitative and quantitative data.

Question 1 of this study asked: What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program? Context evaluation proceeded by examining the change forces, system needs, and system concerns that led to a program being developed. Primary data sources for this question were the 12 program design committee members, who have been with the program since its inception, and key central office figures, such as the

superintendent, the associate superintendent, and the former director of professional development. Much of the qualitative data for this study were obtained through a focus group interview process. Sets of interview questions were developed that focused on the "role sets" to be interviewed. Interview protocols for all groups were developed for this study (see Appendix C). Permission was obtained from the approved university and school district before conducting interviews. Most of the questions and the order in which they were asked were determined in advance of the interview, although extended questioning based on the respondents' answers was pursued. Face-to-face interview responses were audio-taped with the participants' permission. When these individuals were referenced in this study, they were classified by job or role and are not identified by name. As is the case in most qualitative research, the researcher traveled to the site or location to observe all individuals being interviewed and recorded their behavior in their natural setting.

A thorough review of selected documents appropriate for the study was conducted to accompany the interview process. Gall, Gall, and Borg (2003) emphasize that the context in which the document was produced must be considered in order to obtain a full meaning of the document itself. Documents that were released for this study, particularly Question 1, included *The Aspiring Principal Leadership Design Folder*, containing committee meeting notes and agendas. Internal memos detailing the purposes for formation of a program were also reviewed. After determining the relevance of these documents, they were coded and categorized to assess information and assist in analysis and interpretation. Table 3, on page 64, provides a complete overview of data sources, methods of collection, nature of data, and data analysis procedures for each question.

Question 2, which also focused on context evaluation, asked which program elements were created specifically to address those forces impacting the principals. It

explored program elements created specifically to address forces of change impacting the principalship. As with Question 1, the main source of data was program design committee members and key central office staff. The primary method of data collection was face-to-face interviews with the purpose of exploring issues and topics in detail and seeking to understand the respondents' opinions and beliefs.

Question 3 asked what resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program. It examined personnel time and effort, expenditure of system funds, out-of-system personnel, financial support, facilities, and equipment needed in the design and implementation of the program. System financial records, documents, contracts, and program records were the main source of data to answer Question 3. Interviews and document reviews were used to verify and clarify information obtained from these sources.

Question 4 asked what were the essential sources that led to the elements and structure of the designated program, and to what extent they reflected current research about effective principal preparation programs and standards of licensure. What was the process for implementing the program and recruiting participants? It utilized process evaluation and examined program components, delivery, adjustments, and participant selection. Program design members and system principals, along with system records and the initial plan of the program, were the main data sources.

Documents were reviewed with a particular focus on the integration of the ISLLC Standards into the principal preparation program. Interviews with program participants provided information about how participants were recruited into the program.

Question 5 examined product from the CIPP model and involved "impact evaluation," "effectiveness evaluation," and "transportability evaluation." It asked:

What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role? This question sought to understand perceptions of program participants both before and after completing the program. Perceptions of the program design committee were also examined. For this question, non-experimental quantitative research methodology was employed. While the instrument for this study was framed around the specific research questions, it was adapted from a survey titled *The Audit of Principal Effectiveness* developed by Jerry Valentine and Michael L. Bowman in 1986 (Appendix A). The survey was designed to describe teachers' perceptions of principals' effectiveness. Valentine and Bowman created this instrument "to provide insight about the ability of the principal to work with personnel inside and outside the school setting to establish processes and relationships that effectively promote growth and change of the organization" (Valentine & Bowman, 1986, p.1).

Items for this survey were generated from an extensive literature review and research on the role of the principal. This instrument was chosen after the researcher examined a number of instruments. The instrument is closely aligned to significant aspects of the principal preparation program being studied. It also reflected many aspects that are similar to the ISLLC standards, which was used in the formation of the program being studied. It has been used in a number of studies about the principalship, including Henry S. Williams's study in 2001 entitled "Teachers' Perceptions of Principal Effectiveness in Selected Secondary Schools in Tennessee." The instrument as modified for this study has 80 questions. The survey instrument was scored using a Likert type scale ranging from 0 to 4 as the measure of effectiveness. Responses from 4 signified program expectations "to a great extent," a

3 signified "to some extent," 2 signified "to a limited extent," and a 1 signified "not at all." Responses of "Don't know" were coded as a 0.

The Audit of Principal Effectiveness is framed around three domains of principal skills. The Domain of Organizational Development, which addresses organizational direction, linkage, and procedures, is addressed by survey items #1 – 27. The Domain of Organizational Environment, which covers teacher relations, student relations, interacting processes, and affective processes, is addressed by survey items #28 – 64. The Domain of Educational Program, which entails instructional improvement, curriculum improvement, and overall effectiveness, is covered by items #65 – 80.

The survey developed for this study requested demographic information, current job status, as well as career aspirations. This survey was e-mailed to the 99 program participants from the first three cohorts and members of the design team. To protect the anonymity of the respondents, questionnaires were e-mailed along with a tracking code. A log was kept of the individuals to whom the questionnaires were e-mailed and addresses and mailing dates were noted. A follow-up letter and questionnaire were sent to potential respondents when a reply was not received within the allotted two weeks.

Question 6 centered around the product component of the CIPP and addressed "effectiveness evaluation," "sustainability evaluation," and "meta-evaluation." It asked: To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement? This question explored how participants who have attained leadership positions perceive the effectiveness of the program. Interviews for this question were conducted in focus groups. One group included the five program participants who have become

principals. Another focus group included eight participants who have attained leadership positions such as vice principals, instructional specialists, and department heads.

According to Morgan (1998), "Focus groups can be used to complement other research methods, particularly for triangulation and for validity checking" (Morgan, as noted in Harpin, 2003, p.71). In addition to these focus groups, interviews with key Central Office staff helped to determine in what ways these individuals perceived that the program assisted newly appointed leaders.

Data Analysis

Because both quantitative and qualitative methods were used for this study, the data were analyzed using procedures for each method. A great deal of the qualitative data for this study was collected through interviews. Greenhalgh (1997) notes it is not enough to complete interview transcripts and final notes. He states, "It is simply not good enough to flick through the text looking for interesting quotes which support a particular theory. The researcher must find a systematic way of analyzing his or her data and, in particular, must seek examples of cases which appear to contradict or challenge the theories derived from the majority" (p.5). Interviews were transcribed and respondents were able to review the transcripts, making corrections or additions as necessary. This process added internal validity to the study by allowing participants to verify their own words and ensure their thoughts were captured accurately.

These interviews were categorized through context analysis "drawing up a list of coded categories and 'cutting and pasting' each segment of transcribed data into one of these categories" (Greenhalgh, 1997, p.5). From the interviews and records reviewed, the researcher was able to recognize elements referenced frequently. Significant patterns and clusters emerged from which to draw conclusions.

The survey instrument was analyzed through quantitative procedures. The data were analyzed using the mean and standard deviation. Descriptive statistics, including the t-test for independent samples, were utilized. The Chi-square test for independence and the analysis of variance were used to identify areas where significant differences existed between how program participants and designers viewed the program. Table 3 summarizes these activities.

Limitations

The researcher was the primary instrument in the collection of data and analysis for this study. As was previously noted, the researcher has been involved in this system's principal preparation program since its inception, so researcher bias is a concern. While the researcher attempted to limit the amount of bias in the study by using multiple methods of collecting data, the potential for biased judgment was addressed as a limitation of the study.

Gall, Gall, and Borg (2003) noted that in qualitative research, the researcher becomes personally involved with the participants who are being asked to provide honest answers to the questions being studied. Some respondents in this study may not have been completely candid about articulating concerns in the program. Using a survey instrument where the respondents' identities were protected should have helped to limit this possibility.

Finally, this study is restrictive in nature. The findings represent only the system in which the study was conducted and may not be generalizable to other systems attempting to implement "grow your own" principal preparation programs. Nonetheless, it should have informational value as to design, implementation, and outcome possibilities.

Table 3

Data Analysis Chart

Data Analysis Procedure(s)	Qualitative: Organize into clusters Look for patterns Draw conclusions	Qualitative: Organize into clusters Look for patterns Draw conclusions	Qualitative: Organize into clusters Look for pattems Draw conclusions	Qualitative: Organize into clusters Look for patterns Draw conclusions Quantitative: Descriptive statistics Means t-test for independent samples	Quantitative: Descriptive statistics Means t-test for independent samples	Qualitative: Organize into clusters Look for patterns Traw conclusions
Nature of Data	Qualitative	Qualitative Opinions Beliefs	Qualitative Questions Descriptive	Qualitative	Quantitative	Qualitative
Method of Collection	Personal Interviews Records	Personal Interviews Focus Group Interviews Records	Interviews Focus Group Interviews Records	Focus Group Interviews Records Survey	Survey	Focus Group Interview Survey
Data Source	Program Committee Central Office staff	Program Committee Central Office staff	Program Records System Financial Records Program Committee	Program Committee Committee Meeting Records Principals	Program Participants Program Design Committee	Select Program Participants Central Office staff
Research Question	Program Content: change forces system needs system concerns and opportunities other	2. Program Elements dealing directly with forces of change	 3. Program Input: Personnel time and effort System funds Out-of-system personnel and financial support Facilities/equipment Other 	 4. Process Program components Program delivery Program adjustments Program participant selection Other 	 5. Perceptions of Program Participants: Prior to program After completion Perceptions of Program Design Committee 	 6. Leadership: Perceptions of participants who have attained leadership positions

Validity of Data

There are two known research threats to data validity. These two threats are described as internal and external validity of data. Internal validity addresses issues and questions relative to data accuracy. External validity addresses issues and questions relative to the authenticity of the data.

Data authenticity or legitimacy occurs in circumstances where the data sources have questionable origins and there is reason to be concerned about whether or not someone has purposefully manipulated the data. This concern does not arise in a study of this magnitude in that no one stands to gain or lose by altering the data. Data accuracy, or internal consistency, is a larger problem in a study of this nature, especially with respect to the qualitative data and, to a lesser extent, with the quantitative.

The Audit of Principal Effectiveness (Valentine & Bowman, 1986, p. 1) instrument selected for this study was developed with the singular purpose of assessing three aspects of principal effectiveness, each of which is supported by the literature on principal effectiveness (Valentine & Bowman, 1986, p. 1). In addition, the instrument was designed and validated with reference to its consistency and overall accuracy. In addition, this researcher has further analyzed the current data utilizing intercorrelational statistics for the purpose of determining the extent to which the instrument reflects a similar degree of consistency and accuracy in terms of the three dimensions of principal effectiveness that it was designed to assess (Valentine & Bowman, 1986, p. 1).

The issue of internal consistency as it related to the qualitative data was a concern with respect to both the personal and focus group interviews. A concern for the researcher's role in both the design and implementation of the program had to be addressed. Also of concern was the fact that those individuals who were the subjects

of the interview process also had a stake in the program's success. The question arose as to whether or not bias may be operative in their responses and/or the interviewees' interpretations. In order to obtain the fullest degree of accuracy of interviewee responses, the researcher analyzed the interview data in accordance with standard procedures for qualitative data analysis. This consisted of first analyzing the data for consistency within and among respondents, noting any statements that were at the extreme, and conducting follow-up interviews to clarify possible misinterpretations on the part of the interviewer. In addition, the summary of each interviewee's responses was submitted to that individual for any corrections, additions, deletions, or modifications that might have been warranted. These precautions helped to minimize researcher bias.

Once each interviewee had the opportunity to reflect on his/her responses as interpreted by the researcher, those responses were analyzed according to the procedures discussed above in the "Data Analysis" portion of this chapter.

Focus Group Interviews

Several researchers (Krieger, 1988; Merton, Friske & Kendall, 1990; Melinda Lewis, 2000) comment on the focus group interview, citing both its benefits and limitations. Each attests to focus group interviewing as an appropriate tool for assessing programs and/or project development at the various stages of development—beginning, during, and at the end.

These same researchers identify the benefits of its use as including its low costs, its widespread view, and possible consensus building characteristics. They also agree that the focus group interview has certain limitations. Merton, Friske, and Kendall (1990) caution that researchers have a tendency to misuse the data by failing to acknowledge its questionable reliability and validity (p. xxi). Krieger (1988) also

sees this as a concern. Melinda Lewis (2000), in an overview of focus group interview research, also comments on this concern and indicates that focus group researchers need to use focus group data in conjunction with other researcher data. She further advises that the focus group interview be based on data such that the focus group questions should emanate from what is already known.

Lewis cites several early advocates of focus group research (Krieger, 1988; Rice, 1931) who expressed concern relative to focus group questioner bias. Both indicate that the questions need to be more directed and that a recorder, human or machine, be employed to guard against researcher bias.

A U&CD Center for Development Information and Evaluation (1996) offers a number of steps for conducting focus group interviews that emerge from the analysis of focus group research. These steps include the following:

- 1. Select the team. The team should be kept small, 5-10 participants, and include a facilitator, recorder, and selected participants.
- Decide on timing and location. The timing should be such as to assure the support of other research data, and the location should ensure convenience and privacy to the participants.
- 3. Prepare a discussion guide that covers the topics and issues to be addressed.
- 4. Conduct the interview. The facilitator should be someone who has knowledge of the program or project and is able to establish rapport by outlining the purpose and format so as to put participants at ease.
- 5. Phrase questions carefully. Questions need to be open-ended and non-directive. Avoid "why' type questions in favor of "what" and "how" type. The questioner needs to know how to "narrow" such questions if

- the discussion becomes too broad. Use probing techniques. Questions should be limited to no more than 10.
- 6. Record the discussion. A recorder needs to be employed in order to free the facilitator to concentrate on the interview activities. Recorders may be of an audio or audio/video nature.
- 7. Analyze the results. After each session, the team needs to analyze the results. The analysis should include:
 - a. Words -- weigh the meaning of key terms participants used
 (different terms may fit in the same category)
 - b. Framework consider the context in which a comment was made.
 - c. Internal agreement determine if shifts in opinions were the results of group pressure or based on new information.
 - d. Response precision give greater weight to responses based on personal experience as opposed to those based on vague, impersonal impressions.
 - e. Develop the big picture pinpoint and focus on major ideas.Allow time to step back and reflect on major findings.
 - f. Develop the report. Consider the objectives of the program and the information needed for decision-making. The report should include (1) a brief oral report highlighting key findings,
 - (2) descriptive reports that document key findings, and
 - (3) analytical reports that provide trends, patterns, or findings along with selected comments.

The proposed focus group interviews for this research effort incorporated the above steps as well as efforts to prepare the facilitator for his role. To further ensure

that researcher bias was not a problem in this study, another individual was selected to be the facilitator of the focus group interviews. Not only was this individual a professor emeritus of a major university in the state where the study was conducted, but he also served as a consultant for this principal preparation program. This individual not only facilitated the focus group interviews, but also assisted in the analysis of data gathered from them.

Summary

Using qualitative and quantitative methods was necessary in conducting this study. This study attempted to answer why a school system chose to respond to forces of change on the principalship by creating a "grow your own" principal preparation program. The Stufflebeam CIPP model provided a framework to examine the context, input, process, and product of the program this school system ultimately designed and implemented. While the data collected and analyzed only addressed the research questions proposed in this study for the school system in question, this work may have important findings that can assist other school systems trying to design and implement "grow your own" programs. The findings from this study are presented in Chapter IV.

CHAPTER IV

FINDINGS

Introduction

As the country faces a well-documented shortage of qualified teachers, educational leaders are also struggling with another critical shortage—a lack of qualified school leaders to fill principal positions. School systems across the country are reporting principal vacancies and many are finding a lack of qualified candidates. A survey conducted by Educational Research Services (ERS) in 1998 revealed half of the school districts in the country cited a shortage of qualified candidates for principal vacancies. This situation is expected to get worse as the U.S. Department of Labor estimates that 40% of the nation's principals will retire over the next several years (Tracy & Weaver, 2000, p.2).

As the need for principal candidates has continued to grow, so, too, have the expectations of the position. A consistent theme in the professional literature reflects the evolving role of the principal (Baker, 1997; Cohan, 1998; Downs, 1999; Lashway, 2003). This evolution of the position, along with a concern over having a sufficient supply of quality candidates, has resulted in growing concerns over traditional principal preparation. Concerns with university preparation (Anderson, 1991; Lauder, 2000) have led school systems to provide additional training programs to "grow" future principals.

This study is the development of a critical evaluation of one school system's efforts to develop and implement a "grow your own" program to address issues of quality and quantity of principal candidates. The study served two purposes:

- To determine the extent to which this system's efforts design and implement a program that supplements university preparation have succeeded in meeting program goals (product), efficient use of system resources (input), and appropriate selection of program activities (process). The context in which this program was developed was also examined.
- 2. To determine what can be learned from these efforts that might be beneficial to this system as it attempts to improve the program, and what can be learned that might prove useful to other systems coping with the need to address principal shortages.

Chapter IV presents the results of data analysis aimed at answering the following research questions:

- 1. What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program?
- 2. Which program elements were created specifically to address those forces impacting the principals?
- 3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program?
- 4. What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants?
- 5. What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of

- the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role?
- 6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement?

The research design for this study employed both qualitative and quantitative methodologies and was based on Daniel Stufflebeam's conceptual model for professional evaluation known as CIPP. Data were gathered to address the four components of the CIPP model as defined by the six research questions. The findings from this collection of data will be presented under the components of context, input, process, and product. The first two research questions dealt with context, the third question dealt with input, question four dealt with process, while questions five and six dealt with product.

The main source for the collection of quantitative data was a survey distributed to the 99 participants of the past three cohorts of the county's Aspiring Leadership Program who had completed the program as well as the 12 original members of the design team. Cohort group four was not included in the study, as it had only completed Phase I. Through the assistance of Washington Policy Studies, an educational research firm located in Washington, D.C., the survey was distributed electronically to the 99 program participants and the 12 design team members. A descriptive cover letter, letter of consent, and some initial information about the survey were e-mailed to all participants preceding the e-mailing of the actual survey.

The survey was e-mailed to all participants and members of the design team at the start of September 2004. A copy of the survey is included in Appendix A. The researcher's goal for response was 70%.

By the third week of September 2004, the response rate had reached 52%, and the decision was made to send a second request letter to the non-respondents that stated the need for their responses; a second copy of the survey was attached to the email. A copy of the second request letter is included in Appendix D.

Of the 99 surveys e-mailed to program participants, 85 subjects returned responses. Of the 12 surveys e-mailed to design team members, 10 subjects responded. A total of 95 usable responses were received and analyzed for a final total response rate of 86%.

The majority of qualitative data for this study was collected from interviews with key school system leaders, focus group interviews, and a review of records.

These data were analyzed and sorted by topics, clusters, and patterns in an effort to answer the research questions.

Question 1: What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program?

The following forces impacting the principalship were referenced frequently in focus group interviews, a record review, and individual interviews with key school system leaders, including the superintendent, associate superintendent, assistant superintendent of instruction, assistant superintendent of human resources, and director of staff development. An analysis of the data revealed the following six major forces that led to the formation of an Aspiring Leaders Program in the system being studied:

- A reorganization of the school system with a renewed emphasis on the principalship.
- A critical shortage of qualified principal candidates to meet anticipated needs.

- Concerns over the changing role of the principal, especially in the area of accountability.
- Concerns over traditional university preparation of principals.
- New licensure requirements for principals.
- Addressing minority achievement and a lack of minority candidates for the principalship.

Reorganization of the School System with an

Emphasis on the Principalship

The current superintendent of the school system being studied was appointed in May 1996. The new superintendent faced a number of issues confronting the school system, including declining test scores on national and state assessments, a widening gap in the achievement of minority students, severe issues of staff morale, and a significant deficit or deficiency in the system's operating budget. As the new superintendent began to assess the needs of the school system, he drew on his experiences as a principal over a twenty-year period. One of the first things that the superintendent did was to reorganize the Central Office, putting together an executive staff composed almost entirely of individuals who had been former principals. "That was such an important decision to surround myself with top-notch individuals who had demonstrated a standard of excellence in the principalship. As we looked at how to move the system forward, we reflected on our experiences as principals—we all have a tremendous faith in the [ability of the] principal to make a difference for students and instruction."

Table 4 shows the heavy emphasis the superintendent placed on having individuals who were former principals on his executive team. Three of these

executive staff members came directly into their positions from the principalship without prior Central Office experience.

Table 4

Reorganization of School System Executive Staff

Position	Principal Experience	Experience Level
Superintendent	Yes	High School
Associate Superintendent	Yes	High School
Assistant Superintendent of Instruction	Yes	Middle and High School
Assistant Superintendent of Human Resources	Yes	Elementary School
Assistant Superintendent of Facilities	No	Vice Principal – High School
Assistant in Administration	Yes	Middle School
Assistant in Administration	Yes	High School

Having been principals, the individuals interviewed for this study acknowledged that their experiences in that position led them to conclude that refocusing on the principalship was key to school system improvement. A key school system leader stated,

As we talked early on about the school system and what needed to be addressed, we kept coming back to the idea that having an outstanding principal at every school was critically important. We felt like we should invest in the principal position. If we could get good people and keep good people, we would really see positive results. We felt that the position had been devalued over the past several years and that needed to change.

It did not take long for the superintendent and his staff to demonstrate their renewed interest in the principalship. Current principals began to be invited to meet with the superintendent and his staff to set school and school system goals.

Principals were able to articulate needs and concerns for the superintendent and executive staff to consider. Issues raised by the principals were given immediate consideration. The school system budget was reallocated to provide every elementary principal with a vice principal since these schools were operating without one. Instructional leadership teams, consisting of an instructional specialist, a reading resource teacher, and a resource teacher for the gifted and talented were added to every elementary and middle school to support the principal and to assist in his/her role as instructional leader in his/her school improvement efforts. Each high school principal was given a resource teacher and a full time athletic director to allow them to spend more time on issues in the classroom. One member of the design team, who was a principal, stated, "We were being given an opportunity to provide input and obtain resources we had been requesting for a long time. A lot of superintendents talk about how important principals are—this superintendent was showing it."

Key staff interviewed for this study stated that the renewed interest in the principalship could be seen in one other area—salary. As the superintendent made it part of the goal to attract and retain the best available principals, there was agreement that salaries needed to increase. In 1997, the school system ranked 17th out of 24 districts in principal salary. In 2004, this same system ranked 4th.

The superintendent's position on principal salary reflected how valuable he felt the position was to school system improvement. "If you want quality, you have to be prepared to pay for it, but I knew that was not enough. As I looked at where we were with our current principals and where we needed to be, I was very concerned." He and the executive staff acknowledged that some current principals in the system would

need to be replaced and that finding quality candidates to fill those vacancies would be of critical importance.

A Critical Shortage of Qualified Principal Candidates

to Meet Anticipated Needs

As the superintendent and his executive staff began to refocus on the principalship, they noted that a significant turnover of principals had started to occur within the system. Table 5 shows the number of principals leaving their positions in the school system over a ten-year period. The total, 18, represented 58% of the system's principals.

Table 5

Number of Principals Leaving the Position, 1990-2000

Level	1990-1995	1996-2000
Elementary (19)*	3	5
Middle (7)*	1	3
High (5)*	1	5
Total	5	13

^{*} Indicates the total number of principals within the system at that level.

A review of system records indicated several reasons for principals leaving their positions, including retirement and going to other school systems. Some principals were demoted or reassigned by the superintendent after a review of their performances and after appropriate personnel steps had been followed. Said one system leader, "I think if you look over our current group of principals, you will find the majority of them have been appointed during this superintendent's time in charge of the school system." Table 6 reflects the accuracy of this statement.

Table 6

<u>Current Principal Tenure</u>

Level	1-5 Years	6-10 Years	10+ Years
Elementary (19)*	10	5	4
Middle (7)*	5	0	2
High (5)*	2	1	3
Total	17	6	9

^{*} Indicates the total number of principals within the system at that level.

Of the school system's 31 principals, 22 (71%) have been appointed under the current superintendent. Only nine (29%) have more than ten years of experience in the principalship.

Knowing that a number of principals within the school system would soon need to be replaced, the issue for the superintendent and his executive staff was to ascertain if a sufficient supply of qualified candidates was available for replacement. All members of the superintendent's executive staff interviewed for this study indicated that lack of quality candidates was the main reason for implementing a "grow your own" principal preparation program. The superintendent, referencing the sentiment of many on the executive staff, stated,

If you want top-notch leadership, you have to invest in it; you have to create it. Simply put, we were not getting really top administrators from other places. Rarely today do you get good people from the outside. We are now, to a limited extent, but prior to the program that was not the case. So as we looked inside the school system, we were concerned [that] we did not have enough candidates prepared to become excellent principals.

Another system leader echoed these sentiments. He stated, "I remember the last high school principal opening we had a few years ago. We just could not fill it. We had no internal candidates, and despite advertising in all the traditional places, we

could not get a quality candidate from outside the school system. I was really worried about starting school and not being able to fill this incredibly important position."

System records indicate the severity of the issue prior to the program. Table 7 contrasts the number of individuals applying for vice principalships and principalships before and after the program. Table 7 also shows how many of these applicants demonstrated the qualifications to be placed in the pool of acceptable candidates. As Table 7 indicates, during the four years that the program has been in place, the number of applicants and the number deemed "qualified" has nearly doubled.

Table 7

Number of Candidates Applying for Administrative Positions Before and After the

Principal Preparation Program

	1	999	2004		
	Number Applied	Qualified for Position	Number Applied	Qualified for Position	
Vice Principal Candidates	13	8	24	18	
Principal Candidates	8	5	12	9	

The change reflects applicants qualified for positions regardless of the number of vacancies. By 2004, the number of qualified candidates for vice principalships went from eight in 1999 to 18 in 2004, representing a 56% increase. The number of qualified principals candidates rose from five in 1999 to 9 in 2004, representing a 44% increase. Of the 18 qualified candidates for vice principalships in 2004, the assistant superintendent of human resources confirmed that 66% had participated in the Aspiring Leadership Program. He also confirmed that two of the candidates for principalships had completed the program.

Concerns Over the Changing Role of the Principal

In An Era of Accountability

When the superintendent convened a committee of school stakeholders in 1999 to address issues of leadership within the school system, the shortage of principal candidates and the evolution of that position were primary topics of discussion. "I remember at our first meeting, and really at all subsequent meetings leading up to the formation of the program, we talked about changes in the job of the principal, especially with the tremendous accountability placed on the position," stated a school system leader.

As the superintendent of schools began his reorganization of the system, he was able to solicit additional funds from the local county commissioners in exchange for agreeing to benchmarks. The vast majority of these benchmarks were academically focused on state tests and national tests. The superintendent stated, "Agreeing to these benchmarks was vital to our improvement efforts, but we knew the pressure it would place on the schools. That's why we needed excellent principals who were prepared to get results."

In focus group interviews, the design team members recalled how quickly the position of the principalship was changing due to increased accountability. The state in which the study took place was changing its entire assessment measures to new tests aligned with a new curriculum. The local Board of Education was implementing "end of course assessments" as well as other measures that included performing financial audits of school fiscal decisions. President Bush had just been elected and federal standards were looming, eventually becoming the "No Child Left Behind" legislation.

Rather than view this movement negatively, system leaders saw the increased accountability as a way to attract a particular type of individual to the principalship.

As the superintendent used accountability to obtain an increase in the budget that gave us additional resources to show what we could do with them knowing that with better results we could obtain even more resources. We knew we would be putting more on the principals, so we needed to prepare them for that. The program has allowed us to better prepare future school leaders for the tremendous accountability they now face because the bottom line is harsh but true—if they don't perform, then they'll have to understand that they are not going to be in their positions for long.

The superintendent and other school leaders both articulated that the Aspiring Principal Program was an excellent vehicle to address the new responsibilities being placed on principals at all levels.

Concerns Over Traditional University Preparation of Principals

All key school system leaders interviewed for this study concurred that part of the impetus for creating a program dealing with leadership preparation was due to concerns over traditional university preparation. One leader stated, "I think the university sees their work as to provide the theory, but I wish the university would also see the problems that principals face today. While research and theory are certainly important, I think sometimes, unfortunately, that is what the university stresses, and the practical gets lost. We had an opportunity to address this by creating our own program." Other leaders expressed concerns that current university preparation fails to address the role of instructional leader that principals now must be ready to assume.

I think there is a lot of value in university programs, but few colleges and universities really stress how an aspiring principal gets ready to become the instructional leader of the school. You get systems theory, organizational skills, and things of that nature, but not a lot of instruction on how to use data to drive decisions or how to model instructional strategies to your staff.

Several members of the design team indicated that their own issues with university programs contributed to creating a "grow your own" program in the county. A principal on the design team stated, "I felt a tremendous disconnect between what I was taught during my university preparation and what I had to deal with as a principal. You get a great deal of theory, but very little practical application and real life issues principals face. I think that's a problem." The superintendent echoed some of these concerns, but also pointed out,

I feel very strongly that there needs to be a blend between what is happening in the university and what school systems are doing to prepare future leaders. As our program came about, I have never seen it as replacing higher education training; I think it's a way to supplement it. Our future principals need and desire the best colleges and school systems can provide to prepare them for the job they face.

A "grow your own" program was seen as an excellent way to blend what aspiring leaders were learning at the university level and providing the experience they needed to be future principals.

Addressing New Licensure Requirements for Principals

Another force contributing to the development of a principal preparation program was new licensure requirements to become a principal. The Director of Professional Development stated, "As we began to discuss the creation of a program, I think we all felt that a real benefit to our future leaders would be to design a program

that would allow aspiring principals to pass the new licensure exam." Aspiring principals in this state had previously been required to participate in an assessment center that focused on designated principal skills. The new licensure exam was framed around the ISLLC Standards and was seen as a more rigorous process.

Key school system leaders indicated that there had been little system support for helping aspiring leaders with the old process and there was almost nothing in place to assist principal candidates with the new program. A system leader remarked,

When you are already concerned about not having qualified candidates for your administrative openings, you don't want to compound the problem by having the candidates that you select not able to pass the licensure test. A preparation program seemed to be a natural response to help our aspiring leaders.

Addressing Minority Achievement and a Lack of
Minority Candidates for the Principalship

Early in his tenure as the leader of the school system, the superintendent made minority achievement and eliminating the achievement gap between student groups one of his highest priorities. A system focus quickly became addressing minority achievement and diversifying the work force, particularly among the county's administrative positions.

Not only did we have a lack of all candidates, but we really had an issue with no minority candidates. My highest priority was increasing the number of African American candidates. I wanted to make sure we were giving them all of the opportunities in the world to really develop their skills because I wanted them involved in the principalship.

Another system leader was aware of this issue and expressed concern over how to address it. "As we did an assessment of our principals and vice principals, we were not satisfied with the diversity, but how to address this was a real issue."

Table 8

Minority Principals

	1996	2004
Elementary	4	5
Middle	2	3
High	1	1
Total	7	9

Table 8 shows that in 1996, there were seven minority principals in the school system, accounting for 22% of all principals. In 2004, there are nine minority principals, accounting for 28% of all the principals in the county. According to the associate superintendent, "Under this superintendent, this is a school system that prides itself on not just talking about minority achievement, but also seeing a principal preparation program as a way to increase the number of minority candidates in our school system. Are we happy where we are now? Absolutely not, but I think it's worth noting that we had three principal vacancies this year and all three were filled by minority candidates." Two of these three new principals were involved in the principal preparation program.

<u>Question 2</u>: Which program elements were created specifically to address those forces impacting the principals?

The following program elements were referenced frequently in focus groups from a review of documents and from interviews with key school system leaders, including the superintendent, associate superintendent, assistant superintendent of instruction, assistant superintendent of human resources, and director of staff

development. An analysis of the data revealed the following four program elements created specifically to address the forces cited in Research Question 1:

- Making the program unique to the county being studied
- Making the program hands-on and practical
- Framing the program around the ISLLC Standards and identified system themes of leadership
- Making the program free of charge, inclusive, available for college credit, and convenient in terms of place and schedule

Make the Program Unique to the School System

The prospect of a significant deficiency in the number of quality candidates, especially minority candidates, for principalships was one of the main reasons the school system decided to implement a preparation program. The superintendent stated,

I knew that even by increasing salaries, we were not going to be able to attract enough quality candidates for leadership positions. With the critical shortage of leadership around the country, I was convinced we needed a top leadership program right here in the county. I wanted a program that was unique to this county. A "homegrown" program would not only encourage more people to pursue principalships here, but once they attained those positions, they would be better prepared for success.

Key school system leaders, such as the superintendent, were used as keynote presenters, usually followed by state and national experts. Table 9 shows the various school stakeholders (other than principals) who contributed local flavor to the program.

Table 9

<u>Variety of Local School Stakeholders Who Have Presented During the Leadership</u>

Program, 2001

	Summer Program	Monthly Seminars
Superintendent Executive Staff	7	5
Central Office	4	8
Board of Education	2	1
Teachers' Association	5	6
Teacher Leaders	1	0
Local Politicians	3	2
Total	22	22

The superintendent offered,

My participation was necessary to provide the vision for the school system. If you believe that the Central Office exists for the schools, then it needed to be part of this training program. I also asked the committee to make sure our political leaders, such as the board chairperson and the president of the county commissioners, were included. They are an important part of our school system and I thought it would be good for participants to see these key people, to know we are a team and that we don't just go to them when we need a handout.

Another leader said,

I think the most successful part of the program was bringing together our local leaders to address future leaders about education. Through the summer programs and seminars, these people got to hear county commissioners talk about the local community; Central Office staff talk about system expectations; principals talk about instruction and issues

they face in their schools in this county. How do you beat that type of experience?

An agenda from a meeting of the Leadership Development Committee dated May 2001 listed, "Infuse local items into program." The design team revealed that every activity and session was infused with items and issues unique to the county. For example, when an activity asked for participants to discuss observations and evaluation of staff, the school system forms for these duties were covered and utilized in simulations.

Current principals identified topics for sessions and worked to design sessions that reflected system concerns. For example, a monthly seminar on budgeting and finance utilized the county financial manual, financial programs, and was conducted by a principal who had received a perfect financial audit.

As efforts continued to make the program reflect the direct needs and issues of the topic being studied, the direct participation of current principals was vital to the effort. A system leader stated,

Because of the participation of our principals and other school stakeholders, I feel we have offered something unique in preparation of future leaders. This program let us 'grow our own' by allowing us to look within the system. These candidates now have more familiarity with the system and allow us to provide a growth opportunity to help people in the profession. More than anything, they should be more ready to be successful principals here.

Table 10

Participation of Principals in the Program, 2001

School Level	Summer Program	Monthly Seminars	Diagnostic Interview
Elementary (19)*	11	9	10
Middle (7)*	6	7	6
High (5)*	5	5	5
Total	22	23	21

^{*} Indicates the total number of principals within the county at that level.

During the first cohort of the program, every middle and high school principal participated in an activity, and all but two of the elementary principals assisted at some point. The superintendent remarked, "It just seemed natural that as we began to refocus our efforts on principals it would be the principals who would lead the way with this training."

Making the Program Hands-On and Practical

Just as there was a concentrated effort to make the program distinctive and unique to the school system, there was an equally intensive effort "to make the program hands-on and extremely practical" according to the design team. One individual commented, "Nobody wanted this training to be lecture or one-way from us to them. If it was going to be valuable, it had to afford the participants the opportunity to 'get their hands dirty' and be active." Several of the members of the design team indicated that efforts to focus on the more practical aspects of the principalship were in direct contradiction to the more theoretical focus of their university preparation.

Design team members indicated that topics were aligned to what principals were actually doing in their schools. A design team member commented, "The strength of this program was its practicality; the in-basket activities, the presentations by principals. It's what aspiring principals needed to hear in addition to the theory of the university courses."

As the program was being formulated, the design team was careful to monitor the proposed activities for a balance of practical with theoretical and presentations with active learning. The design team planned many activities such as simulations, mock interviews, and written responses to school-based scenarios to make sure participants got to actively make decisions using the information they were receiving. Participants received data for a fictitious school that they would use as they went through the summer program and monthly seminars. Ultimately, they would be able to complete a portfolio from what they had experienced in the program. According to a system leader,

The theory is wonderful, but a program like this allows you to prepare people for what they will face every day. Application and process are whole different pieces of work. We had to help aspiring leaders figure this job out; how do you move from the theory of the classroom into the practice of getting results? How do you figure that out? How do you make it happen? How do you provide the tools? This program allows us to begin to answer these questions.

Framing the Program Around the ISLLC Standards and Identified System Themes of Leadership

As members of the design team began to suggest a principal preparation program to address many of the issues being considered about leadership in the county, one of the fundamental questions became what should serve as the foundation or focus of the training. "I just remember we were looking at so many things—surveys from aspiring leaders, research articles, national standards on leadership—wow! At the same time, we were trying to remember what the superintendent wanted us to do; it was not an easy undertaking," said one member of the design team.

Another member of the design team commented, "From our dialogues with the

superintendent, Central Office, and principals, there were certain themes that kept coming up such as vision, accountable leadership, instructional leadership, and community involvement. There was a real system expectation to include these in the program, but we all felt that an effective program needed to be based on some sort of national standards. It felt like we were stuck in neutral for a long time."

The director of professional development recalled that it was when the committee aligned the recurring themes on leadership with the ISLLC Standards that things began to move forward. She stated, "We really respected the ISLLC Standards and because they were the basis for the new licensure test, we knew we needed to use them. At one meeting, we did a match with the ISLLC Standards and saw them as mutually compatible. It really was another example of our efforts to make this program a blend of national, state, and local standards."

Table 11 presents an agenda item for a meeting of the Leadership Development Committee for January 2000, showing a proposed match of the suggestion system leadership themes and the ISLLC Standards:

Table 11

Match of System Themes and the ISLLC Standards

System Themes	ISLLC Standards		
Vision	Standard 1		
Accountability	Standard 3, Standard 5		
Instruction	Standard 2		
Community Relations	Standard 4, Standard 6		
Using Resources Wisely	Standard 4		

Members of the design team indicated the match between the ISLLC Standards and identified system themes provided a foundation around which to build the program.

The first phase of the leadership program was a five-day training session.

These summer sessions were scheduled and devoted to each of the major system

themes. Activities were incorporated that came directly from selected ISLLC training modules. For example, a review of system records highlights the second day of the summer institute with a focus on the system theme of "instructional leadership." The assistant superintendent of instruction and a professor from a local institution of higher education were the featured speakers, presenting information regarding various ways to differentiate learning in the classroom. A reflective activity, taken directly from the ISLLC training manual, was given to participants.

Key school system leaders seemed to concur that basing the program around the system's core themes of leadership and the ISLLC Standards provided a program that would help aspiring principals to be better prepared for the tremendous demands of the position, particularly in accountability, while also better preparing them for the licensure exam. In the focus group interviews with participants of the program who have attained principalships, the consensus was these elements of the program were essential to its success. One principal, who was a former program participant, stated,

There is no doubt that the program helped me be more prepared for both the licensure test and my new position. During the exam, and all throughout my first year, I found myself remembering things we had been taught in the program and that I am now able to use.

Making the Program Free of Charge, Inclusive, and Available for College Credit

Because one of the main reasons for creating an Aspiring Leaders Program was to attract a great number of individuals, particularly minority candidates, into administration, one of the key elements decided early by the design team and system leaders was to make the program free of charge for all participants. Other leadership programs reviewed by the design committee had included some charges to participants to defray operating expenses to the system. One consultant to the design team not

only promoted charging a fee to keep costs down, but also argued it was a way to separate those individuals who might be truly willing to commit to intensive leadership training. Some of the research on effective principal preparation promoted being selective in program participants to ensure only the highest quality candidate is being considered.

Design team members responded that the idea of charging for this training and being selective was discussed, but it was ultimately decided to make the program free of charge and inclusive to anyone who expressed interest in leadership. One member of the design team stated,

Because we had so few people wanting to pursue leadership, particularly the principalship, we really felt that by making the program available to everyone and with no fees, we might attract some candidates who had not considered being a principal previously. Even if a person decided that after going through the program they were not interested in pursuing a principalship, they would probably have a new appreciation for the position.

The superintendent confirmed that making the program free and inclusive was a large part of his system improvement plans. "I felt it was crucial to not limit access to this program. Yes, it came with a high price tag, but it was an investment for the system. Also, by being inclusive rather than exclusive, I really thought we had an opportunity to increase the number of minority candidates in administration."

Actively recruiting minority participants into the program became a system expectation for principals. One member of the design team, who was a principal, commented, "We were asked to tell all staff members about the program, but to particularly look to engage minority candidates. It was an easy sell because it was free and looked to be a quality professional development opportunity."

To further encourage participation in the program, individuals could obtain college-level credit for the program. School system leaders designed a partnership with a local university so that the program would not only afford participants credits toward their Administration and Supervision (A&S) certification, but also give them an opportunity to become involved in advanced graduate study. This partnership resulted in a new program called CASE. The Certificate of Advanced Study in Education in Instructional Leadership was a 30-credit program for experienced educators who have completed a master's degree. It was offered through a local college and equivalent to the Education Specialist (EdSp) certification that meets the minimum requirement/standard for a superintendent's endorsement. In the system being studied, the CASE program was initially designed to be a precursor for a future doctorate program (Ph.D.) in Instructional Leadership. The first cohort enrolled in this program were on track to graduate in May 2003. Those who completed CASE paid \$390 per course without the possibility of tuition reimbursement. In addition, the county paid an additional \$6000 for each course. This agreement applied to cohort II, which is scheduled for completion in May 2005.

Program participants confirmed that being able to obtain college credit was a powerful inducement to their participation in the program. "I probably would have done it anyway, but the opportunity to become involved in a program that would lead to getting my doctorate made all of the difference in the world," stated one individual.

Question 3: What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program?

The following resources were referenced frequently in focus group interviews with the design team, a review of system records, and interviews with key school system leaders, including the superintendent, associate superintendent, assistant superintendent of instruction, assistant superintendent of human resources, and

director of staff development. Resources made available for the design and implementation of the principal preparation program fell into two primary categories—human and financial.

Human Resources

In interviews with key school system leaders and members of the design team, there seemed to be a great degree of consensus that the human resources made available to this program were the key to its design and implementation. When the director of staff development was given the task of putting together a committee to focus on leadership, primarily at the principal level, a diverse group of system leaders were brought together. Table 12 shows the make-up of the initial design team.

Table 12

Composition of Leadership Advisory Committee

	White Male	White Female	Black Male	Black Female	Other Ethnic Group Male	Other Ethnic Group Female	Total
Central Office	2	3	1	0	1	0	7
Principal	2	1	1	1	0	0	5
Vice Principal	0	3	0	3	0	0	6
Teacher Leader	0	1	0	0	0	0	1
Teacher	1	1	0	1	0	0	3
Other	1	2	0	0	0	0	3
Total	6	11	2	5	1	0	25

System records indicated that the committee has been chaired by two awardwinning principals in the system. Vice principals and teachers who were selected for the committee were considered to be excellent staff members who had expressed interest in becoming principals. Four members of the design team had been trained as assessors for the principal preparation program previously offered by the state in which the study was conducted.

The first meeting of this committee was held in May 1999. The committee met once a month, except in July and August, from that point until implementation of the program in June 2001. Including several supplemental meetings and work sessions, every member of the design team committed over 60 hours to the program's creation. One member of the design team stated, "When you looked around the table and saw some of the busiest people in the school system, including the associate superintendent and principals, you just had a feeling we were getting the opportunity to contribute to something really important."

There were a number of other human resources utilized in the design of the system's Aspiring Leaders Program. A consultant on leadership development was brought in from the U.S. Department of Education. An agenda from a meeting of the Leadership Development Advisory Committee identified this individual as having "a quantity of differentiated experiences in education with particular expertise in leadership development. Her purpose in joining the group [was] to serve in an advisory capacity to the design team in its development of a local leader preparation program."

Additionally, several individuals from the State Department of Education's leadership division were included as mentors to the project's development. Finally, two consultants from Broward County, Florida, were made available to members of the design group for three days in November of 2000. These consultants had developed a principal preparation program in their school system and were contracted to work with the design team to finalize preparations for the program.

As the program was implemented, a tremendous amount of human resources were needed. Each day of the summer program required no less than three key school system leaders and the contributions of no less than three to four system principals. Several program participants commented about the human resources made available. One program participant indicated, "I got so much out of hearing from local system leaders and the principals. I heard first-hand testimonials about leadership and the principalship that I would not be able to get anywhere else. It also made an impression on me that the associate superintendent was there every day."

Members of the Central Office, principals, local politicians, and educational experts from across the state in which the system is located were involved in program implementation. Said one school system leader, "Of course we had to hire the consultants, but the contributions from system leaders within the county and our principals, while free, were invaluable. That's a great combination to make a 'grow your own' principal preparation program work."

Financial Resources

While a great deal of the resources for the program were free of direct costs from system leaders and principals, designing and implementing a program of this magnitude required a lot of financial support. To assist in program design and implementation, the director of professional development applied for, and received, a grant from the State Board of Education made available from a federal grant on leadership development. This grant was for \$5,000 a year and was available for a three-year period. Table 13 shows the proposed spending allocations of the grant each year.

Table 13

Grant Application

Area of Expense	Amount Allocated		
Salaries and Wages	\$500		
Contracted Services	\$3000		
Supplies and Materials	\$1500		
Equipment	0		
Other	0		
Total	\$5000		

After receiving the grant (Appendix E), the superintendent of schools utilized the system's operating budget to fund other resources to support the program. He stated, "Early in my tenure as superintendent, I was able to get the county commissioners to allow us to keep our fund balance. With this money, I wanted to invest in long-term issues that would benefit the school system. I considered providing the money for a local principal preparation program as an outstanding investment in our school system."

System records indicated that a great deal of money was spent on program design. While some records had not been kept from early in program development, estimates are that over \$25,000 was spent to create the program, not including salary of system staff such as the director of professional development and design team members. This money was spent on hiring consultants, visits to various leadership programs in and out of state, supplies, and other materials.

Once program implementation began, the greatest outlay of funds was for the summer institute. Table 14 shows a breakdown of monies spent for the first cohort in 2001.

Table 14
Financial Breakdown for Summer Institute

Consultants	\$8912
Room and Food	\$9273
Materials and Books	\$2053
Total	\$20,238

Consultants for the program included college professors in educational administration, representatives from organizations like National Association of Secondary School Principals (NASSP), and leadership organizations running programs such as "Who Moved My Cheese?" Each participant was provided with a binder full of research articles on the principalship and leadership. Books were also provided, including *Organizational Vision*, *The 7 Habits of Highly Effective People*, and copies of *Leadership Magazine*.

While the monthly seminars were held at the local Board of Education, the summer program was held at a first-class facility in the county. Breakfast and lunch were provided to participants each day. One member of the design team remarked, "The superintendent wanted individuals in the program to feel special and treated like professionals. Having the training outside of a system-owned facility sent a very strong message."

Many program participants agreed with this assessment. "Having the program at this location, the food each day—everything was just first class. I was blown away by the binder and books. All of these things not only got me excited about being a principal, it got me excited about being a principal in [this] county."

Finally, another large expense assumed under the program was funding participants who were going through the CASE program. Not only did the school system pay \$6000 per course for program participants, it absorbed the costs for facility use as the classes were taught at a local high school. While records were not made

available for this expense, the director of professional development estimated that the total cost to the school system was in excess of \$60,000 for facilities, tuition, and program materials.

Question 4: What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants?

The following elements and processes were referenced frequently in focus group interviews, record reviews, and individual interviews with key school system leaders including the superintendent, associate superintendent, assistant superintendent of instruction, assistant superintendent of human resources, and director of staff development. An analysis of the data revealed several significant sources that led to the designed principal preparation program. The analysis also revealed a great deal about the process of formulating the program and recruiting participants into this training.

Program Sources and the Process of Designing the Program

When asked what was the most important source that led to the creation of a program to train principals in the system, the following items were referenced frequently:

- ISLLC Standards
- Experiences of system personnel (Central Office and school-based administrators)
- Models from other school systems
- A State Department of Education report called "Clearing the Plate"
- State Assessment Center previously used to certify principals

- Consultant from the U.S. Department of Education
- Current research about principal preparation

The ISLLC Standards were by far the most frequently referenced program source in the development of the program. When asked if any other standards in leadership were considered for this program, key school system leaders and members of the design team indicated that others had been reviewed, but no one could recall any that were significant. A review of system records reveals the ISLLC Standards were incorporated into the original grant to attain funding for this program. In a summary of notes from January 4, 2001, the consultant from the U.S. Department of Education stated, "I want to work with you to develop a program that links your principal preparation program to the ISLLC Standards. This will not only give your program credibility, but also help participants be prepared for the licensure test."

Several members of the design committee recalled that the ISLLC Standards consistently showed up in research articles on the principalship and as the basis for program formation in other schools systems. Stated one member of the design team,

As we read articles on the principalship from ERIC searches and listened to the two-day presentation from the folks from Broward County, Florida, we kept coming across the ISLLC Standards.

Gradually, after reviewing the ISLLC Standards again, things began to fall into place—the relationship among the various documents, articles, and research studies began to come together. We began to see that among all the information sources, the ISLLC Standards were truly the guiding light to the program.

After settling on the ISLLC Standards as the foundation for the program, design team members matched program elements that had come from several key system leaders, current principals, and aspiring principals. As was introduced in

Question 2, design team members drew on the experiences of system administrators to create a program truly unique to the school system being studied. A document that reflects that blending of the ISLLC Standards and system ideas of leadership can be found in Appendix F.

Having the opportunity to visit and review several existing models of principal preparation was also key to the design team's efforts. Several local programs have been developed in the state in which the study is being conducted so many members of the design team were able to visit these training programs for principals and talk directly to individuals responsible for their creation. One member of the design team stated, "Before those visits, I had never even heard of a 'home grown' principal training program. After those visits, we got an opportunity to work with the consultants from Florida, and I really got a sense and direction for what we needed to do."

The work sessions with the Florida group were frequently cited as a key to implementation. The consultants shared the history of their program development, its structure, materials used for training, and reviewed some initial data from their first several cohorts. Two members of the design team visited this program and were able to provide insight into how the training could be replicated in the school system being studied.

As the program began to take shape early in 2001, the most influential source for its completion was using materials and activities from the now-defunct State Principal Assessment Center that many members of the design team were familiar with as assessors or participants. This assessment center had been a requirement for aspiring leaders to obtain their Administrator II certification needed to become a principal in the state being studied. It had been a two-day intensive assessment requiring participants to display the essential skills needed to be an effective principal.

Several weeks after completion, participants would receive a written diagnostic report articulating particular areas of strength or areas to improve. Many of the design team felt strongly that aspects of the assessment center would work well in the program they were developing. One member stated, "We all felt like the old assessment center had been really effective so we borrowed very liberally from it. The in-basket activities, the interviews with written diagnostics, and a lot of the skills we would focus on came from that program."

Finalizing the Process and Initiating the Program

"Looking back on it now, the process of creating this program was really remarkable and something I have been proud to have been a part of," stated one design team member. Over two years elapsed from the time the committee was first formed until program implementation. Several members of the design team expressed frustration over the slow pace of the process. Members indicated that the decision to use the ISLLC Standards and the visit from the Florida consultants really got the process moving. A small sub-committee met over a two-week period to draft a plan for the initial program. Early in March 2001, the committee met to review the proposed plan. After implementing some changes, the committee voted to endorse the Aspiring Leadership Program. The plan was sent to the superintendent and his executive staff for review. At the same time, the program was shared with system principals for feedback. One school system leader commented,

At this point, I got very excited about what was being proposed.

Watching the reaction of the executive staff and the principals, I knew we had something viable. Not only were there very few suggestions about the proposed plan, but individuals indicated a sense of validation as they saw many of their recommendations incorporated into the

program. We had worked hard to keep executive staff and principals involved throughout the process, and I think that's why they were so eager to participate. They felt a sense of ownership in the program.

After endorsements from the superintendent, executive staff, and principals, the Aspiring Leadership Program was adopted by the system. Key components of the program included:

- Phase I: A five-day summer institute. This was required for individuals who wanted to participate in the larger program. The institute objective aligned with the ISLLC Standards and incorporated system themes of vision, accountability, instruction, community involvement, and resource use.
- Phase II: Seminars throughout the year. Participants attended seminars to complete Phase II of the program. Topics included: oral communication, written communication, due process for teachers and support staff, budget and finance, effective delegating, facilities, special education issues, developing professional portfolios, interviewing skills, current trends, High School Assessment, multi-tasking, and technology use.
- Phase III: Project to benefit the school system. Participants self-selected a project from a suggested list and gained prior approval before beginning.
- Phase IV: Mock interview. Participants had an opportunity for a diagnostic interview with written feedback.
- Phase V: Internship experience. Participants had an opportunity to gain administrative experience through an on-site internship. The purposes of the internship were to promote professional growth, improve

participants' resumes, and create a bank of human resources where principals could visit other settings, vice principals could move up into principalships, and aspiring vice principals could move into vice principal temporary placements.

Recruiting Participants

With the program set in place, the task became to promote the training to interested participants. As was indicated in Question 2, it had been decided that participation in the program would be open to any certificated staff interested in pursuing leadership training. The associate superintendent of schools was charged with encouraging principals to be active in recruiting individuals to participate in the program, with a particular emphasis on seeking minority candidates. When the program began in June 2001, 47 participants had enrolled.

The last item in the demographic section of the survey instrument used in this study asked the respondents "What motivated you to pursue the county leadership program?" Table 15 presents the options and the number of respondents for each one.

Table 15

Reasons That Motivated Respondents to Pursue the County Leadership Program

Number	Reason
29	Encouragement from a mentor
25	Enhance job opportunities as a principal
4	Required for building or district level position other than a principal
58	Broaden knowledge base
16	Broaden range of influence
45	Opportunity to use leadership skills
3	Prestige and status
9	Greater professional freedom
20	Ability to perform required tasks and functions of principal
20	Increased responsibility
29	Effect change on a greater scale
8	Other (please specify)

The most common responses to this question were "broaden knowledge base" (58), "opportunity to use leadership skills" (45), "encouragement from a mentor" (29), and "effect change on a greater scale" (29). Reasons that received the least number of responses included "prestige and status" (3) and "required for building or district level position other than a principal" (4).

These responses reflect the message that was sent throughout the county about participating in the Aspiring Leadership Program. It was advertised as a way for program participants to increase their knowledge base in leadership and to use that knowledge in a "hands-on," practical way. Program participants were told before, during, and after the training that this program was not required to advance in leadership nor would it afford anyone greater status or rank in obtaining future leadership positions.

Question 5: What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role?

In Chapter III, the researcher described efforts to modify an existing survey to collect the perceptions of the program participants and the design team members. An extensive search for a survey that would be suitable for this research project produced only a few possible surveys. The one selected, the Audit of Principals' Effectiveness, was developed by Jerry Valentine and Michael L. Bowman in 1986. The survey was designed to describe teachers' perceptions of principals' effectiveness. The researcher modified the survey after an extensive review of literature on the role of the principal. The Valentine-Bowman survey reflected many of the aspects of the ISLLC standards that had been used as the foundation for the creation of the "Grow Your Own Principal" program. The researcher retained the Audit of Principals' Effectiveness as a

three-domain survey of principals' skills. Domain one, consisting of 27 items, addressed organizational direction linkage and procedures. The second domain, Organizational Environment, covers teacher relations, student relations, interacting processes, and affective processes. This domain had 37 items. The third domain, Educational Programs, which entails Instructional Improvement, Curriculum Improvement, and Overall Effectiveness, was covered by 16 items. The revised instrument included the same number of items as the original, of which three items were changed and several others had one or more words changed to assure consistency throughout the instrument. A fourth section of the survey sought to collect demographic information and current job status as well as aspirations on the part of the participants and the design team.

In order to establish the validity of the instrument before its use, the revised instrument was administered to several principals and people who were aspiring to be principals in jurisdictions other than the one where the study was conducted to determine if they felt that it asked the kinds of questions that were relevant to the proposed study of principal leadership. This was done to establish the content and construct validity of the instrument. The people who reviewed the survey suggested several changes. The researcher made these changes and submitted the final version to computer specialists in the school system, since the survey was to be conducted using the school system's computer e-mail capability.

The researcher had no information on the reliability of the instrument. It was decided to wait until the data had been collected prior to computing Cronbach alphas on each of the domains. These data are presented in the section on quantitative analysis.

In September 2004 the survey was e-mailed to the 99 program participants from the first three cohorts and members of the design team. The actual survey was

preceded by a descriptive cover letter and consent form. The cover letter contained background information about the survey. A log was kept of the individuals to whom the survey was e-mailed and addresses and mailing dates were noted. A follow-up letter and survey were sent to potential respondents when a reply was not received within two weeks. By the 3rd week in September, the response rate was 52%. The follow-up letter, along with a copy of the survey, was sent to non-respondents stating the need for their responses. Eighty-five participants returned responses. Ten of the 12 design team members returned responses. A total of 95 usable responses was received and analyzed, for a final response rate of 86%.

When the researcher received the data, the first set of statistics computed were Cronbach alphas. The Cronbach alpha is a measure of inter-item reliability, used when no pretest-posttest reliability is available. Because the researcher wanted to establish the reliability of the instrument, Cronbach alphas were computed for each of the three domains. For domain 1, Organizational Development, the Cronbach alpha was .87. For the second domain, Organizational Environment, the Cronbach alpha was .95. The third domain, Instructional Improvement, the Cronbach alpha was .95. All of these Cronbach alphas are classified as strong and all were statistically significant at the .001 level, thus confirming instrument reliability.

Next the researcher computed correlation coefficients on the three domains. These are displayed in Table 16. Interpreting these tables, the researcher used an established set of criteria to make judgments about the significance of the correlations. First, a level of .05 was used to identify those correlations that were statistically significant. Second, the correlations were judged in the following manner. If the correlation was between 0.0 and 0.30 . it was considered weak; if it was between 0.31 and 0.70, it was considered modest; if it was above 0.71, it was judged to be strong (Gliner & Morgan, 2000). The correlations presented in Table 16 indicate that they

are all in the modest range and all are statistically significant at the .001 level, since all of the correlations were in the .51 to .65 range. The correlations accounted for approximately 25% to 40% of the variance among the three domains. This indicated that there was a modest relationship among the domains while also indicating that each was assessing something unique to its domain.

Table 16

Correlation Coefficients for Domains One, Two, and Three

	Domain 1	Domain 2	Domain 3
Domain 1	1.00	.65	.51
	(73)	(63)	(68)
		P=.001***	P=.001***
Domain 2		1.00	.59
		(72)	(67)
			P=.001***
Domain 3			1.00
			(73)

P = < .05*; <.01**; <.001***

The survey asked for background data on the participants' gender, age, and race. Table 17 presents the data on gender, age, and race.

Table 17

Number and Percentage of Respondents by Gender, Age, and Race

	Number	Percentage
Gender		
Male	24	29
Female	59	71
Age		
< 35	23	28
36-45	28	34
46-55	27	32
56 >	5	6
Race		
African American	19	23
Asian	1	1
Caucasian	63	75
Native American	1	1

The first statistical analysis was done on domains one, two and three. The researcher compared the responses of cohorts 1, 2, and 3 on each of the three domains. As displayed in Table 18, the results of this statistical analysis show that there was no statistically significant difference among cohorts one, two, and three on domains one, two, and three.

Table 18

One-Way Analysis of Variance of Differences among Cohorts One, Two, and Three

on Domains One, Two, and Three

Domain One

		Sum of	Mean		
	df	Squares	Square	F	Sig.
Between Groups	2	10.85	5.42		
				.08	.92
Within Groups	69	4,636.47	67.20		
		Domain 7	Γwo		
		Sum of	Mean		
	df	Squares	Square	F	Sig.
Between Groups	2	439.36	219.68		
				1.06	.35
Within Groups	68	14,091.96	207.23		
		Domain T	Three		
		Sum of	Mean		
	df	Squares	Square	F	Sig.
Between Groups	2	89.90	44.95		
				.81	.45
Within Groups	74	4,068.62	55.22		

P = <.05*; <.01**; <.001***

In each of the domains, the last item was a summary question reflecting the entire domain. In domain one, this was question 27. For domain two, it was question 64, and for domain three, it was question 80. The results of this analysis are presented in Table 19. The results indicate that there was no statistically significant difference among the three cohorts on the respective summary questions.

Table 19

One-Way Analysis of Variance of Differences among Cohorts One, Two, and Three
on Summary Questions for Domains One, Two, and Three

Domain One - Item 27

	df	Sum of Squares	Mean Square	F	Sig.
Between Groups	2	1.53	.76		
				2.29	.11
Within Groups	81	27.04	.33		

Domain Two - Item 64

	df	Sum of Squares	Mean Square	F	Sig.
Between Groups	2	.59	.29		
				.73	.49
Within Groups	81	32.97	.41		

Domain Three - Item 80

	df	Sum of Squares	Mean Square	F	Sig.
Between Groups	2	.94	.47		
				1.02	.37
Within Groups	81	37.47	.46		

P = <.05*; <.01**; <.001***

Next, the researcher compared the responses of the total cohort, one, two, and three, to the design team on the same domains. The results of this analysis are presented in Table 20. They show that there was no statistically significant difference in the means for domains one and two. For domain three, there was a statistically significant difference favoring the design team, whose mean was 62.00. For the combined cohort groups, it was 57.68.

Table 20

Independent t-Test of the Total Cohorts' Responses versus the Design Team's

Responses to Domains One, Two, and Three

_	•	\sim	
Llom	2111	()+	ı۵
Dom	ıaııı	VI.	IC

-	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Total Cohort	72	96.15	8.09			
				.13	79	.95
Design Team	9	95.77	7.40			

Domain Two

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Total Cohort	71	129.58	14.81			
				.40	76	.69
Design Team	7	127.29	15.31			

Domain Three

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Total Cohort	77	57.69	7.41			<u>8</u> -
				3.23	23	.01**
Design Team	10	62.00	3.27			

P = <.05*; <.01**; <.001***

The researcher also compared the means of the summary questions at the end of each of the domains, questions 27, 64, and 80, comparing the total cohort versus the design team's responses. For items 27 and 64, there was no statistically significant difference, but for summary item 80 there was a statistically significant difference in which the design team again had higher means than did the total cohorts. The design team mean was 3.90; for the cohort, it was 3.42. Table 21 displays these results.

Table 21

Independent t-Test of the Total Cohorts' Responses versus the Design Team's

Responses to Summary Questions

Domain One - Item 27

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Total Cohort	84	3.57	.59			
				1.19	92	.24
Design Team	10	3.80	.42			

Domain Two - Item 64

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Total Cohort	84	3.37	.64			
				.32	92	.75
Design Team	10	3.30	.68			

Domain Three - Item 80

	No. of	No. of				
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Total Cohort	84	3.42	.68			
				3.88	21	.001***
Design Team	10	3.90	.32			

P = <.05*; <.01**; <.001***

The researcher next performed a statistical analysis of the three domains, comparing males' and females' responses for domains one, two, and three. There was no statistically significant difference in the means for domains one and two. For domain three, there was a statistically significant difference. The male respondents had a higher mean score of 60.32 compared to the females' mean score of 56.42 (see Table 22).

Table 22

Independent t-Test Comparing Males' and Females' Responses for Domains One,
Two, and Three

Domain One

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Male	19	98.79	8.02			
				1.70	69	.09
Female	52	95.04	8.28			

Domain Two

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Male	20	130.10	13.88			
				.39	67	.70
Female	49	128.57	14.97			

Domain Three

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Male	22	60.32	4.34			
				2.14	75	.01**
Female	55	56.42	8.07			

P = < .05*; <.01**; <.001***

The researcher next used the variable of age to do an analysis of variance of domains one, two, and three. The results of the analysis are presented in Table 23. The data show that there were no statistically significant differences based on age among the three domains.

Table 23

One-Way Analysis of Variance of Differences Based on Age for Domains One, Two, and Three

and Three Domain One									
		Sum of	Mean						
	df	Squares	Square	F	Sig.				
Between Groups	3	174.93	58.31						
				.84	.48				
Within Groups	67	4,673.94	69.76						
Domain Two									
		Sum of	Mean						
	df	Squares	Square	F	Sig.				
Between Groups	3	1,571.88	523.96						
				2.65	.06				
Within Groups	65	12,869.11	197.99						
		Domain T	hree						
		Sum of	Mean						
	df	Squares	Square	F	Sig.				
Between Groups	3	149.76	49.92						
				.91	.44				
Within Groups	73	3,997.41	54.76						

P = < .05*; < .01**; < .001***

The researcher then compared the means of the three domains based on race. Because there was only one Asian participant and one Native American participant, those two participants were dropped from the statistical analysis. The analysis compared the responses of African-Americans and Caucasians for domains one, two and three. The results of the statistical analysis indicate there were no statistically significant differences in any of the three domains based on race. These results are shown in Table 24.

Table 24

<u>Independent t-Test Comparing African Americans and Caucasians for Domains One,</u>

Domain One

-	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
African-American	14	95.36	5.67			
				.35	68	.73
Caucasian	56	96.23	8.97			

Domain Two

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail
	Cases	Mean	ა.D.	t- v arue	υ.г.	Sig.
African-American	13	134.15	13.17			
				1.37	66	.17
Caucasian	55	127.6	14.92			

Domain Three

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
African-American	17	59.41	4.64			
				1.25	74	.22
Caucasian	59	56.88	7.96			

P = < .05*; <.01**; <.001***

Two, and Three

The next statistical analysis, Table 25, compared different levels of education—elementary versus middle and high school. For domains one through three, there were no statistically significant differences across the domains.

Table 25

Independent t-Test Comparing Elementary versus Middle and High School
Respondents on Domains One, Two, and Three

Domain One

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Elementary	28	94.21	9.05			
				1.50	68	.14
Middle and High	42	97.26	7.78			

Domain Two

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Elementary	26	125.08	16.92			
				1.76	65	.08
Middle and High	41	131.43	12.62			

Domain Three

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Elementary	28	56.75	7.39			
,				1.15	73	.26
Middle and High	47	58.60	6.33			

P = <.05*; <.01**; <.001***

The next analysis was based on whether the respondent was applying for a principalship. For domains one and two, there were no statistically significant differences. For domain three there was a statistically significant difference in which those who were applying for a principalship had a higher mean than those who were not. The mean was 60.23 for applicants and 56.52 for non-applicants. The data are presented in Table 26.

Table 26 Independent t-Test Comparing Applicants and Non-Applicants for Principalships in Domains One, Two, and Three Domain One

	No. of Cases	Mean	S.D.	t-Value	D.F.	2-Tail Sig.
Applicants	42	95.24	9.33			
				1.10	66	.28
Non-Applicants	26	97.54	6.54			

No. of 2-Tail Mean S.D. t-Value D.F. Cases Sig. **Applicants** 43 129.93 14.38 .52 65 .60 Non-Applicants 24 127.96 15.63

Domain Two

No. of 2-Tail Cases Mean S.D. t-Value D.F. Sig. **Applicants** 48 60.23 8.31 2.11 72 .01 Non-Applicants 26 P = < .05*; <.01**; <.001*** 56.52 4.49

Domain Three

The next statistical analysis was based on participants applying for a leadership position within the system other than the principalship. For domains one and two, there was no statistically significant difference between those applying and those not applying. For domain three, there was a statistically significant difference favoring those persons who were applying for positions within the school system. The mean for applicants was 59.74. For non-applicants it was 55.74. These results appear in Table 27.

Table 27

<u>Independent t-Test Comparing Applicants and Non-Applicants for Other Leadership</u>

Positions in Domains One, Two, and Three

Domain One

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Applicants	32	95.41	7.87			
				.61	68	.55
Non-Applicants	38	96.63	8.85			

Domain Two

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Applicants	29	125.72	15.18			
				1.75	66	.09
Non-Applicants	39	131.90	13.84			

Domain Three

	No. of					2-Tail
	Cases	Mean	S.D.	t-Value	D.F.	Sig.
Applicants	34	59.28	8.75			
				2.15	75	.01
Non-Applicants	43	55.74	5.65			

P = < .05*; < .01**; < .001***

The next item sought information on how participation in the program had affected the respondent's thoughts about applying for future leadership positions. There were three response categories. "More likely to apply" elicited 58 responses; "less likely to apply" drew 4 responses; "unchanged" brought 21 responses. The researcher did an analysis of variance across the three domains based on the three categories of choice for respondents. There were no statistically significant differences across domains one and two based on future thought about applying for

leadership positions. For domain three, there was a statistically significant difference, but after applying the Scheffé test, it was judged to be not significant. Table 28 displays these results.

Table 28

One-Way Analysis of Variance of Differences Based on Respondents' Thoughts

About Applying for Future Leadership Positions in Domains One, Two, and Three

Domain One

	Sum of	Mean		
df	Squares	Square	F	Sig.
3	401.11	200.56		
			3.04	.06
68	4,478.18	65.86		
	Domain 7	Γwo		
	Sum of	Mean		
df	Squares	Square	F	Sig.
2	918.70	459.35		
			2.24	.11
66	13,537.94	205.12		
	Domain T	hree		
	Sum of	Mean		
df	Squares	Square	F	Sig.
2	410.74	205.37		
			4.12	.02
74	3,690.97	49.88		
	3 68 df 2 66 df 2	df Squares 3 401.11 68 4,478.18 Domain 7 Sum of Squares 2 918.70 66 13,537.94 Domain T Sum of Squares 2 410.74	df Squares Square 3 401.11 200.56 68 4,478.18 65.86 Domain Two Sum of Mean Squares 2 918.70 459.35 66 13,537.94 205.12 Domain Three Sum of Mean Squares 2 410.74 205.37	df Squares Square F 3 401.11 200.56 3.04 68 4,478.18 65.86 65.86 Domain Two Sum of Mean Squares Square F 2 918.70 459.35 2.24 66 13,537.94 205.12 Domain Three Sum of Mean Squares Square F 2 410.74 205.37 4.12

P = < .05*; < .01**; < .001***

Summary of Quantitative Data from the Survey Instrument

A summary of the survey data reveals a high level of response from program participants (86%) and design team members (83%). Overall, all respondents rated

the program as highly effective in enhancing the leadership skills measured in the domains tested. A summary of the quantitative findings is as follows:

<u>Finding #1</u>: The instrument had a high degree of reliability across the three domains tested and across the three cohorts. They were all statistically significant at the .001 level.

<u>Finding #2</u>: In each of the domains, a summary question reflecting the entire domain was inserted. An analysis of variance conducted on these summary questions revealed no statistically significant difference among the three cohorts.

Finding #3: A one-way analysis of variance compared the responses of all three cohorts to the design team and revealed no statistically significant difference for Domains 1 and 2. There was a statistically significant difference with Domain 3 favoring the design team.

<u>Finding #4</u>: An independent t-test of all three cohorts compared to the design team further confirmed no statistically significant difference in Domains 1 and 2, but a statistically significant difference at the .01 level for Domain 3.

<u>Finding #5</u>: An independent t-test of the total cohorts' responses versus the design team's responses to the three summary questions revealed identical results. There was no statistically significant difference in Domains 1 and 2. There was a statistically significant difference with Domain 3 at the .001 level.

<u>Finding #6</u>: An independent t-test comparing male and female responses revealed no differences in Domains 1 and 2. There were statistically significant differences at the .01 level of Domain 3 favoring male respondents.

<u>Finding #7</u>: A one-way analysis of variance for age differences among respondents revealed no statistically significant difference across the three domains.

<u>Finding #8</u>: An independent t-test comparing African American and Caucasian respondents revealed no statistically significant difference across the three domains.

<u>Finding #9</u>: An independent t-test comparing elementary and secondary school respondents across the three domains revealed no statistically significant differences.

<u>Finding #10</u>: An independent t-test comparing applicants and non-applicants for principalships revealed no statistically significant differences for Domains 1 and 2. It did reveal differences at the .01 level favoring applicants for principalships for Domain #3.

<u>Finding #11</u>: An independent t-test examining applicants and non-applicants for other leadership positions across the three domains revealed no statistically significant differences for Domains 1 and 2. It did reveal differences favoring applicants for Domain 3; however, after applying the Scheffé test, it was judged to be not statistically significant.

<u>Finding #12</u>: A one-way analysis of variance of differences based on respondents' thoughts about applying for future leadership positions in the three domains tested revealed no statistically significant differences.

Question 6: To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? What areas of the program did participants see as needing improvement?

Data for this question were collected from focus group interviews with program participants who have advanced in leadership and program participants who have become principals. Some data was collected from interviews with key system

leaders and the survey of program participants. An analysis of the data revealed the following areas of the program as benefiting participants:

- A clearer understanding of the principalship, especially in the area of accountability
- A greater sense of local preparation and desire for becoming a principal in the designated school system
- The establishment of a strong network of colleagues to provide support in their new position.

The analysis of data revealed the following areas of need in the program:

- A greater emphasis on time management, communication, and data analysis
- The need for a stronger partnership with the college involved with the program and support for advanced graduate level work
- Completing all phases of the designated program, including the project and internship components.
- Separate training for aspiring principals and for other leadership positions

A review of system records indicates that a number of participants in the system's Aspiring Leaders Program have advanced to various leadership positions over the first three years of the program. Table 29 shows the number of program participants who have advanced to leadership positions.

Table 29

Program Participants Who Have Advanced in Leadership Positions

	Instructional Specialist	Admin. Assistant	Vice Principal	Principal	Central Office Admin.
Cohort 1	1		1		
Teacher	7	1	11	1	0
Guidance Counselor	0	0	0	0	0
Instructional Leader	2	0	0	0	2
Vice Principal	0	0	0	8	1
Cohort 2					
Teacher	2	2	4	0	0
Guidance Counselor	0	0	0	0	0
Instructional Leader	0	0	0	0	0
Vice Principal	0	0	0	0	0
Cohort 3					
Teacher	4	2	7	0	1
Guidance Counselor	0	0	0	0	0
Instructional Leader	0	0	0	0	0
Vice Principal	0	0	0	0	0
TOTAL	15	5	22	9	4

Out of 99 program participants over the first three years of the program, a total of 55 have advanced in leadership (56% of total). Nine program participants have become principals while 22 have advanced to the vice principal level. Fifteen program participants have become instructional specialists, five are administrative assistants, and four others have gone onto Central Office leadership positions. Table 30 shows the gender and race of those individuals who have advanced in leadership.

Table 30

Gender and Race of Program Participants Who Have Advanced in Leadership

	White	White	Black	Black	Other	Other
	Male	Female	Male	Female	Male	Female
Instructional	3	10	0	2	0	0
Specialist Administrative Assistant	3	0	1	1	0	0
Vice Principal	8	12	0	2	0	0
Principal	1	4	1	3	0	0
Central Office Administrator	0	4	0	0	0	0
TOTAL	15	30	2	8	0	0

Because the system placed a high priority on including minority participants in the program, it should be noted that ten of the 55 individuals (18%) who advanced in leadership were minorities. Four of the nine individuals who have become principals, 44%, were minorities.

<u>Program Benefits—A Clear Understanding of the Principalship, Especially in the Area of Accountability</u>

A focus group interview with program participants who now have become principals revealed a common theme of what one participant called "a demystifying of the principalship" as a result of participation in the program. These individuals indicated that as current vice-principals, they had a good understanding of the responsibilities of the principalship, but their job requirements were not providing the background and experiences needed to feel prepared to advance. One program participant stated,

I was at a school that was getting good results. The principal was leading the staff and obtaining good test results and perfect audits with the budget, but I had no idea how he was doing it. As a vice principal,

I was relegated to doing IEPs and buses. Going through the program put me through some experiences that made me think I could do the principal's job. It provided the foundation and road map I had been missing.

These individuals cited the "hands-on" activities and practical topics of the program as the key to feeling confident in their new positions. "I remember my first day on the job my secretary brought me requisitions and the "green bar" sheet. I think she thought this would intimidate me. Before the program it might have because I had no idea how to handle the budget process. But we had done a monthly seminar that explained these issues, so I felt prepared to be able to do my job," stated one participant.

Every program participant interviewed in a focus group of individuals who had advanced in leadership discussed the issue of accountability as it relates to the principalship. They were all agreed that the program had a purposeful bias toward accountable leadership and cited that as a strength in the training. One participant said, "You hear all of the expectations placed on the principal and it really makes you question if you want the position. The program not only gave us training on how to meet those expectations, but also gave us access to current principals who are getting results. The notes I got from these people [current principals] meant more than any class I have taken."

Consistent areas cited by those interviewed in the focus group as assisting them in meeting the accountability requirements of their new positions included the focus on creating school improvement plans, hiring and terminating staff, finances, and understanding the requirements of the "No Child Left Behind" legislation.

A Greater Sense of Local Preparedness and Desire for Becoming a Principal in the Designated School System

As articulated by several system leaders and members of the design team, a major goal of this program was to prepare aspiring leaders to be successful principals in the local school system. Every program participant who had advanced in leadership indicated that not only had the emphasis on the local school system prepared them for success in their new position, but it had also fostered a strong drive to seek leadership positions in the designated county. Stated one program participant:

I got to hear from the superintendent and most of his executive staff during the program. I got to listen to Board of Education members and local politicians. I was in training with principals who were sharing tricks of the trade on how to be successful. I got to work with school system forms, policies, and processes. Yes, the focus on the school system was obvious and was helpful in training me in my new role. It also made me feel that the best place to seek a principal position where I could meet with immediate success was right here in [this] county.

Program participants indicated that the appeal of emphasizing the local school system went deeper than just spending time with system leaders. One program participant stated, "I think it was because the superintendent talked about the shared vision of the school system and then hearing it reinforced by the Central Office staff, principals, and local politicians. The way they discussed a common sense of system mission made me feel like I would be a part of a team and that there would be support to help me be successful as a new principal."

Another area that program participants cited as a strength of the program was the active participation of the system's principals. Principals serving on panels, facilitating monthly seminars, conducting diagnostic interviews, and a variety of other actions gave program participants tremendous access to current practitioners. A program participant, who is now a principal, indicated,

I will forever be in debt to my colleagues who helped in this program. Almost every day I reflect on some part of this training that they were responsible for and that I now use as a principal. We had the opportunity to hear from so many different and outstanding principals throughout this profession development activity. I was inspired by them to want to stay in [the school system] and volunteer to help in the program now that I am a principal.

Program participants indicated that the emphasis on local programs helped them know how and where to obtain system resources, to deal with the political aspects within the system community, and how to prepare for their own evaluations as principals.

The Establishment of a Strong Network of Colleagues to Provide Support in New Positions

The superintendent had hoped that providing a preparation program for principals would lead to a strong network of support for these individuals. Focus group interviews with program participants who had advanced in leadership indicated that the program had more than met the superintendent's expectations for networking opportunities. One program participant stated,

The strongest aspect of the program for me was the networking and relationships I built with the individuals in my group. As I moved into my principalship, I had no question or hesitation about contacting a fellow principal or Central Office leader because we had spent so much time with them in the training. I also developed really close bonds with

many of my peers who went through the training. I felt like I wasn't alone as I transferred to a new position.

Program participants cited the interactive nature and time allotted for questions and discussions as being a key to forging strong relationships between individuals in each cohort of the Aspiring Leadership Program. One program participant stated, "So often, training is so one-sided with people talking at you. This was so different. We had the opportunity to engage in activities, debrief, and ask questions. That's what made it so rewarding was the personalized attention we received."

The structure of the program, with its emphasis on being hands-on and practical, was cited as a key to building a strong network for program participants. "It really provided a safe harbor to learn about leadership. As participants, we were made to feel comfortable making mistakes. The principals running each session stressed the willingness to help us at any time, and believe me, I have taken them up on that offer."

The majority of focus group participants indicated that the strongest networking opportunities were established with their fellow program participants, followed by system principals, Central Office staff, and local politicians.

Needed Program Improvements—A Greater Emphasis on Time Management, Communication, and Data Analysis

As various program participants moved into leadership positions, they were in agreement that the training had been extremely beneficial in preparing them for their new roles. However, there were some areas that participants agreed had not been stressed as much in the program and were noted as areas for improvement in the future.

Three participants who became principals cited a greater need to focus on time management as it relates to the principalship. One individual stated,

That's the one area I was not ready for as I became a principal.

Everywhere I walked in the building, somebody needed to talk to me or get my opinion on something. I felt myself falling behind early with all the things I needed to get done. I wish we could have done more in the program with how successful principals manage the multiple commitments on their time.

Other participants took a different position on the time management issue, citing that the program needed to help individuals cope with balancing the professional and personal demands of the principalship. Every program participant who had become a principal indicated that they were unprepared for the demands on their personal lives that the job requires. One individual explained, "I really would have liked to have dialogued with current principals on how they manage to have a personal life. Everyone tells you to take a vacation and take time for yourself, but when do you do this? How do you find the time? Hearing how other principals balance these issues would really have helped me in my transition."

Participants also indicated a greater need to focus on communications. "We spent a lot of time being told to think like a principal, which we did, but we didn't spend a lot of time speaking or writing like a principal during the program," indicated one participant. These individuals suggested incorporating more opportunities during the monthly seminars to help participants practice oral and written communication skills expected of principals. Simulations that would incorporate communications skills were also suggested. A program participant, who is now a principal, stated,

During my first year, I had to deal with a student bringing a weapon to school. I had to write a letter to my parents, address the faculty, and talk about it at my PTSO meeting. I made the best decisions I could,

but I kept thinking this would have been a great 'real life' situation to have dealt with during the leadership program.

Focus group participants also indicated that communications with key Central Office staff should be more realistically addressed. One participant explained, "How do you communicate frustrations or issues when it's your boss? I remember going crazy my first few weeks on the job because I had filled out an extra staffing request and nobody could tell me if I was going to get it. I called, e-mailed, and wrote memos. I kept thinking, 'Maybe I'm not doing what I should to express my needs.'" All but one participant who became a principal indicated a need to address more "realistic" communications with Central Office staff.

Finally, several participants indicated a desire to focus more on the idea of making "data-driven decisions." While the focus groups indicated that effective use of data had been discussed often in the training, this "was a concept that really needed us to get our hands dirty and work with school data, not just hear about it," said one newly appointed principal. Several participants remembered that they had been given a fictitious school with data early in the program, but they never really used it. One participant stated,

This was a real hole in the program. You don't get the chance to work with this kind of data as a vice principal and then you are made a principal and expected to know how to have data-driven decisions.

Doing more with data would really benefit the program.

The Need for a Stronger Partnership with the College Involved with the Program and Support for Advanced Graduate Level Work

In focus group interviews, all but two individuals had taken the system's Aspiring Leadership Program in conjunction with the CASE Program. This partnership with a local university was meant to serve as a precursor for future

doctoral studies in Instructional Leadership. Focus group participants expressed their anticipation of a blending of the county program and the university courses. The consensus of the group interviewed was that this did not happen. One participant stated, "There were so many times topics crossed over between the program and the college, but instead of building upon each other, it just seemed these things were done by accident, which I think was the case. Just some common planning between what we were doing with the system training and our courses would have made such a difference."

Several participants also expressed displeasure in a perceived lack of system support for continuing in the CASE Program. System records indicated that of the participants who originally signed up for CASE, only four enrolled in the first set of classes that would lead to a Ph.D. Focus group participants stated the following reasons for not continuing in the program:

- Poor communication about registration and completing paperwork to enroll
- Moving the classes from the designated school system to a neighboring system
- A lack of financial commitment from the schools system to help with tuition

The lack of financial support was a concern raised by all focus group participants. One individual stated, "Originally, CASE was \$390 per course. When I found out the next phases was going to be \$1,300 per class, plus fees, there was no way I could continue. I was really disappointed and felt let down." Several school system leaders interviewed for this study were unaware of this issue and stated they would look into the situation.

Completing All Phases of the Designated Program, Including the Project and Intern Components

Perhaps the most common criticism of the Aspiring Leaders Program by focus groups and key system leaders was not providing the last phases of the program—an intensive research project and an internship component. System records were somewhat unclear about this issue, however out of the 99 participants who comprised the first three cohorts, only four completed any formal project and seven have been afforded an opportunity to do any type of internship. A key system leader said, "This is an area we need to focus on; we haven't done well on this front. The opportunity for people to get actual field experience is critical to build our principal preparation program." The vast majority of participants in the focus groups indicated that including the internship would substantially improve the program. One participant stated, "After we were done with the first three phases of the program, things just kind of stopped. It wasn't until I became a vice principal two years later that I felt I got the chance to put to use the things I had learned."

Several of the participants who became principals indicated an internship experience for the position would have been an excellent foundation for their transition. "Some people see the vice principal as the training ground for a principalship, but you're so busy doing that job that I'm not sure you get to reflect on how to handle the principal's job. Even a brief time interning as a principal would have really been a tremendous professional experience."

Several participants suggested that an internship did not have to be for an extended period of time or even during the school year. The school system being studied runs several major programs during the summer involving thousands of students, and participants thought that these activities would provide excellent internship opportunities. A participant stated, "Giving people who go through the Aspiring Leaders Program the opportunity to run a summer reading program or fill in

while a principal is on vacation during the summer would not require a lot of money or take people out of the classroom. I think the benefits to the system would outweigh the costs."

Many participants also saw the research project component as being valuable to the program, especially if the projects could be tied to graduate level work. "Most graduate programs in administration require some sort of research project. You should be able to do research that can satisfy your graduate program and the Aspiring Leaders Program that will also really benefit a school or school system."

Separate Training for Aspiring Principals from Training for Other Leadership Positions

Many of the individuals from the various focus groups suggested that the program would be strengthened by separating training for future principals from the other educational leadership positions. All focus group participants agreed that the training had a strong principal bias, which presented both positive and negative issues. One program participant stated, "While I learned a great deal about the principal position, I never really wanted to pursue that job. I really want to be an instructional specialist, so while the training was useful, it did not really address my specific needs." Several program participants who had advanced to non-administrative leadership positions agreed with this idea. One individual reflected, "As an instructional specialist, the training has helped me better assist the principal because I know now more about the issues they are dealing with. However, the training would have assisted me in my transition if it was not totally focused on the principalship."

The focus group of program participants who had become principals also agreed with this suggestion. Several indicated that some of the training was not focused on the principalship as it should have been because it tried to address other leadership issues. One current principal stated, "I took this program to prepare to be a

principal. It was great training, but at times it seemed presenters and facilitators felt bad about all of the attention on this role [the principalship]. They seemed to go out of the way to give nominal mention to other leadership positions in the school system. I don't think that really helped individuals for pursuing a vice principal or principal position, and it took time away from the rest of us." All but two individuals in focus groups suggested separating training for the different types of leadership positions people may pursue would enhance the program.

Summary

This chapter presented the findings associated with this study. Qualitative and quantitative methods were used to address the six research questions raised in Chapter I. A number of recommendations for practice and further research were drawn from these findings and are presented in Chapter V. The following chapter also presents conclusions reached as a result of this study.

CHAPTER V

SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

This chapter consists of six parts: "Introduction," "Summary of the Study," "Findings of the Study," "Conclusions," "Recommendations," and "Suggestions for Further Research." The Introduction frames the issues that led to this study. The Summary includes the purpose of the study, problem statement, research questions, and methodology. Observations made from the analysis of data are found in the Findings section. Based on the findings, reflections about the principal preparation program are cited in the Conclusions. Suggestions for the local school system, state level policy makers, and university faculty involved in principal preparation are included in the Recommendations section. Implications for further research are reflected in the Suggestions for Further Research section.

Introduction

As schools nationwide face serious challenges ranging from random outbursts of violence to budget shortfalls, an increasingly diverse student population, and a growing teacher shortage, educational experts continue to assert the need for strong principal leadership (Lashway, 2003; Marshak, 2003). In response to these and other forces, the role of school principal has become increasingly complex and demanding. An unprecedented era of educational accountability has increased the responsibilities of principals, making their top priority providing leadership for learning. As standards-based reform has continued to increase, heightened by the recent passing of the "No Child Left Behind" legislation, the federal and state governments have placed

responsibility for student achievement directly on individual schools and, consequently, principals.

As time demands and expectations for principals have increased, the supply of qualified candidates for the position has decreased. Based on a survey conducted by Education Research Services (ERS) in 1998, half of the school districts across the country reported shortages of qualified principal candidates to fill vacancies (Tracy & Weaver, 2000, p.2). ERS and the U.S. Department of Labor estimate over 40% of all public school principals will retire or leave the position for other reasons before 2010 (Lovely, 2004, p.1). Many experts suggest "that a scarcity of capable education leaders ranks among the most severe of the problems facing schools nationwide" (Murphy, 2000, p.2).

Faced with an increasing concern over finding a sufficient quantity of qualified principal candidates, many school districts have responded by developing leadership within their own organizations and instituting "grow your own" principal preparation programs. While there appears to be an increasing number of "grow your own" principal preparation programs across the county (ERS, 2000; Tirozzi, 2001), there is limited research about their effectiveness or if they are a possible solution to addressing a shortage of qualified candidates for future principalships.

Summary of the Study

This study is the development of a critical evaluation of one school system's efforts to develop and implement a "grow your own" program to address issues of quality and quantity of principal candidates. The study serves two purposes:

 To determine the extent to which this system's efforts to design and implement a program that supplements university preparation have succeeded in meeting program goals (product), efficient use of system resources (input), and appropriate selection of program activities (process). The context in which this program was developed was also examined.

2. To determine what can be learned from these efforts that might be beneficial to this system as it attempts to improve the program, and what can be learned that might prove useful to other systems coping with the need to address principal shortages.

Chapter IV presented the results of data analysis aimed at answering the following research questions:

- 1. What forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program?
- 2. Which program elements were created specifically to address those forces impacting the principals?
- 3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program?
- 4. What were the essential sources that led to the elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure? What was the process for implementing the program and recruiting participants?
- 5. What perceptions did participants have about the effectiveness of the program and did these perceptions match the goals and expectations of the program design committee? Did program participation increase or decrease participants' interest in advancing to a leadership role?
- 6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities

of their positions? What areas of the program did participants see as needing improvement?

The research design for this study employed both qualitative and quantitative methodologies and was based on Daniel Stufflebeam's conceptual model for professional evaluation known as CIPP. Data were gathered to address the four components of the CIPP model as defined by the six research questions. The findings from this collection of data were presented under the components of context, input, process, and product. The first two research questions dealt with context, the third question dealt with input, question four dealt with process, while questions five and six dealt with product.

Summary of Findings for the Research Questions

The study revealed a wide array of information about "grow your own" programs, much of which was supportive; however, this researcher found few evaluative reports on these efforts. This raises a concern that these programs may not be receiving the critical analysis that they require.

While this formative evaluation revealed wide support for the "grow your own" program under study, the data collected and analyzed, along with the review of research, provided ample evidence that these programs must be subjected to rigorous evaluation procedures if they are to achieve their intended ends.

Findings Supported by Review of Research

In addition to the numerous findings that this study of one school system's efforts to design and implement a "grow your own" program produced, there were several significant findings from the literature on principal preparation programs. Most revealing of these were:

- 1. There is considerable concern among the various factions of the educational community related to the quality of current university preparation programs, the most significant of which is the concern that these programs, though strong in the area of theory and philosophical percepts, fail to adequately address the practical side of principal preparation.
- 2. The review of research also contained cautionary comments relative to "grow your own" principal preparation programs. These data indicate that "grow your own" is not synonymous with "go it alone." The research overwhelmingly supports, as did the interviews with key administrators in the school system where the study was conducted, for a blending of university programs and resources with school system efforts and resources to prepare school-based leadership personnel for the challenges of a diverse and demanding society. The importance of education in contemporary society cannot be overstated, and the dependence of present society to have not only a "highly qualified workforce" but equally a fully enlightened citizenry is, or should be, abundantly clear.

Findings Supported by the Research Study

As indicated, a number of significant findings also emerged from the data related to the formative evaluation of one system's efforts to develop a "grow your own" leadership program. Prior to stating these findings, it is important to note that these findings are based on "formative" and not "summative" data. The study did not include any evidence on the "performance" of those individuals who have completed the program and who were promoted to principalships and/or other leadership

positions. The majority of them have been in these positions for two years or less when data for this study was collected, and their performance reviews are, at best, preliminary in nature.

Summary of Quantitative Survey Findings

Overall, the survey findings indicated that the instrument had a high degree of reliability and the respondents viewed the program as being highly effective in preparing them for the principalship as well as other leadership positions. It also revealed few differences among the various groups that participated in the survey. The important statistical findings, all of which were at a statistically significant level, were as follows:

<u>Finding #1</u>: The instrument had a high degree of reliability across the three domains tested and across the three cohorts.

<u>Finding #2</u>: In each of the three domains, a summary question reflecting the entire domain was inserted. An analysis of variance conducted on these summary questions revealed no statistically significant difference among the three cohorts.

Finding #3: A one-way analysis of variance compared the responses of all three cohorts to the design team and revealed no statistically significant differences for Domains 1 and 2. There was a statistically significant difference with Domain 3 favoring the design team.

<u>Finding #4</u>: An independent t-test of all three cohorts compared to the design team further confirmed no statistically significant differences in Domains 1 and 2, but a statistically significant difference at the 01 level for Domain 3.

<u>Finding #5</u>: An independent t-test of the total cohorts' responses vs. the design team's responses to the three summary questions revealed identical results. There

were no statistically significant differences in Domains 1 and 2. There was a statistically significant difference with Domain #3 at the 001 level.

<u>Finding #6</u>: An independent t-test comparing male and female responses revealed no statistically significant differences in Domain 1 and 2. There were statistically significant differences at the 01 level of Domain 3 favoring male respondents.

<u>Finding #7</u>: A one-way analysis of variance for age differences among respondents revealed no statistically significant difference across the three domains.

<u>Finding #8</u>: An independent t-test comparing African American and Caucasian respondents revealed no statistically significant difference across the three domains.

<u>Finding #9</u>: An independent t-test comparing elementary and secondary school respondents across the three domains revealed no statistically significant differences.

<u>Finding #10</u>: An independent t-test comparing applicants and non-applicants for principalships revealed no statistically significant differences for Domains 1 and 2. It did reveal statistically significant differences at the .01 level favoring applicants for principalships for Domain #3.

<u>Finding #11</u>: An independent t-test examining applicants and non-applicants for other leadership positions across the three domains revealed no statistically significant differences for Domains 1 and 2. It did reveal statistically significant differences favoring applicants for Domain 3; however, after applying the Scheffé test, it was judged to be not statistically significant.

<u>Finding #12</u>: A one-way analysis of variance of differences based on respondents' thoughts about applying for future leadership positions in the three domains tested revealed no statistically significant differences.

Summary of the Qualitative Data Findings

A large amount of data was collected from personal interviews, focus group interviews, and the review of records. The findings are as follows:

1. The data indicated that a number of forces combined to prompt system executives to initiate a "grow your own" program—some internal and some external in nature. The data indicated that a primary concern was the lack of qualified candidates to meet existing and projected needs at the principal level. While this concern for quantity and quality of the candidate pool is national in nature, the system executives focused on local, not national, needs. While the need for principal candidates has been acute in this school system [a 71% turn-over rate in eight years], there was a particular need to attract minority candidates to the position. Creating a "grow your own" program was seen as an excellent vehicle to do this.

The data also indicated that program design was largely in response to external forces—the nationwide school accountability movement and the research literature support for the importance of the principal in providing the instructional leadership needed to address these accountability demands.

In addition, system administrators were influenced by a concern that principals be skilled in relating school-based decisions on input from community and business leaders, focusing on the belief that school and communities are one.

2. The important program elements tended to match the concerns that gave rise to the initiation of this training. There was the somewhat obvious need to change the focus of principals from that of "school manager" to "instructional leader." The issue was how best to do that and, at the same time, prepare school leaders who met state and national accreditation standards. A match between system goals and these standards, notably ISLLC licensing standards, was highly correlated. Hence these standards and the state's previous Principal Assessment Center served to guide

the design team's direction. School system leaders and design team members all expressed concern with the perceived theoretical training provided by university preparation of principals. In response to this, there was a great emphasis on making the program "hands-on" and practical in nature.

- 3. "Grow your own" programs are not without costs—the resource investment, both human and fiscal, is comparatively high. The program's initial cost for the first three years was over \$100,000. One of the benefits that helps to offset "human costs" is the growth benefits derived to those system personnel who contributed to program design and operations. Key school system leaders indicated that serving as presenters and facilitators in the program became excellent professional development for the school system's principals. Whether or not long-term costs will be less than start-up costs is unknown.
- 4. While the conceptual base for the program emerged out of concern for the system's need to create a sufficient supply of qualified candidates (system goal), it was also embedded in the research literature on the principalship and the increasing standards for principal certification and licensing.
- 5. The process of creating a "grow your own" program took over two years of intense system efforts and required the participation of numerous school stakeholders. The active participation of the school system's principals in the design created a sense of shared vision and ownership in the program, leading to their active participation. The process of designing the program was enhanced by including state and federal consultants as well as visiting other school systems with principal preparation programs.
- 6. Program participation seemed to have greatly affected interest in pursuing leadership positions. Fifty-nine percent of the program participants indicated that the program greatly increased their desire to pursue leadership positions. Of the

99 participants from the first three cohorts of the Aspiring Leaders Program, 55 have already advanced to various leadership positions: nine at the principal level and 22 at the vice principal level. Not only did program participants say that the training truly prepared them for future principalships, but also it greatly enhanced their desire to pursue leadership in the designated school system, reflecting one goal of a "grow your own" model.

- 7. Program participants, while outwardly enthusiastic relative to program content and delivery, had much to say about ideas for program improvement. Two of the most often expressed needs were for an internship experience that mirrored the responsibilities of building principals and for leadership preparation aimed at individuals who, while not presently interested in pursuing a principalship, were interested in roles in the area of instruction and curriculum development.
- 8. Those who had obtained principalships indicated that they felt the program addressed system expectations of the principalship while preparing them in the effective use of system financial and security requirements of the position. On the other hand, they expressed concern that the program needs to provide for additional emphasis on how to blend positional demands with personal and family needs. They reported with praise for the opportunities to interact, often on a one-to-one basis, with system leaders as well as with board members and community leaders. These exchanges, along with interactions with school-based administrators, elevated their "comfort level" in networking with these individuals now that they are principals.

Conclusions

Because a large amount of data was collected for this study, there were a number of significant findings and conclusions. Based on the findings of this study,

the following are conclusions regarding the Aspiring Leadership Program implemented in the system being studied.

Conclusion #1

From the data that were gathered, the school system should continue its "grow your own" program for aspiring leaders. The program was viewed as successful and very effective in addressing its initial purposes, specifically to increase the pool of qualified candidates for leadership positions. Data analyzed from the survey reflected a high degree of satisfaction from all respondents in the training and the preparation greatly increased participants' knowledge and skills in the targeted leadership areas. Program involvement tended to increase participants' understanding, knowledge, and appreciation of the principalship. Program participants who went on to become principals indicated that the program greatly helped them in making the transition to their new positions. Additionally, the program provided a large number of qualified individuals for future leadership opportunities. Of the 99 individuals who have completed the first three cohorts, 55 have advanced in leadership, including nine participants who have become principals and 22 who have become vice principals.

Conclusion #2

"Grow your own" principal programs need to deal with forces unique to individual school systems while also addressing state and national forces impacting the principalship. Program participants indicated one of the most appealing features of this program was the "local flavor" it provided. Participants who had advanced in leadership indicated it not only made them want to stay and pursue leadership in the designated county, but also provided a greater degree of confidence that they were prepared for their new positions. While the focus on local issues was appreciated by program participants, so too was the emphasis on aspects of leadership affected by

state and national policies and requirements. The clear focus on accountability in the program benefited program participants who advanced in leadership roles, helping them feel ready to meet state mandated testing requirements as well as the provisions of the "No Child Left Behind" legislation.

Conclusion 3

Creation of "grow your own" leadership programs needs system-wide leadership and resources, requiring a shared vision that unites and ignites system action and which sparks broad community support and increased financial support from local and state officials. One of the most commonly reflected ideas from key school system leaders, design team members, and program participants was the clear vision and mission for educational leadership, which was articulated in the designated county. As the program was created, the school system's vision was infused in all aspects of its implementation, indoctrinating aspiring leaders to the standards expected of them while involving local political leaders and outside officials to establish a unity of purpose and mission. This resulted in the system obtaining additional resources to continue providing the training as local officials felt a greater connection and need to support the system's efforts in leadership preparation.

Conclusion 4

Program participants saw the value in separating efforts to prepare future principals from other educational leaders. Because of the inclusive nature of the program, teachers with limited experience were in the same training as experienced vice principals. Numerous individuals indicated that creating a separate strand for aspiring principals would be beneficial to all participants. It was also felt that this would allow for including the internship experience, which was cited as one of the areas of needed program improvement.

Conclusion 5

"Grow your own" programs are not synonymous with "Go it alone." While the vast majority of design team members, presenters, and resources were from the local school system, experts and consultants were tapped from other school systems, the State Department of Education, the U.S. Department of Education, and local universities. All of these elements combined to provide an extremely satisfying program, according to program participants. Participants not only valued the collaboration with a local university that allowed them to obtain college credit, but also indicated that the school system should further pursue a university partnership more directly related to program improvement and implementation.

Recommendations For Practice

The research results document several areas for program improvement for the designated school system that could also benefit other systems' planning to implement "grow your own" programs. The following recommendations are offered as a result of the conclusions.

Recommendation #1

The school system should continue to provide training for future educational leaders, but provide a separate program for aspiring principals.

Brief Rationale

One of the main reasons for creating the Aspiring Leaders Program was to increase the quantity and quality of future candidates for the principalship. The fact that nine principals and 22 vice principals have already come from the first three cohorts seems to indicate that the program represents a good investment in leadership development. As was recommended by many of the program participants, separating the training with one focus on aspiring principals and a second on leadership at all

levels would enhance the program. An unexpected benefit of the program has been the number of individuals who have advanced to other leadership positions (24), to include instructional specialists, administrative assistants, and Central Office positions. While these individuals indicated that training was still beneficial for them, they noted the clear focus and bias towards the principalship, and felt separating the training would allow for a greater differentiation of preparation for the variety of future leaders for the school system.

Recommendation #2

The school system should implement the internship and action research components of the Aspiring Leaders Program.

Brief Rationale

Failure to fully implement the internship and action research components that were part of the system's proposed leadership model were consistently noted areas of improvement. Many experts in leadership preparation cite a strong internship as a vital component of preparing individuals for the principalship (Lauder, 2000; Marshak, 2003; Newsom, 2001). The internship would afford program participants the opportunity to work in conjunction with an experienced principal and receive critical feedback on issues that individuals may not be facing as vice principals. The action research component would assist program participants in the data analysis and instructional leadership aspects of the principalship. It would also increase the value of research results in general to guide the leadership decision-making processes.

Recommendation #3

As "grow your own" programs are designed and implemented, an evaluation component should be built into the design to ensure an organized collection of data

and research, allowing system leaders the opportunity to monitor progress and enhance program improvement.

Brief Rationale

The school system studied operated its Aspiring Leaders Program for three years, spending over \$100,000 in its design and implementation without a format for evaluation of its effectiveness. Because the program went from design to implementation rather quickly, data collection and record keeping were diffused throughout the organization. Not only did this lead to key records and data not being shared, but it also created a situation where program evaluation was not assigned to a specific person or division. The lack of an evaluative component in principal preparation programs is not an uncommon occurrence as only two of the 17 programs in the state in which the study took place could articulate any type of evaluation procedure in their training. Research on "grow your own programs" substantiates this concern. Incorporating an evaluation component will allow system leaders to ensure the training provided is meeting system goals while also allowing them to target needed areas of improvement.

Recommendation #4

School system leaders and local university officials need to proactively engage in a partnership to blend the best opportunity for preparation of future educational leaders.

Brief Rationale

While the school system being studied did form a collaborative relationship with a local university, it was for the primary purpose of allowing participants to obtain graduate level credits. When key school system leaders first met with

representatives from the university, there was no dialogue about collaborating or blending materials or curriculum taught in university classes and the system's program. The relationship was symbiotic; the university received students and the school system was able to provide credits as an incentive for participation.

Current research about "grow your own" principal preparation programs indicates that they are most effective when they are in collaboration with university preparation. Key school system leaders and program participants all indicated forging a strong partnership with the local university would greatly benefit the experience individuals received from the training. Aligning courses and curriculum taught at the university with key aspects of the system's program would not be difficult and would provide a richer experience for participants.

Recommendation #5

School system leaders should expand partnerships with other systems and State Department of Education to pool resources and provide the most current training to ensure an adequate supply of future principal candidates.

Brief Rationale

While ensuring the training had a "local flavor" was cited as one of its strongest features, the preparation program was benefited by seeking the assistance of the State Department of Education as well as other school systems that had already implemented such programs. The Southern Regional Education Board has cited several examples of regional principal preparation programs that allowed for a blending of resources while still retaining the unique aspects of each system (SREB, 2001, 3). The school system being studied benefited greatly from the expertise provided from consultants from the federal and State Departments of Education and from nationally based professional associations, such as NASSP and NAESP.

Recommendation #6

The school system should consider a needs assessment to determine if the goal of promoting more minority candidates to leadership positions is being met.

Brief Rationale

One of the major forces that led to the formulation of a leadership program in the school system was a desire to increase the number of minorities in administrative positions, particularly at the principal level. As the program enters into its fourth year, the results are mixed. Data show that four of the nine participants who have become principals were minorities (44%). Certainly this would appear to be encouraging to school system leaders. However, of the 55 individuals who have advanced in leadership after participating in the program, only 10 were minorities (18%). Of the 22 program participants who have been promoted to vice principalships, only 2 were minorities. Having such a limited number of candidates who are minorities advance to the vice principalship does not bode well for diversifying the principal ranks in the future. The design committee needs to work with key school system leaders to see if changes or modifications need to be made in the leadership program and other aspects of the system's organization to encourage and promote a greater number of minority candidates into leadership positions.

Recommendations for Further Research

Recommendation #1

Examine the forces that have lead to an increase in "grow your own" principal preparation programs across the state being studied and the nation.

Brief Rationale

While this study identified the major forces, which led one school system to design and implement a "grow your own" principal preparation program, the findings are unique to this system and not generalizable to other school districts. Current research indicates that such programs are increasing rapidly across this country, but little research has been conducted to identify common reasons or national implications.

Recommendation #2

Explore how data can be collected in school systems that have developed "grow your own" programs to indicate their overall level of effectiveness and areas for improvement.

Brief Rationale

Of the 17 "grow your own" programs developed in the state in which the study took place, only two had any type of formal evaluative component. This is a trend also reflected in the literature review. If school systems are going to expend significant human and fiscal resources on the development and implementation of "grow your own" programs, then there should be methods employed before, during, and after such training to assess if system goals are being met. While Stufflebeam's CIPP model worked well in evaluating this school's program, alternative approaches may need to be employed in other school districts. Further research is needed to develop effective and efficient evaluation models. As noted earlier, the CIPP model is very complex and may involve more costs than most systems will desire or be able to invest.

Recommendation #3

Initiate a series of studies to assess the effectiveness of university principal preparation programs to explore the perception that they are "out of touch" with the reality of today's schools.

Brief Rationale

A consistent finding in the literature review and research conducted for this study indicated a common perception that university programs for principal preparation are ineffective and need to be overhauled. However, there seems to be little research to support this notion. In one such study, Louis Williams (2001) stated, "After a review of literature revealed the inescapable conclusion ... there is not much research" evaluating the effectiveness of university leadership programs (Williams, as cited in Lashway, 2003, p.3). If research does indicate major deficiencies in university preparation programs, school systems can tailor "grow your own" programs for principals accordingly and universities can better address program revisions so as to provide improved preparation programs for future educational leaders. While these improvements would be very beneficial, they would not negate the need for collaborative program efforts; in fact, they may tend to intensify collaborative efforts.

Personal Reflection on This Study

The researcher's interest in this topic stems from his work as a secondary principal and participation on several state and national committees on principal preparation. This study not only afforded opportunity to examine the most current research about the development of aspiring principals, but also focused on the efforts of one school system to design and implement a "grow your own" program. As public education continues to face challenges, ensuring a sufficient supply of qualified principal candidates will be one of the most significant. School systems will need to take a greater role in the identification and preparation of future administrative

candidates. With increased expectations on student performance and accountability, future principals will need a blending of proven theoretical and pedagogical training, along with practical, hands-on leadership experiences that "grow your own" programs and universities can provide together.

Appendix A

The Survey

Evaluation of the Charles County Aspiring Leaders Program



The purpose of this survey is to evaluate the effectiveness of the Charles County Aspiring Leaders Program in helping aspiring school leaders develop the knowledge and skills necessary for them to be successful. The survey contains three sets of statements that describe elements of knowledge and skills associated with effective school leadership. The first set of statements focuses on the domain of organizational development. The second set of statements focuses on the domain of organizational environment, and the third set of statements focuses on the domain of the educational program.

After reviewing each statement, please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas by clicking on the appropriate box. If you served as a member of the Aspiring Leaders Program design team, please indicate your assessment of the effectiveness of the program in each of the areas included in the various statements. At the end of the survey, you are invited to provide some information about your background and experience.

To proceed to additional statements, click on "Next Page" at the bottom of each screen. To review your responses to previous statements, click on "Previous Page" at the bottom of each screen.

When you have completed the survey, click on "Submit Survey."

Thank you very much for taking the time to complete the survey.

Next Page 5%

This online survey is powered by WebSurveyor.

SECTION I. ORGANIZATIONAL DEVELOPMENT: VISION AND ACCOUNTABLE LEADERSHIP

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
 The principal helps the faculty to develop an understanding of, and support for, the beliefs and attitudes that form the basis of its educational value system. 	C	C	C	C	C
 The principal provides opportunities for identifying and reaching consensus on educational goals. 	r	C	C	C	C
 The principal sets high, professional expectations and standards for the faculty and school. 	C	C	C	C	(
 The principal helps the faculty develop high professional expectations and standards for themselves and the school. 	C	C	C	C	C
The principal envisions future goals and directions for the school.	<u>(</u>	C	C	C	(
 The principal encourages changes in school programs that lead to a better school for the students. 	C	C	C	C	C
The principal communicates to the faculty ways in which school programs can grow.	C	C	C	C	C
8. The principal involves the community and community agencies in the planning process	C	C	C	C	C

		oal uses outside assist in school 	C	C	C	C	C	
			***************************************	# <u>1:49:49</u> :20:40000000000000000000000000000000000			***************************************	er - u -ucor
Pre	evious Page	Next Page	(1.20 kg)	14%				

This online survey is powered by WebSurveyor.

ivatuan metara Asatuan metuara

SECTION I. ORGANIZATIONAL DEVELOPMENT: VISION AND ACCOUNTABLE LEADERSHIP (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
 The principal gathers information and feedback from individuals and community agencies. 	C	C	C	C	C
11. The principal provides for the dissemination of information to individuals and community agencies.	C	C	C	C	C
The principal supports and operates within the policies of the district.	C	C	C	C	(
 The principal maintains good rapport and working relationships with other administrators of the district. 	C	C	۲	C	C
14. The principal invests time with the district office and community agencies to obtain their support and resources.	C	۲	C	C	C
The principal strives to achieve autonomy for the school.	C	C	C	C	(
16. The principal develops and implements school practices and policies that synthesize educational mandates and theories.	C	C	C	٢	r

and effectively interaction local, state, and nation communities.		C	C	C	C	C	
 The principal information staff of new developmed ideas in education. 	rms the nents and	C	C	C	r	C	
					Agenta (1975)		>
Previous Page Next I	Page		19%				

This online survey is powered by WebSurveyor.



SECTION I. ORGANIZATIONAL DEVELOPMENT: VISION AND ACCOUNTABLE LEADERSHIP (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
 The principal supports and participates in the identification of needed change rather than directing change. 	C	C	C	C	C
 The principal supports and participates in the evaluation of change rather than directing change. 	C	C	C	C	C
21. The principal anticipates the effects of school-wide decisions.	C	C	C	(C
 The principal fairly and effectively evaluates school personnel. 	C	C	C	(C
23. The principal involves the faculty in the decision-making process.	C	C	C	(C
24. The principal sees the value of discussing school-related problems with the faculty and seeking their opinions about the problem.	C	C	C	C	C
25. The principal uses a systematic process the faculty knows and understands to implement change.	C	C	C	C	C
26. The principal has the patience to wait to resolve a problem if the best solution to that problem is not readily	C	C	C	C	C

evident.

27. Thinking about your knowledge and skills in the domain of "Organizational Development: Vision and Accountable Leadership" (Questions 1-26) prior to your participation in the Aspiring Leaders Program, to what extent do you feel your knowledge and skills in this domain have improved as a result of participating in the Aspiring Leaders Program?

- To a great extent
- To some extent
- To a limited
- Not at all
- C Don't know



This online survey is powered by WebSurveyor.



SECTION II. ORGANIZATIONAL ENVIRONMENT: HUMAN RESOURCES AND COMMUNITY INVOLVEMENT

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
28. The principal is willing to admit to making incorrect decisions and corrects them, if feasible.	C	C	C	C	C
29. The principal perceives teacher needs.	(<u></u>	C	C	\subset
30. The principal sees the value of giving the faculty the support they need to be effective.	C	r	C	۲	C
31. The principal diagnoses the causes of conflict and successfully mediates or arbitrates conflict situations.	C	C	C	C	C
32. The principal creates an environment where the faculty feel at ease in his/her presence.	C	C	C	C	C
33. The principal compliments the faculty in a sincere and honest manner.	(C	C	C	C
34. The principal is receptive to suggestions.	C	C	<u></u>	\subset	C
35. The principal is accessible when needed.	\subset	C	C	C	\subset
36. The principal takes the time to listen.	Γ	(C	\subset	\subset

Previous Page Next Page 29%

This online survey is powered by WebSurveyor.

SECTION II. ORGANIZATIONAL ENVIRONMENT: HUMAN RESOURCES AND COMMUNITY INVOLVEMENT (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
37. The principal creates an environment where the faculty feel free to share school-related ideas and concerns with him/her.	C	C	C	C	C
38. When the faculty discuss a problem with the principal, the principal understands and appreciates the faculty's views on the problem.	r	r	r	C	C
39. The principal fosters relationships where the faculty feel he/she is sincerely interested in what they are saying.	C	C	C	C	r
40. Through effective management of the day-to-day operations of the school, the principal promotes a feeling of confidence in the school among staff, parents, and community.	C	C	۲	C	C
41. The principal finds the time to interact with students.	C	((((
42. The principal fosters an environment where students feel free to initiate communication with him/her.	C	C	C	C	C
43. Students view the principal as a leader of school spirit.	C	C	C	(C
44. The principal encourages student leadership.	\subset	($\overline{}$	\subset	C

45. The princ student respo	cipal helps develop onsibility.	\subset	C	C	C	(
	- Manadan (- Manadan (- 2013) Manada ya maraba ka manada manada (2012) Manada ka k		***************************************				P
Previous Page	Next Page	33	3%				

This online survey is powered by WebSurveyor.

as a second of the second of t	
1 9.0a 1 9.0a 1 9.0a	

SECTION II. ORGANIZATIONAL ENVIRONMENT: HUMAN RESOURCES AND COMMUNITY INVOLVEMENT (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
46. The principal is highly visible to the student body.	(C	C	Γ	\subset
47. The principal positively reinforces students.	\subset	C	C	(\subset
48. The principal enjoys working with students.	<u></u>	C	C	C	(
49. The principal keeps the faculty informed about important issues.	C	C	C	C	r
50. The principal provides the faculty with clear and easily understood information about school operations.	C	C	C	C	C
51. When the faculty are informed of administrative decisions, they are aware of what the principal expects of them as it relates to the decision.	C	C	C	C	۲
52. The principal is able to effectively organize activities and tasks.	C	C	C	C	r
53. The principal develops appropriate rules and procedures.	C	C	C	C	C
54. The principal uses systematic procedures for staff appraisal, such as retention, dismissal, and promotion procedures.	C	C	C	C	C

Previous Page Next Page 38%
rievious rage Next rage 38%

This online survey is powered by WebSurveyor.

SECTION II. ORGANIZATIONAL ENVIRONMENT: HUMAN RESOURCES AND COMMUNITY INVOLVEMENT (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
55. The principal establishes the overall tone for discipline in the school.	C	C	C	C	(
56. The principal establishes a process to make students aware of school rules and policies.	C	C	C	C	C
57. The principal explains the reasons for his/her administrative practices to the faculty.	C	C	C	C	C
58. The principal works with other leaders of the school to implement a team approach to managing the school.	C	C	C	C	C
59. The principal encourages the faculty to be sensitive to the needs and values of other faculty in the school.	C	C	C	C	C
60. The principal helps the faculty clarify or explain their thoughts by discussing those thoughts with them.	C	C	C	C	C
61. During meetings, the principal involves persons in the discussion who might otherwise not participate.	C	C	C	C	C
62. The principal shares personal feelings and opinions about school issues with the	C	C	C	C	C

faculty.

63. The principal helps the faculty develop a sense of pride and loyalty by showing his personal thoughts about the school.

64. Thinking about your knowledge and skills in the domain of "Organizational Environment: Human Resources and Community Relations" (Questions 28 – 63) prior to your participation in the Aspiring Leaders Program, to what extent do you feel your knowledge and skills in this domain have improved as a result of participating in the Aspiring Leaders Program?

- \cap To a great extent
- To some extent
- To a limited
- Not at all
- ∩ Don't know

Previous Page Next Page 43%

This online survey is powered by WebSurveyor.

SECTION III. EDUCATIONAL PROGRAM: INSTRUCTIONAL LEADERSHIP

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
65. The principal is about the general goals and objectives of curricular areas.	C	C	C	C	C
66. The principal is about the varied instructional strategies the faculty might appropriately use.	C	C	C	C	(
67. The principal possesses instructional observation skills that provide the basis for accurate assessment of the teaching process in the classroom.	r	C	C	C	C
68. The principal actively and regularly participates in the observations and assessment of classroom instruction, including teaching strategies and student learning.	C	C	r	C	r
69. The principal uses effective techniques for helping ineffective the faculty.	C	C	C	C	C
70. The principal maintains an awareness and knowledge of recent research about the learning process.	۲	C	C	C	C
71. When criticizing poor practices, the principal provides suggestions for improvement.	C	C	C	C	C
72. The principal is committed to instructional improvement.	((C	C	C

73. The principal promotes the development of educational goals and objectives that reflect societal needs and trends.

Previous Page Next Page 48%

This online survey is powered by WebSurveyor.

A second of the second	
unit for the second sec	

SECTION III. EDUCATIONAL PROGRAM: INSTRUCTIONAL LEADERSHIP (continued)

Please indicate the extent to which your participation in the Aspiring Leaders Program contributed to your knowledge and skills in each of the areas of school leadership included in the following statements:

	To a great extent	To some extent	To a limited extent	Not at all	Don't know
74. The principal promotes diagnosing student learning needs and applying appropriate instruction to meet those needs.	C	C	C	C	C
75. The principal administers a school-wide curricular program based upon identification of content goals and objectives and the monitoring of student achievement toward those goals and objectives.	r	C	C	C	C
76. The principal participates in instructional improvement activities, such as program and curriculum planning and monitoring of student learning outcomes.	C	C	۲	C	C
77. The principal uses objective data, such as test scores, to make changes in curriculum and staffing.	C	C	C	C	C
78. The principal has a systematic process for program review and change.	C	C	C	C	r
79. The principal encourages articulation of the curricular program.	C	C	C	C	C

80. Thinking about your knowledge and skills in this domain of "Educational Program: Instructional Leadership" (Questions 65 – 79), to what extent do you feel your knowledge and skills in this domain have improved as a result of

participating in the Aspiring Leader Program?

- \cap To a great extent
- To some extent
- To a limited
- ∩ Not at all∩ Don't know

Previous Page	Next Page	Market 5-7
rievious rage	HEALTOGE	52%

This online survey is powered by WebSurveyor.

SECTION IV. DEMOGRAPHICS	
Your position at the time of progran	n participation (please check all that apply):
☐ Teacher	
☐ Resource Teacher	
☐ Athletic Director	
☐ Media Specialist	
☐ Guidance Counselor	
☐ Principal	
☐ Vice Principal	
☐ Other (please	
If you selected other, please speci	ify:
<u> </u>	
Your current position (please check	all that apply):
☐ Teacher	
☐ Resource Teacher	
☐ Athletic Director	
☐ Media Specialist	
Guidance Counselor	
☐ Principal	
☐ Vice Principal	
☐ Other (please	
If you selected other, please spec	ifv:
Trybu selected seller, preuso spas	
The state of the s	and the second s
Your gender:	
⊂ Male	
← Female	
Your age:	
○ 35 or under	

€ 36-45
← 46-55
○ 56 or over
Race/ethnic classification (check one):
C African American
C Asian
Caucasian
○ Hispanic
○ Native American
○ Other (please
If you selected other, please specify:
In which level of education do you currently work?
☐ Elementary
☐ Middle School
☐ High School
Are you currently applying or will you be applying for a principal position in the near future (within five years)?
CYes
○ No
Will you be applying for another type of instructional leadership position in the near future (within five years)?
C Yes
○ No
How did participation in the program affect your thoughts about applying for future leadership positions?
○ More likely to○ Less likely to apply○ Unchanged
What motivated you to pursue the Charles County Leadership Program?
☐ Encouragement from a mentor ☐ Enhance job opportunities as a principal

extstyle ext
☐ Broaden knowledge base
☐ Broaden range of influence
☐ Opportunity to use leadership skills
☐ Prestige and status
☐ Greater professional freedom
☐ Ability to perform required tasks and functions of principal
☐ Increased responsibility
☐ Effect change on a greater scale
☐ Other (please specify)
If you selected other, please specify: Thank you for taking the time to complete the survey. Please hit the "Submit Survey"
outton below.
Cub-out Cur ov
Previous Page Submit Survey 100%

This online survey is powered by WebSurveyor.

Appendix B

ISLLC Standards

INTERSTATE SCHOOL LEADERS LICENSURE CONSORTIUM Standards for School Leaders (ISTTC)

STANDARD #1

A school administrator is an educational leader who promotes the success of all students by facilitating the development, articulation, implementation, and stewardship of a vision of learning that is shared and supported by the school community.

STANDARD #2

A school administrator is an educational leader who promotes the success of all students by advocating, nurturing, and sustaining a school culture and instructional program conducive to student learning and staff professional growth.

STANDARD #3

A school administrator is an educational leader who promotes the success of all students by ensuring management of the organization, operations, and resources for a safe, efficient, and effective learning environment.

STANDARD #4

A school administrator is an educational leader who promotes the success of all students by collaborating with families and community members, responding to diverse community interests and needs, and mobilizing community resources.

STANDARD #5

A school administrator is an educational leader who promotes the success of all students by acting with integrity, fairness, and in an ethical manner.

STANDARD #6

A school administrator is an educational leader who promotes the success of all students by understanding, responding to, and influencing the larger political, social, economic, legal, and cultural context.

available from: Council of Chief State School Officers One Massachusetts Avenue, NW

Washington, DC (202) 336-7016 FAX-(202) 408-8072

Appendix C

Request to do Research and Interview Protocols

110 Leeds Way La Plata, MD 20646 August 20, 2004

Dear Mr. Richmond:

I am asking for your permission to request the participation of program participants and designers in the Charles County Public Schools Leadership Institutes over the past three years in a dissertation study I am conducting. I am a doctoral candidate in the Department of Education Policy and Leadership at the University of Maryland. The focus of my research is to examine how a local school system designed and implemented a principal preparation program to address a shortage of qualified principal candidates.

Current literature indicates the nation will face an unprecedented shortage of qualified principal candidates in the next few years. While many systems have responded to this issue by creating principal preparation programs that they design, implement, and control, there is limited research about how effective these programs have been. This study will allow for a greater understanding of how and why a system decides to create such programs to meet system needs in leadership development. The Audit of Principal Preparation Survey, developed by Jerry Valentine and Michael L. Bowman, has been used in numerous studies and will be used for this study. I will follow-up the survey by interviewing program design members and participants.

I am requesting that all ninety-nine (99) participants of the first three cohorts of the Charles County Leadership Institute, as well as key members of the design team and principals who have helped implement this program, be a part of this study. Each participant will receive a copy of the survey and a stamped, addressed envelope for returning the survey. After tabulating results, data for specific groups and individuals will be treated anonymously. Results of the study will be available to those requesting them.

A copy of the survey and its cover letter are enclosed for your review. Your permission to request participation from Charles County Leadership Institute designers and participants would be greatly appreciated. In a few days, I will call you for a response. If you have any questions regarding the study or the survey, please call me at 301-753-1656 (home) or at 301-645-2601 (work).

Sincerely,

Heath E. Morrison

Enclosures

September 8, 2004

Dear Colleague:

I am writing to invite you to participate in a study of the Charles County Aspiring Leaders Program. The study, which is being conducted as part of my doctoral program, will examine the effectiveness of the Aspiring Leaders Program. Study findings will inform ongoing efforts to strengthen the program to ensure that it meets the needs of the Charles County Public Schools for highly-skilled leaders in every school in the district.

You have been chosen to be part of this study because of your involvement and participation in the Charles County Leadership Institute over its four year development. You will be asked to respond to a survey based on an instrument developed by Jerry W. Valentine and Michael L. Bowman. Data collection for the study includes an online survey of all participants in the first three annual cycles of the Aspiring Leaders Program. The survey asks you to rate the contributions of the program to the development of your knowledge and skills in areas related to effective school leadership. The survey also asks about your background and experience. Participation in the survey should take approximately twenty minutes. While some individuals participated in the program several years ago, all respondents will provide valuable information for this study to improve the program in Charles County.

The study design and instruments have been reviewed and approved by the Charles County Public Schools Department of Research and Assessment. Additionally, Mr. Richmond, the Superintendent, and Mr. Cunningham, the Associate Superintendent, have endorsed the survey and value this research as a way to improve the program.

Your responses are confidential. All survey data will be maintained in secure files and will be accessible only to me and members of the dissertation committee. Reports and other communications related to the study will not identify respondents by name, nor will they identify any schools. Study results will be available in a summary report, which will be transmitted to the Charles County Public Schools.

If you are willing to complete the survey, please click on the following link and follow directions which appear at the beginning of the survey.

If you have any questions or concerns about this study, you may contact me by calling 301-645-2601 or you may send me an e-mail athmorrison@ccboe.com. You may also contact Dr. Carol Parham, chairperson of my committee, by directly calling the university at 410-405-3590.

Thank you very much for your participation.

Heath E. Morrison

110 Leeds Way La Plata MD 20646 September 1, 2004

Dear Participant:

As a doctoral student at University of Maryland, I am currently involved in the dissertation segment of my program. I have designed a quantitative and qualitative research study. The purpose of my study is to examine how a school system designs and implements a principal preparation program to address issues of a qualified principal candidate shortage. The means of collecting data will be interviews that will be audiotaped and a survey.

You have been chosen to be a part of this study because of your participation in the Charles County Leadership Institute.

All responses will be kept confidential, and the participants will not be identified by name. Participants will be referenced according to their positions. Only the members of my dissertation committee and I will have access to the records of information obtained directly from the interviews and survey. The benefit of participating in this study will be to help in the development of research in the area of principal preparation programs and to help in the continued improvement of the system's leadership development efforts.

If you have any questions or concerns about this study, you may contact me by calling 301-645-2601 or you may send me an e-mail athmorrison@ccboe.com. You may also contact Dr. Carol Parham, chairperson of my committee, by directly calling the university at 410-405-3590.

The results of the study will be provided in the form of an executive summary and made available to the institution and the participants upon request.

Signing below signifies that you will allow this researcher to complete the study at your institution.

Thank you for your consideration in this matter.

Sincerely,

Heath E. Morrison	
Signature and Position at Institution	Date

Name and Address of Institution

110 Leeds Way La Plata MD 20646 September 1, 2004

Dear Participant:

As a doctoral student at University of Maryland, I am currently involved in the dissertation segment of my program. I have designed a quantitative and qualitative research study. The purpose of my study is to examine how a school system designs and implements a principal preparation program to address issues of a qualified principal candidate shortage.

You have been chosen to be a part of this study because of your involvement and participation in the Charles County Leadership Institute over its four year development. You will be asked to respond to a survey instrument developed from Jerry W. Valentine and Michael L. Bowman's "Audit of Principal Effectiveness." This survey appears as a link on this cover letter. To respond, simply click on the link. Participation in the survey should take approximately twenty minutes. While some individuals participated in the program several years ago, all respondents will provide valuable information for this study to improve the program in Charles County.

All responses will be kept confidential, and the participants will not be identified by name. Participants will be referenced according to their positions. Only the members of my dissertation committee and I will have access to the records of information obtained directly from the interviews and survey. The benefit of participating in this study will be to help in the development of research in the area of principal preparation programs and to help in the continued improvement of the system's leadership development efforts. Mr. Richmond and Mr. Cunningham have endorsed this survey and value this research as a way to improve the program.

If you have any questions or concerns about this study, you may contact me by calling 301-645-2601 or you may send me an e-mail athmorrison@ccboe.com. You may also contact Dr. Carol Parham, chairperson of my committee, by directly calling the university at 410-405-3590.

The results of the study will be provided in the form of an executive summary and made available to the institution and the participants upon request.

Signing below signifies that you will allow this researcher to complete the study at your institution. Thank you for your consideration in this matter.

Sincerely,

Heath E. Morrison			
Signature and Position at Institution	Date		

INFORMED CONSENT FORM Survey

I,	_, hereby consent to my
participation in the following doctoral dissertation re	search project:

A Critical Evaluation of a School System's Efforts to Develop and Implement a "Grow Your Own" Principal Preparation Program
Heath Morrison, Researcher
Department of Educational Leadership Policy
University of Maryland, College Park

The purpose of the research is to evaluate one school system's efforts to develop and implement a program that will provide a continuing supply of highly qualified principals to fill anticipated needs. The research will require participants to complete an electronic survey "Evaluation of the Charles County Aspiring Leaders Program," modified from a survey originally designed by Jerry W. Valentine and Michael L. Bowman. It contains eighty (80) items, uses a five-point Likert scale, and will take approximately 20 minutes to complete. This survey also includes a section with demographic questions. The following is a representative question appearing on the survey.

EXAMPLE: Please indicate the extent to which your participation in the Charles County Aspiring Leaders Program contributed to your knowledge and skills in each of the following areas:				
	To a	To	To a	Not
Don't	great	some	limited	at
know	ovtont	ovtont	extent	all
The principal encourages changes in school programs that lead to a better school for	extent	extent	extent	an
the students				

Anonymity, except to the researcher, will be protected by a coding system. Subjects may decline to answer any of the questions and will not be penalized in any way.

Participants will be asked to describe and reflect on their experiences through design or participation in the Charles County Aspiring Leaders Program. It is not anticipated that participation in this research will cause harm or professional risk to the participants. Participation in the project involves minimal risks or involves risk that is

equivalent to the level of risk associated with ordinary daily living. Anonymity and confidentiality will be honored and protected throughout the research and reporting procedures. All data will be secured in Heath Morrison's office at Thomas Stone High School and access will be restricted to the researcher and the dissertation committee. All data will be destroyed one year following the completion of this dissertation.

It is hoped that the information and insights provided with this study will help school system personnel make improvements to the Aspiring Leaders Program. It is also hoped that the information will help contribute to the expanding literature on principal preparation programs.

Questions about this study should be directed to Heath Morrison, Researcher, at 301-645-2601 or (e-mail) hmorrison@ccboe.com. If you have questions about your rights as a research subject or wish to report a research-related injury, please contact: Institutional Review Board Office, University of Maryland, College Park, Maryland, 20742; (e-mail) irb@deans.umd.edu: (telephone) 301-405-4212.

My signature below indicates that I have read the above and that I agree to participate in this research study.

I understand that participation is voluntary, and that remuneration will not be provided. I also understand that I am free to withdraw consent and to discontinue participation in the project at any time, without prejudice.

Signature	Date
Print Name	Telephone
Address	
E-mail Address	

September 7, 2004

Dear Colleague:

In the next day or so, you will receive an e-mail from me inviting you to participate in a survey. This survey asks you to reflect on your participation in the Charles County Aspiring Leaders Program. While I realize some people receiving the survey participated in the program several years ago or only did the summer program, all responses will be extremely valuable in this research.

While the primary reason for this survey is for my dissertation research project, it will also be extremely valuable in helping assess the leadership program in Charles County and contribute to making the program stronger. Mr. Richmond and Mr. Cunningham both support this research.

The University of Maryland requires all people involved in a study to sign an "Informed Consent Form," which I am enclosing with this letter. Please take a few moments to review this form and then return it to me at Thomas Stone High School via the school system pony.

Again, I hope you will agree to participate in this survey. You should receive it in your e-mail in the next day or two, and it will take between fifteen and twenty minutes to complete. Over the first three years of the program, there were approximately 100 participants, so each voice is critical to providing assessment of this program. Please contact me with any questions you may have. I very much appreciate your support in this endeavor.

Sincerely,

Heath Morrison

Thomas Stone High School 3785 Leonardtown Road Waldorf, MD 20601 301-645-2601 hmorrison@ccboe.com

enc.

Initials	Date
----------	------

INFORMED CONSENT FORM Focus Groups

I,		hereby	consent to	my j	participation
in	the following doctoral dissertation research project:				

A Critical Evaluation of a School System's Efforts to Develop and Implement a "Grow Your Own" Principal Preparation Program

Heath Morrison, Researcher

Department of Educational Leadership Policy

University of Maryland, College Park

The purpose of the research is to evaluate one school system's efforts to develop and implement a program that will provide a continuing supply of highly qualified principals to fill anticipated needs. The research will require program participants, participants who have become principals, and members of the design committee to participate in focus group interviews of approximately one (1) hour in length. These interviews will be scheduled at a time and location convenient to the participants. Interviews will be tape-recorded, and transcripts will be professionally transcribed. The interviews will be conducted by Dr. James Dudley, Professor Emeritus at University of Maryland. Examples of the questions asked during the focus group interviews are listed below:

EXAMPLE:

- 1. What perceptions about the principalship did participants have before completing the program?
- 2. What perceptions about the principalship did participants have after completing the program?

Subjects may decline to answer any of the questions and will not be penalized in any way.

Participants will be asked to describe and reflect on their experiences through design or participation in the Aspiring Leadership Program. It is not anticipated that participation in this research will cause harm or professional risk to the participants. Participation in the project involves minimal risks or involves risk that is equivalent to the level of risk associated with ordinary daily living. Confidentiality will be honored and protected throughout the research and reporting procedures. All data will be secured at Heath Morrison's office at Thomas Stone High School, and access will be restricted to the researcher and the dissertation committee. All data will be destroyed one year after the completion of this dissertation.

It is hoped that the information and insights provided w system personnel make improvements to the Aspiring I hoped that the information will help contribute to the expreparation programs.	Principals Program. It is also					
Questions about this study should be directed to Heath Morrison, Researcher, at 301-645-2601 or (e-mail) hmorrison@ccboe.com . If you have questions about your rights as a research subject or wish to report a research-related injury, please contact: Institutional Review Board Office, University of Maryland, College Park, Maryland, 20742; (e-mail) irb@deans.umd.edu : (telephone) 301-405-4212.						
My signature below indicates that I have read the above and that I agree to participate in this research study. I understand that participation is voluntary, and that remuneration will not be provided. I also understand that I am free to withdraw consent and to discontinue participation in the project at any time, without prejudice.						
Signature	Date					
Print Name	Telephone					
Address						
E-mail Address	_					

Initials____ Date____

UNIVERSITY OF MARYLAND, COLLEGE PARK HUMAN SUBJECTS REVIEW COMMITTEE APPLICATION FORM Addendum

1. Abstract:

The research focus of this study is the development of a critical evaluation of one school system's efforts to develop and implement a "grow your own" program that will provide a continuing supply of highly qualified principals to fill anticipated needs. The study serves two purposes:

- 1. To determine the extent to which this system's efforts to design and implement a program that supplements university preparation have succeeded in meeting program goals (product), efficient use of system resources (input), and appropriate selection of program activities (process)?
- 2. To determine what lessons can be learned from these efforts that might be beneficial to this system as it attempts to improve the program, and what can be learned that might prove useful to other systems coping with the need to address principal shortages?

Research activities will include a survey of program participants and focus group interviews with select participants as well as members of the program design team. The survey poses no potential for harm, and subjects will be notified of their right not to respond. Respondents will be provided full anonymity, and confidentiality of all data will be provided. Interviews will be conducted by the project faculty advisor, and interviewees will be provided an opportunity to review what was recorded to ensure it was properly transcribed.

2 Subject Selection:

- a. The subjects for the survey instrument will be the seventy-eight program participants from the first three cohorts of the principal preparation program being studied. These seventy-eight individuals include vice principals, administrative assistants, and teachers. It is expected almost all participants will be willing to participate in this study as it will provide data for program improvement in the school system being studied. The subjects who will participate in focus group interviews will be members of the program design committee (approximately five individuals), program participants (approximately five individuals), and program participants who have attained principalships (approximately four individuals). It is expected they will be willing to participate to assist in program improvement.
- b. Subjects will be selected because of their participation in the principal preparation program or their assistance in designing the program.
- c. Because this study is a formative evaluation of one school system's principal preparation program, only subjects who helped create it or have participated in it will be able to accurately determine its effectiveness.

3. Procedures:

The seventy-eight program participants will be asked to complete a survey developed in 1986 by Jerry W. Valentine and Michael L. Bowman. A copy of this proposed instrument will be attached to the application. The instrument, Audit of Principal Effectiveness, contains eighty

questions and uses a Likert-type scale ranging from 1 to 9. I am working with Policy Studies Associates in Washington, DC to provide subjects the opportunity to respond to the survey on-line. This should not only require less time and increased response rate, but also assist in ensuring confidentiality of all participants. An initial request for participation will be sent to all subjects. Subjects will then be asked to sign the Informed Consent Form, agreeing to participate in the survey. Next, the survey will be e-mailed to all participants who agree to be part of the study. They can be e-mailed back to an individual at the Charles County Public Schools Board of Education.

There will be focus group interviews with members of the program design team, select program participants, and program participants who have attained principalships. A letter requesting their participation in this study and the Informed Consent Form will be sent to these individuals. Focus group interviews will be arranged at a time and locations convenient to participants and conducted by Dr. James Dudley, Project Faculty Advisor. Proposed questions are attached to this application.

4. Risks and Benefits:

There are no risks to subjects in this study.

5. Confidentiality:

To protect the anonymity of the respondents, questionnaires will be e-mailed along with an electronic tracking card. A log will be kept of the individuals to whom the questionnaires are e-mailed and addresses and mailing dates will be noted. A follow-up e-mail and questionnaire will be sent to potential respondents when a reply is not received. The e-mail containing the survey will be sent by Kathy Perriello, an Instructional Specialist for Charles County Public Schools, and returned to her. Data will be sent to a protected file using a program called WebSurveyer. Once surveys have been collected and analyzed, the file will be destroyed.

Dr. Dudley will audio record all focus group interviews. These recordings will be transcribed and sent to participants to review for accuracy. Once all research is done for this study, these documents will also be destroyed. Only the Student Investigator, the Project Faculty Advisor, and members of the dissertation committee will have access to all recorded data. It will be stored in a secured office in the home of the Student Investigator.

6. Information and Consent Forms:

Subjects will be informed about the nature, purposes, and intent of this research study. They will be clearly informed of the reasons why they are being asked to participate in this research. A copy of the invitation letter and the consent form will be included with this application.

7. Conflict of Interest:

There is no conflict of interest for program participants in this study.

8. HIPAA Compliance:

"PHI" does not apply to this study because there is no use of protected health information.

June 3, 2004

TO: Emily Cole Bayer, Ph.D.

FROM: Heath Morrison, Principal

RE: Research Request

I am providing you all of the requested paperwork for conducting research in Charles County Public Schools. Please let me know if you need any other information. Both Mr. Richmond and Mr. Cunningham are aware and supportive of my dissertation work.

Thank you for your help.

INDEPENDENT RESEARCH REQUEST FORM

RESEARCHER INFORMATION:

List the name(s) and title(s) of the author(s) of this study:

Heath E. Morrison, Principal of Thomas Stone High School

Telephone Number:

301-753-1656 (home); 301-645-2601 (work)

Identify the institution/organization sponsoring the study: (Name, Address)

University of Maryland, EDPL Dept., College Park, Maryland

OBJECTIVES:

State the objectives of your study:

This research study examines one school system's effort to address the principal quality/quantity shortage through a "grow your own" program of leadership development. The study serves two purposes:

- 1. To what extent has this system's efforts to design and implement a program that supplements university preparation succeeded in meeting program goals (product), efficient use of system resources (input), and appropriate select of program activities (process)?
- 2. What can be learned from these efforts that might be beneficial to this system as it attempts to improve the program, and what can be learned that might prove useful to other systems coping with the need to address principal shortages?

Cite your research question(s):

- 1. What external and internal forces impacting the principalship combined to influence school authorities in the creation of their principal preparation program? (context)
- 2. Which program elements were created specifically to address those forces impacting the principals? (context)
- 3. What resources (financial, facilities, human) were made available in the design and implementation of the principal preparation program? (input)
- 4. What were the essential elements and structure of the designated program, and to what extent did they reflect current research about effective principal preparation programs and standards of licensure (ie. ISLLC)? What was the process for recruiting participants into the program? (process)
- 5. What perceptions did program participants have about the principalship before and after completing the first phase of the program? To what extent did these perceptions match the goals and expectations of the program design committee? (product)

6. To what extent did participants who have attained leadership positions report that the program has helped prepare them for the responsibilities of their positions? To what extend do annual evaluations of newly appointed principals reflect program goals? (product)

State study rationale (Why is this study important?):

The significance of this study is to further the research on a local school system's attempt to create its own principal preparation program. According to the State Department survey, over 2/3 of the principal preparation programs in the state the study was conducted were conducted in the last five years. At this time, there is limited research about these types of programs so this study will provide some insight into the challenges of creating and implementing them.

This study will seek to identify how a school system planned, developed, and implemented a principal preparation program. School districts with programs aligned with the ISLLC Standards should not only provide leaders who meet state licensing standards, but should be able to meet the ever increasing demands for accountability of principals as instructional leaders. Using the CIPP model to evaluate this school system's efforts to create a "grow your own" principal preparation program will provide feedback, suggestions, and recommendations to the school system for program improvement.

How will data from this study benefit the school(s) studied or CCPS?

I believe data collected from this study will help in the continued improvement of the Leadership Institutes offered by the Charles County Public School System. The literature review for this dissertation has already provided numerous "best practices," ideas, and concepts that have been implemented in the program. The survey of program participants will provide meaningful feedback to assess how effectively the program has aligned to its original goals.

DATA COLLECTION:

Identify the CCPS school(s) that will be involved in this study:

There will be no particular schools involved. I would like to survey the seventy-eight (78) participants of the Charles County Public Schools Leadership Institute from the past three (3) years and several of the decision makers for the program.

How much time overall will you need to collect all the data for your study? (Indicate days per school):

Several months

Suggest multiple dates on which you will be available to come to each school:

Not applicable

Describe the content of the instrument(s) that will be used:

I will combine qualitative and quantitative measures for this study. Surveys and interview questions will be used. The survey instrument will be a revision of The Audit of Principal Effectiveness created by Jerry W. Valentine and Michael L. Bowman. A copy of the proposed instrument will be attached this application.

Describe how the instrument(s) will be administered:

The study will be mailed to each of the seventy-eight participants of the Charles County Leadership Institute. Interviews will be conducted face-to-face at the interviewees' convenience.

State the range of time that similar populations needed to complete the instrument(s):

The survey should last about twenty minutes.

Describe special accommodations, if any, for special populations (e.g. Special Education/ESOL/etc.):

Not applicable

State requirements for staff/teacher participation in terms of time and level of effort, if any:

Staff taking the survey will be asked to take approximately twenty minutes to answer the instrument. Face-to-face interviews should last approximately half-an-hour.

Describe your efforts to reduce impact on student instructional time:

I will try to conduct most of my research during the summer, so as not to have any impact on instructional time.

SELECTION:

Identify target population (number of students or classrooms, grades, age, etc.):

The target population for this survey will be the seventy-eight participants of the Charles County Leadership Institute over the past three years. Members of the design team and key system leaders will be targeted for interviews.

Indicate selection/sampling procedures to be used:

Because the N is so small, I will be trying to get survey results from all seventy-eight program participants.

Describe intended data analysis procedures:

Interviews will be transcribed and respondents will be able to review the transcripts, making corrections or additions if necessary. This process will add validity to the study by allowing participants to verify their own words and ensure their thoughts were captured correctly. These interviews will be categorized through

context analysis "drawing up a list of coded categories and 'cutting and pasting' each segment of transcribed data into one of these categories" (Greenhalgh, 1997, p.5). From the interviews and records reviewed, the researcher should be able to recognize elements referenced frequently. Significant patterns and clusters should emerge from which to draw conclusions.

The survey instrument will be analyzed through quantitative procedures. The data will be analyzed using the means and standard deviation. Descriptive statistics, including t-tests for the independent sample, will be utilized. The Chi-square test for independence and the analysis of variance will be used to identify areas where significant differences existed between how program participants and designers viewed the program.

DISSEMINATION:

Where will the data and/or report be published?

The data will be used for my dissertation work published through the University of Maryland.

Describe your efforts to ensure confidentiality:

The survey that will be developed for this study will request demographic information, current job status, as well as career aspirations. This survey will be mailed out to the seventy-eight program participants from the first three cohorts, principals who helped provide input into the program and served as facilitators and presenters, and members of the design team. To protect the anonymity of the respondents, questionnaires will be mailed along with a tracking card. A log will be kept of the individuals to whom the questionnaires are mailed and addresses and mailing dates will be noted. A follow-up letter and questionnaire will be sent to potential respondents when a reply is not received.

Describe your efforts to ensure that all data, analysis, and final report are shared with R&A, school principal(s), staff, and other affected parties:

As chairperson of the Charles County Leadership Committee, I will share and disseminate relevant information to the committee to assist in program improvement. I will provide any data, research, or results from the study as requested by staff from the CCPS.

PERSONAL INTERVIEWS (Superintendent)

1. What were the forces that led to the formation of a committee to address leadership issues in the county? [context]

Probes: What were the internal forces (issues of quantity and

quality of principal candidates)?

What were the external forces (issues of traditional

preparation/university preparation)?

2. As the committee began to suggest the formation of a principal preparation program, what were your perceptions on potential benefits and any problems with its implementation? [context]

Probes: What were short-term benefits and problems?

What were long-term benefits and problems?

3. What parts of the program addressed specific needs clearly with principal preparation in the school system? [context]

Probes: What knowledge did you have regarding principal

preparation programs at that time? What knowledge have you gained?

4. How did this program align with other system initiatives and the system budget? [input]

Probes: How did this program fit into the Five Year Plan?

What were the major funding sources?

5. Were there any limitations to resources provided for this program? [input]

Probes: Were there any program elements not included because of

budgetary issues?

How was the issue of internships addressed within

budgetary limitations?

6. How were major system leaders kept current and allowed to provide input during the program development and implementation? [process]

Probes: How were Central Office leaders involved in the process?

How were principals in the school system involved in the

process?

7. What were system leaders' expectations for the program? What was done to share these expectations for the program with participants? [process]

Probes: What did you hope this program would achieve?

What was told to program participants upon entry into the

program?

8. What perceptions do system leaders have about the success of the program, specifically, is it meeting system goals? Do you have any information or feelings about whether participants share your views? [product]

How would you rate the overall effectiveness of this Probes:

program?

Has any data been collected to justify your opinion? From your experience, how have program participants

reacted to the program?

9. What perceptions about the principalship do you think program participants had both before and after the program? [product]

What were the major attitudes and thoughts about the **Probes:**

principalship that program participants shared before and after the program?

10. What was the overall assessment of the program's positive and negative outcomes? [product]

Was a formal evaluation of the program ever conducted? Probes: What other data is available to assess the overall success of

the program?

11. How do program design team members and key system leaders think program success should be sustained and needed improvement implemented? [product]

What elements of the program do you feel are most **Probes:**

What elements of the program might need to be changed or

altered?

12. Do program leaders think that providing this program has helped develop a stronger, more qualified pool of leaders in the system? [product]

Has this program assisted in developing a larger pool of **Probes:**

candidates for vice principalships/principals? Has this program enhanced the overall quality and preparedness for vice principal and principal candidates?

PERSONAL INTERVIEWS (Associate Superintendent)

1. What were the forces (internal and external) that led to the formation of a committee to address leadership issues in the county? [context]

Probes: What were the internal forces (issues of quantity and

quality of principal candidates)?

What were the external forces (issues of traditional

preparation/university preparation)?

2. As the committee began to suggest the formation of a principal preparation program, what were your perceptions on potential benefits and any problems with its implementation? [context]

Probes: What were short-term benefits and problems?

What were long-term benefits and problems?

3. What parts of the program addressed specific needs clearly with principal preparation in the school system? [context]

Probes: What knowledge did you have regarding principal

preparation programs at that time? What knowledge have you gained?

4. How did this program align with other system initiatives and the system budget? [input]

Probes: How did this program fit into the Five Year Plan?

What were the major funding sources?

5. Were there any limitations to resources provided for this program? [input]

Probes: Were there any program elements not included because of

budgetary issues?

How was the issue of internships addressed within

budgetary limitations?

6. How were major system leaders kept current and allowed to provide input during the program development and implementation? [process]

Probes: How were Central Office leaders involved in the process?
How were principals in the school system involved in the

process?

7. Who was the target audience for participation in the program, and how were they selected? [input]

Probes: Were participants specifically selected?

How were participants encouraged to participate?

8. What were system leaders' expectations for the program? What was done to share these expectations for the program with participants? [process]

Probes: What did you hope this program would achieve?

What was told to program participants upon entry into the program?

9. What perceptions about the principalship do you think program participants had both before and after the program? [product]

Probes: How would you rate the overall effectiveness of this

program?

Has any data been collected to justify your opinion? From your experience, how have program participants

reacted to the program?

10. What perceptions do system leaders have about the success of the program, specifically, is it meeting system goals? Do you have any information or feelings about whether participants share your views? [product]

Probes: What were the major attitudes and thoughts about the

principalship that program participants shared before and

after the program?

11. How do program design team members and key system leaders think program success should be sustained and needed improvement implemented? [product] What have been some of the major changes in the program? [product]

Probes: What elements of the program do you feel are most

effective?

What elements of the program might need to be changed or

altered?

12. Do program leaders think that providing this program has helped develop a stronger, more qualified pool of leaders in the system? [product]

Probes: Has this program assisted in developing a larger pool of

candidates for vice principalships/principals?
Has this program enhanced the overall quality and

preparedness for vice principal and principal candidates?

PERSONAL INTERVIEWS (Assistant Superintendent of Instruction)

1. What were the forces (internal and external) that led to the formation of a committee to address leadership issues in the county? [context]

Probes: What were the internal forces (issues of quantity and

quality of principal candidates)?

What were the external forces (issues of traditional

preparation/university preparation)?

2. What parts of the program addressed specific needs clearly with principal preparation in the school system? [context]

Probes: What knowledge did you have regarding principal

preparation programs at that time? What knowledge have you gained?

3. What parts of the program addressed specific needs clearly with principal preparation in the school system? [context]

Probes: What knowledge did you have regarding principal

preparation programs at that time? What knowledge have you gained?

4. How did this program align with other system initiatives and the system budget? [input]

Probes: How did this program fit into the Five Year Plan?

What were the major funding sources?

5. How were major system leaders kept current and allowed to provide input during the program development and implementation? [process]

Probes: How were Central Office leaders involved in the process?

How were principals in the school system involved in the

process?

6. What were system leaders' expectations for the program? What was done to share these expectations for the program with participants? [process]

Probes: What did you hope this program would achieve?

What was told to program participants upon entry into the

program?

7. What perceptions do system leaders have about the success of the program, specifically, is it meeting system goals? Do you have any information or feelings about whether participants share your views? [product]

Probes: How would you rate the overall effectiveness of this

program?

Has any data been collected to justify your opinion? From your experience, how have program participants

reacted to the program?

8. Do program leaders think that providing this program has helped develop a stronger, more qualified pool of leaders in the system? [product]

Probes:

Has this program assisted in developing a larger pool of candidates for vice principalships/principals? Has this program enhanced the overall quality and preparedness for vice principal and principal candidates?

PERSONAL INTERVIEWS (Staff Development Coordinator)

1. What were the forces (internal and external) that led to the formation of a committee to address leadership issues in the county? [context]

Probes: What were the internal forces (issues of quantity and

quality of principal candidates)?

What were the external forces (issues of traditional

preparation/university preparation)?

2. As the committee began to suggest the formation of a principal preparation program, what were your perceptions on potential benefits and any problems with its implementation? [context]

Probes: What were short-term benefits and problems?

What were long-term benefits and problems?

3. What parts of the program addressed specific needs clearly with principal preparation in the school system? [context]

Probes: What knowledge did you have regarding principal

preparation programs at that time? What knowledge have you gained?

4. What research on current programs on principal preparation was used to develop the program? [input]

Probes: What specific existing principal preparation programs were

examined?

Was current literature utilized in the design of the program?

5. Were there any limitations to resources provided for this program? [input]

Probes: Were there any program elements not included because of

budgetary issues?

How was the issue of internships addressed within

budgetary limitations?

6. What were the major resources offered in support of program development, and which of these resources were actively used in program implementation? [input]

Probes: What were the major financial resources?

What were the major facilities resources? What were the major human resources?

7. How did this program align with other system initiatives and the system budget? [input]

Probes: How did this program fit into the Five Year Plan?

What were the major funding sources?

8. How were major system leaders kept current and allowed to provide input during the program development and implementation? [process]

Probes: How were Central Office leaders involved in the process?

How were principals in the school system involved in the

process?

9. What perceptions about the principalship do you think program participants had both before and after the program? [product]

Probes: How would you rate the overall effectiveness of this

program?

Has any data been collected to justify your opinion? From your experience, how have program participants

reacted to the program?

10. What were the essential elements and structures of the designed program? How were they agreed upon by the design team and system leaders? [input/process]

Probes: Describe the essential elements of the program?

How specific was the program design to the school system

being studied?

How did the committee come to consensus on the program?

11. To what extent were the ISLLC Standards incorporated into the design of the program? [process]

Probes: Were any other standards besides ISLLC considered in the

formation of the program?

Describe how the ISLLC standards are reflected in the

program.

12. Who was the target audience for participation in the program, and how were they selected? [input]

Probes: Were participants specifically selected?

How were participants encouraged to participate?

PERSONAL INTERVIEWS (Focus Group Design Team)

- 1. What research on current programs on principal preparation was used to develop the program?
- 2. What were the essential elements and structures of the designed program? How were they agreed upon by the design team and system leaders?
- 3. To what extent were the ISLLC Standards incorporated into the design of the program?
- 4. What elements of the program were most effective in preparing participants for a future principalship?
- 5. How do program design team members and key system leaders think program success should be sustained and needed improvement implemented?
- 6. Do program leaders think that providing this program has helped develop a stronger, more qualified pool of leaders in the system?

PERSONAL INTERVIEWS (Focus Group Design Team)

- 1. What research on current programs on principal preparation was used to develop the program?
- 2. What were the essential elements and structures of the designed program? How were they agreed upon by the design team and system leaders?
- 3. To what extent were the ISLLC Standards incorporated into the design of the program?
- 4. What elements of the program were most effective in preparing participants for a future principalship?
- 5. How do program design team members and key system leaders think program success should be sustained and needed improvement implemented?
- 6. Do program leaders think that providing this program has helped develop a stronger, more qualified pool of leaders in the system?

PERSONAL INTERVIEWS (Focus Group Aspiring Principals Program Participants)

- 1. What perceptions about the principalship did participants have before completing the program?
- 2. What perceptions about the principalship did participants have after completing the program?
- 3. Did participation in the program meet the expectations of the participants?
- 4. What were areas of strength of the program?
- 5. What were areas of the program that needed improvement?
- 6. Did participation in the program strengthen, alter, or modify your desire to pursue leadership in the future?

PERSONAL INTERVIEWS (Focus Group Aspiring Principal Program Participants Who Attained Leadership Position)

- 1. What perceptions about the principalship did participants have before completing the program?
- 2. What perceptions about the principalship did participants have after completing the program?
- 3. Did participation in the program meet the expectations of the participants?
- 4. What were areas of strength of the program?
- 5. What were areas of the program that needed improvement?
- 6. Did participation in the program strengthen, alter, or modify your desire to pursue leadership in the future?
- 7. What elements of the program were most effective in preparing participants for a future principalship?
- 8. To what extent did participants who have attained leadership positions indicate the program helped prepare them for their new responsibilities?

Appendix D

Second Request Letter

Dear Colleague:

Recently, I sent you an e-mail inviting you to participate in an online survey, which is being administered as part of an evaluation of the Charles County Aspiring Leaders Program. According to my records, I have not received your survey. If you are willing to complete the survey, click on the following link:

http://webmailer.com/C.dll/Jya70B5CpL6C8B51KD9sU4J.htm

and follow directions which appear at the beginning of the survey. Please complete the survey as soon as possible, but no later than September 24, 2004. It may not be possible to process survey results received after that date.

It you have any questions, or concerns about this evaluation, you may contact me by telephone at 301-645-2601 from 7:00 AM to 3:00 PM, or by e-mail at hmorrison@ccboe .com.

Thank you very much for your participation.

Heath E. Morrison

Appendix E

Leadership Grant

Grant Application Role of the Principal as Instructional Leader

Local School System: Charles County Public School

Superintendent: Mr. James Richmond

Contact Person for Grant: Carolyn Graham

Phone: 301 934-7332 E-mail: cgraham@ccboe.com

I. Project Summary (Describe what you wish to do -- 15 points)

Charles County Public Schools (CCPS) and the Superintendent are committed to providing to new and aspiring principals, high quality professional development which links directly to the school systems focus on increased Academic Achievement, Career Readiness and Personal Responsibility. To accomplish this goal CCPS's will establish a leadership training program. The CCPS Leadership Training Program will complete a design that builds upon the experiences of new and aspiring principals. The purpose will be to further develop their knowledge, skills, and competencies to prepare them for the challenges of becoming a principal in Charles County.

One major challenge of the contemporary principalship is to create professional learning communities. Principals must become facilitators of learning for students, faculty and community. A challenge for those responsible for providing the leadership development is to better have the program participants understand how they can support those individuals willing to assume the responsibility of the principalship.

A Leadership Development Committee, established in 1999, began to review the role of the principal for the 21st Century. Using the Interstate School Leaders Licensure Consortium (ISLLC) National Standards, the system identified skills which are necessary for a school administrator under each of the six standards. (See Appendix A)

The "Role of the Principal as Instructional Leader Grant proposal" will focus on the following Leadership Development Objectives:

- To develop curriculum/modules for post graduate training credits.
- To provide relevant learning experiences for new and aspiring principals.

- To assist new and aspiring principals in becoming the educational leader of their school.
- To develop and provide a networking system of experienced principals and central office administrators.
- To provide activities and experiences which will enhance the knowledge, skills, and abilities necessary to become a high performing principal.

I. Project Discussion (briefly respond to the following questions):

1. How is your project unique (15 points)?

One unique component of this project is the identification of knowledge, skills, and abilities of teacher leaders within our schools. CCPS believes that the future principals of our system will come from the initial phase of this leadership development. Through this process, potential leaders will be nurtured and trained to maximize their leadership skills.

This model allows for any school system to utilize the national standards to identify the needs in their own county.

2. What do you hope to learn from your project that will have value for dissemination across the state.

The goal is to develop a comprehensive model, partnering with the University of Notre Dame, that will be institutionalized in Charles County and become a model for leadership development which can be replicated in any school system.

3. How will your project help with the looming shortage of administrators?

The Maryland State Department of Education Task Force on the Principalship made three major recommendations: a) The State Department of Education, and all 24 local school systems, must "clear the plate" of extraneous responsibilities assigned principals to ensure they have time to fulfill their primary role as instructional leaders/facilitators; b) Comprehensive programs must be developed for the identification of principal candidates and school systems must provide professional development programs for new and current principals; and c) Principal salary and compensation packages must be adjusted to better reflect the responsibilities, accountability, and stress of the principalship.

In support of the above "recommendation b" of the Task Force report, CCPS's is developing a model which identifies knowledge, skills, and abilities for all instructional leadership in the schools. By defining these roles, educators will know what is expected and will be provided a clear plan for professional growth. We

believe people who know what is expected and are provided a clear plan for professional growth are willing to commit to a future vision.

3. How will you evaluate the effectiveness of your project?

A model of evaluation will be developed including these components:

- · Each participant will identify a Leadership Learning Plan through selfassessment. They will verify their self-assessment through personal reflections, observations, and documentation of leadership activities. These experiences will be documented in the Principal's Portfolio.
- A Professional Development (experienced principals and central office staff) Team will work with each candidate to monitor their continued learning schedule. These teams will provide support, assistance, and opportunities to reflect on experiences critical to each one's growth and development. The Professional Development Team will be responsible for evaluating the candidate's progress.
- Semi-annual evaluations will be completed by experienced principals and central office administrators. The final evaluation will include the results from their instructional modules and their performance based experiences consistent with Charles County Board of Education policies.

This process will be piloted and evaluated by the candidates and the Professional Development Team.

4. Who will be responsible for preparing your final report on this project?

The Leadership Development Advisory Committee

5. How will you spend the \$5,000 that you receive?

Salaries and Wages:	<u>\$500</u>
Contracted Services:	\$3,000
Supplies and Materials:	\$1,500
Equipment:	000
Other:	000
Total:	\$5,000

Total Possible Points 100

Total Points Awarded: (to be completed by MSDE staff)

/6/13/60 Date

Appendix F

School System Leadership Themes and ISLLC Standards

ASPIRING PRINCIPAL

		Haran San San San San San San San San San S	
STANDARD #6	A school administrator is an educational leader who promotes the success of all admines by understanding, responding to, and influencing the larger political, social, economic, legal, and cultural context.	Change and Confled Resolution as I spoke to Social, Legal, Po- libral Systems D-boky Development D-boxy Development C-boxy Development D-boxy Development C-box of Legal and Poffical Issues C-box of Legal and Poffical Issues Social, Legal and Poffical Issues Social, Legal and Fortext Soluce C-box (Detrot Community, State, and Federal Governance)	O Legal krasuse O School District Policies
STANDARD #5	A school administrator is an educational leader who promotes the success of all students by acting with integrity, fairness, and in an ethical manner.	Modele Integrity, Feitness in an Elizabi Marrier and Elizabi Marrier and Purpose of Schooling Marcey of Education	D Professional Cade of Ethica
STANDARD #4	A school administrator is an educational leader who promotes the success of all students by collaborating with families and community members, responding to deverse community interests and needs, and mobilizing community resources.	Dealing with the Media-School and Dathick Perspective Debits Relations/Promoting Schools Community involvement and Participation Demonstripe Community Resources	School Promotions 1 Knowledge of Resources Avail- able
STANDARD #3	A school administrator is an educational leader who promotes the success of all students by ensuring menagement of the organization, operations, and resources for a safe, efficient, and effective learning environment.	Organizational Development Cortisis Intervention Human Reacousce Management and Development And Development Cortisis Intervent and Decision Making Corticol District Operational Pro- Corticol District Operational Pro- Corticol District Operational Pro- Corticol Selety and Security Corticol Selety and Secur	School Cperations School Cherations Speec Unitration Speec Unitration Speec Unitration Administrator Procedures Substitute Technology Emergency Preparation Pan Managing of School Frances Transportation Substitute Revorte
STANDARD #2 ST	A school administrator is an educational leader who promotes the success of all students by advocating, muturing, and sustaining a school culture and instructional program conducive to student learning and staff professional growth.	Curriculum K-12 Assessment and Evatuation of Program Adut Learned Envisorment School Culture Chroniga/School Reform Chroniga/School Refor	School Improvement Matter Scheduling County Wide Assessment Program of Studies for District Constitution of Studies for District Constitution of Studies for District Constitution of Instruction Testing Programs Student Arbatation/Transition
STANDARD #1	A school administrator is an educational teader who promotes to success of all students by facilitating the development; articulation, implementation, and stewardship of a vision of learning that is shared and supported by the school community.	Strategic Planning Systems Theory Concerna Building Communication Skills Negotiation Skills Negotiation Skills Negotiation Skills Negotiation Skills Negotiation Skills Negotiation Skills Newtran Reletions Skills	Share and promote the vision of learning for the action. Writing appropriately to a variety of audiences
	INTERSTATE SCHOOL LEADERS LICENSURE CONSORTIUM (ISLLC) ANTIONAL STANDARDS	Азяния Рамстра. ТиэтиоЭ маярояч	Aspiring Vice Principal Program Content

	ENVIRC	ENVIRONMENTS THAT ARE CONDUCTVE TO THE LIEVELOPMENT OF SCHOOL LEADERS	JOUCIVE TO THE LIEVEL	OPMENT OF SCHOOL LE	AOEKS	
DEVELOPMENTAL FOCUS	RECOGNITION	COLLEGIALITY AUTORONY PARTICIPATION OPEN COMMANMENT POSSITIVE ENVIRONMENT	AUTOROMY	PARTICIPATION	Орен сомичисьтон	POSITIVE ENVIRONMENT
Assist in learning new Recognized for the profee. Collaborate on instructional Encauraged to take the in- involved actively in making Send and receive communit. Experience a possible city for control of the process and student and student-related markers. Italive to make improve decisions and have logic and student-related markers. Italive to make improve decisions and have informed about trikine leadership. Intellegation of the control	Recognized for the professional rotes they take and the contributions they make.	Assest in learning new Recognized for the profes. Collaborate on instructional Encouraged to take the inf- Involved addresty in making the make sines and student-related maries. Itahie to make improve- decisions and have into recouraged to help the the contributions they make. Discuss attractive, share ments and to be invovative. Important mafferts. Instructing of others.	Excaraged to take the initialive to make improve- ments and to be innovative.	the in- prove- decisions and have hour or ative, important matters.	Swid and receive communitations in drading in open ways and leef informed about hisppenings in their school.	Experience a possible climate and effective administrative leadership.

REFERENCES

- Anderson, L. (1995, August). High school principals and school reform: Lessons learned from a standard story of Project RE:LEARNING. *Educational Administration Quarterly*, 31, 420-429.
- Anderson, M. (2000). *Principals: How to train, recruit, select, induct, and evaluate leaders for America's schools*. Washington, DC.: Office of Educational Research and Improvement. (ERIC Document Reproduction Service No. ED 320223).
- Argetsinger, A. (2000, June 21). Vacancies predicted in principal's office: Maryland schools told to boost pay, training. *The Washington Post*. Washington, DC.
- Barker, S. L. (1997, November). Is your successor in your schoolhouse? Finding principal candidates. *NASSP Bulletin*, *81*(592), 85-91.
- Blackman, M. C., & Fenwick, L. T. (2000, March 29). Looking for leaders in times of change. *Education Week*, 19(29), 68. Bethesda, Maryland: Editorial Projects in Education.
- Boeckmann, M. E., & Dickinson, G. B. (2001, Spring). Leadership: Values and performance. *Education*. Retrieved June 13, 2002 from ProQuest database.
- Bottoms, G., & O'Neill, K. (2001, April). *Preparing a new breed of school principals: It's time for action*. Atlanta, GA: Southern Regional Education Board.
- Bowling, G., Principal, McDonough High School, Charles County, Maryland. Personal interview, November, 2001.
- Bradshaw, L. K., Perreault, G., McDowelle, J. O., & Bell, E. W. (1997, November). Evaluating the results of innovative practices in educational leadership programs. Charleston, SC: Southern Regional Council on Educational Administration. (ERIC Document Reproduction Service No. ED 416200).
- Brockett, D. (1999, September). Qualified principals are getting harder to find. *School Board News*. NSBA National Affiliate.
- Brouillette, L. (1997, December). Revising an innovative high school: What happens when the principal leaves? *Educational Administration Quarterly*, 33(546).
- Browne-Ferrigno, T. (2001, November). *Preparing school leaders: Case study summary and implications*. Paper presented at the Annual Meeting of the

- University Council for Educational Administration, Cincinnati, OH, Nov. 2-4, 2001. (ERIC Document Reproduction Service No. ED 465226).
- Copland, M. A. (2001, March). The myth of the superprincipal. *Phi Delta Kappan*, _____. Bloomington, IN: Phi Delta Kappa, Inc.
- Croghan, J. H., & Lake, D. G. (1984, November). Strategies considered for improving the principalship. *Viewpoints*. Research Triangle Park, NC: Southeastern Regional Council for Educational Improvement.
- Cunningham, R., Associate Superintendent, Charles County Public Schools, Maryland. Personal interview, November, 2001.
- Daresh, J. C. (1990, May). Formation: The missing ingredient in administrator preparation. *NASSP Bulletin*, 74(526), 1. Reston, VA: National Association of Secondary School Principals.
- Daresh, J. C., & Playko, M. A. (1992, April). What do beginning leaders need? Aspiring and practicing principals' perceptions of critical skills for novice administrators. Paper presented at the annual meeting of the American Educational Research Association, San Antonio, Texas, April 20-24, 1992. (ERIC Document Reproduction Service No. ED 344312).
- Daresh, J. C. (1997). Improving principal preparation: A review of common strategies. *NASSP Bulletin*, 18(January), 3-8.
- DeAngelis, K., & Rossi, R. (1997, February). Programs for aspiring principals: Who participates? Washington, DC: American Institutes for Research in the Behavioral Sciences. (ERIC Document Reproduction Service No. ED 405644).
- DeBevoise, W. (1982). The principal's role: How do we reconcile expectations with reality? *R & D Perspectives*. Eugene, OR: Oregon University, Eugene Center for Educational Policy and Management.
- DiPaola, M., & Tschannen-Moran, M. (2003, March). The principalship at a crossroads: A study of the conditions and concerns of principals. *NASSP Bulletin*, 87(634), 43-65. Reston, VA: National Association of Secondary School Principals.
- Donaldson, G. A. (2001, October). The lose-lose leadership hunt. *Education Week*. Bethesda, Maryland: Editorial Projects in Education. Retrieved March 18, 2004 from http://www.edweek.org/ew/ew/printstory.cfm?slug=05donaldson.h21

- Donaldson, G. (2001). Cultivating leadership in schools: Connecting people, purpose, and practice. New York: Teachers College Press.
- Dosdall, E., & Diemert, N. (2001, Winter). Working from a higher rung. *Journal of Staff Development*, 22(1), 26-29. Oxford, Ohio: National Staff Development Council.
- Fenwick, L. T., & Pierce, M. C. (2001, March). The principal shortage: Crisis or opportunity? *Principal*, 19-23. Cambridge, MA: The Principals Center at Harvard Graduate School of Education.
- Ferrandino, V. (2001, February). Challenges for 21st-century elementary school principals. *Phi Delta Kappan*, 82(6), 440-448. Bloomington, IN: Phi Delta Kappa, Inc.
- Ferrandino, V., & Tirozzi, G. (2000). *The principal, keystone of a high-achieving school: Attracting and keeping the leaders we need.* Arlington, VA: Educational Research Service, National Association of Elementary School Principals, and National Association of Secondary School Principals.
- Fitzpatrick, J. L., Sanders, J. R., & Worthen, B. R. (2004). *Program Evaluation: Alternative Approaches and Practical Guidelines*. Boston: Pearson Education, Inc.
- Foran, J., Director of Secondary Principals, Maryland State Department of Education, Baltimore, Maryland. Personal interview, November, 2001.
- Gall, M. D., Borg, W. R., & Gall, J. P. (1996). *Educational research: An introduction*. New York: Longman Publishers.
- Goodwin, L. D. (1975). A proposed model for educational accountability. Doctoral dissertation, University of Maryland. *Dissertation Abstracts International* UMI No.75-28,746.
- Greenhalgh, T., & Taylor, R. *How to read a paper: Papers that go beyond numbers* (Qualitative Research). London, England: BMJ Publishing Group. Retrieved April 22, 2004 from http://bmj.bmjjournals.com/cgi/content/full/315/7110/740.
- Griffith, J. (1999, April). The school leadership/school climate relation: Identification of school configurations associated with change in principals. *Educational Administration Quarterly*, 35(2), 267-291. University Council for Educational Administration.
- Groff, F. (2001, October). Who will lead? The principal shortage. *State Legislatures*. Denver, CO: National Conference of State Legislatures.

- Henry, T. (2000, August 22). A principal crisis in education: Districts train their own to fill a growing need for top administrators. *USA Today*. Arlington, VA: USA Today. Retrieved March 16, 2004 from Proquest database.
- Hessel, K., & Holloway, J. (2002). A framework for school leaders: Linking the ISLLC standards to practice. Princeton, NJ: Educational Testing Service.
- Hettel, K. (2001, November). Executive Director of Human Resources of Charles County, Maryland. *Personal interview*.
- Hoyle, J. R. (2002). Superintendents for Texas school districts: Solving the crisis in executive leadership. Fort Worth, TX: Sid W. Richardson Foundation.
- In hiring principals, schools face another struggle. (1999, September). *Techniques: Making Education and Career Connections*, 74(6), 8.
- Interstate School Leaders Licensure Consortium. (1996). *Standards for School Leaders*. Washington, DC: The Council of Chief State School Officers. Retrieved June 13, 2002 from http://www.ccsso.org/isllc1.html
- Institute for Educational Leadership. (2000, October). *Leadership for student learning: Reinventing the principalship*. Washington, DC: Institute for Educational Leadership.
- Johnson, D. (2001, March 29). Maryland board backs ideas to recruit principals. *The Washington Post*.
- Jordan, D. (1994, July). *The supply and demand trends of public school principals and administrators in southwestern Louisiana: 1993-1997.* [Research Report]. Retrieved on April 22, 2004 from http://ericae.net/eric.db/ED375525.htm.
- K-12 Principals' Guide to No Child Left Behind. (2003). Arlington, VA: Educational Research Services, National Association of Elementary School Principals, National Association of Secondary School Principals.
- Keedy, J. L., & Grandy, J. (1999, April). *Trends in GRE scores in education administration: Implications for principal preparation programs*. Paper Presented at the Annual Meeting of the American Educational Research Association, Montreal, Quebec, Canada, April 19-23, 1999. (ERIC Document Reproduction Service No. ED436828).
- Keil, V. L., & Czerniak, C. M. (2003, October 1). The urban principal leadership program. *Connections*, 5. Retrieved Nov. 1, 2003, from http://www.nassp.org/publications/connections/keil_czerniak.cfm

- Keller, B. (1999, March 10). Principals' salaries rising, but slowly, survey finds. *Education Week*. Retrieved Nov. 28, 2001 from http://www.edweek.com/ew/ew_printstory.cfm?slug=26prin.h18.
- Keller, B. (2000, May 3). Building on experience. *Education Week*. Retrieved Nov. 28, 2001 from http://www.edweek.com/ew/ew_printstory.cfm?slug=34prins.h19.
- Kerrins, J. (2001, May/June). Take this job and fill it. *Leadership*, 30(5), 20.
- Lashway, L. (1998, Fall). Standards for administrators. *Research Roundup*, *15*(1). Reston, VA: National Association of Secondary School Principals. Retrieved July 2, 2002 from http://eric.uoregon.edu/publications/roundup/F98.html.
- Lashway, L. (2002a, February). Transforming principal preparation. Eugene, OR: Clearinghouse on Educational Management. (ERIC Document Reproduction Service No. ED473360).
- Lashway, L. (2002b, Spring). Rethinking the principalship. *Research Roundu*, 18(3). Reston, VA: National Association of Secondary School Principals. Retrieved July 2, 2002 from http://eric.uoregon.edu/publications/roundup/Spring_2002.html.
- Lauder, A. (2000, September). The new look in principal preparation programs. *NASSP Bulletin*, 84(617), 23-28. Reston, VA: National Association of Secondary School Principals.
- Lezotte, L. W. (1991). Correlates of effective schools: The first and second generation. Okemos, MI: Effective Schools Products, Ltd.
- Lord, M. (2000, August 5). Hello? Anybody there? School districts scramble to fill principals' chairs. *U.S. News and World Report*, 133(5), 38. Washington, DC.
- Lumsden, L. (1992-93, Winter). The new face of principal preparation. *Research Roundup*. Reston, VA: National Association of Elementary School Principals. (ERIC Document Reproduction Service No. ED352730).
- Lyons, J. (1992, May). Competencies of beginning principals are they prepared for the role? (ERIC Document Reproduction Service No. ED346589).
- Maryland State Department of Education. (1999, December 8). *State superintendent names principal task force*. News Release. Baltimore, Maryland.

- Maryland State Department of Education. (2000, August 30). News Release. Baltimore, Maryland.
- Maryland State Department of Education. (2000, September). News Release. Baltimore, Maryland.
- Maryland Task Force on the Principalship. (2000). Recommendations for redefining the role of the principal; recruiting, retaining, and rewarding principals; and improving their preparation and development. Baltimore, MD: Maryland State Board of Education.
- Mercado, L. P. (2002). The secondary principalship academy: A critical ethnography of the Houston Independent School District and the University of Houston's innovative principal preparation program. (Doctoral dissertation, College of Education, University of Houston). Ann Arbor, MI: Proquest Information and Learning Company. (UMI No.3067773).
- Merriam, S. (1998). *Qualitative research and case study applications in education*. San Francisco, CA: Jossey-Bass.
- Milwaukee schools look within for quality principals. (2001, November). Newsleader. Reston, Virginia: National Association of Secondary School Principals.
- Murphy, F. M. (2001, January). Growing leaders. *American School Board Journal*, 30-32. National School Boards Association.
- Murphy, J. (2001, September). *Re-culturing the profession of educational leadership: new blueprints*. Paper commissioned for the Meeting of the National Commission for the Advancement of Educational Leadership Preparation, Racine, WI, Sept. 19-21, 2001. (ERIC Document Reproduction Service No. ED464380).
- Nakamura, D., & Samuels, C. A. (2000a, June 25). In school, changes at the top: Area faces shortage of principals. *The Washington Post*.
- Nakamura, D. (2000b, July 8). Montgomery money attracts Prince George's principals. *The Washington Post*.
- Nakamura, D. (2000c, August 31). In principle, ready for the job: School administrator shortage creates an age of opportunity. *The Washington Post*.
- National Association for Schools of Excellence and Northwest Regional Educational Lab. (1999). *Leading America's schools: The critical role of the principal*. Portland, OR: Northeast Educational Lab.

- National Association of State Boards of Education. (1999, October). *Principals of change: What principals need to lead schools of excellence*. Alexandria, VA: National Association of State Boards of Education. (ERIC Document Reproduction Service No. ED467060).
- Newsom, J. (2001, November). Leadership 101: A shortage of qualified administrators has sparked a boom in principal-preparation program. *American School Board Journal*, 30-33.
- Orozco, L. C., & Oliver, R. E. (2001, July 1). Orange County commentary: A lack of principals. *The Los Angeles Times*.
- Pellicer, L. O., & others. (1981, December). *The evolution and training of school principals*. Washington, DC: Fund for the Improvement of Postsecondary Education. (ERIC Document Reproduction Service No. ED218707).
- Qualitative social science research methodology. (2000). Retrieved April, 2004 from http://faculty.ncwc.edu/toconnor/308/308lect09.htm.
- Quinn, T. (2002, October). Succession planning. *Principal Leadership*, 24-28. Alexandria, VA: Association for Supervision and Curriculum Development.
- Richardson, M. D., et al. (1989, November). *Changing principal preparation programs: A five-phase model*. Paper Presented at the Annual Meeting of the Southern Regional Council on Educational Administration, Columbia, SC, Nov. 11-14, 1989. (ERIC Document Reproduction Service No. ED313790).
- Rossi, R., & Daugherty, S. (1996, April). Where do minority principals work? Collected Works, Serials (022), Statistical Data (110). Washington, DC: National Center for Education Statistics. (American Institutes for Research)
- Scriven, M. (1992). *Hard-won lessons in program evaluation*. Retrieved March 19, 2004, from http://eval.cgu.edu/lectures/hard-won.htm.
- Soranno, K., & Schmitt, D. M. (1991, October). *The sources of administrative training in the preparation of building level administrators*. Paper presented at the annual meeting of the University Council for Educational Administration, Baltimore, MD, Oct. 25-27, 1991. (ERIC Documents Reproduction Service No. ED343198).
- Stufflebeam, D. L. (1971a, February). *Critique of the report of the Phi Delta Kappa study committee on evaluation*. Paper presented at the annual meeting of the American Educational Research Association, New York, New York. (ERIC Document Reproduction Service No. ED056074).

- Stufflebeam, D. L. (1971b, February). *The relevance of the CIPP evaluation model* for educational accountability. Paper presented at the annual meeting of the American Association of School Administrators, Atlantic City, NJ, February 24, 1971. (ERIC Document Reproduction Service No. ED062385).
- Stufflebeam, D. L. (1976, February). *Educational evaluation: Some questions and answers*. Paper presented at the annual meeting of the American Association of School Administrators. Atlantic City, New Jersey, February, 1976. (ERIC Document Reproduction Service No. ED128425).
- Stufflebeam, D. L. (1981, October). *A review of progress in educational evaluation*. Paper presented at the Joint Annual Meetings of the Evaluation Network and the Evaluation Research Society. Austin, TX, October, 1981. (ERIC Document Reproduction Service No. ED216031).
- Stufflebeam, D. L. (2002, June). *CIPP evaluation model checklist*. Retrieved October 31, 2003 from http://www.wmich.edu/evalctr/checklists/cippchecklist.htm.
- Thomson, S. D. (1988, May). The principalship: Ingredients of programs to prepare effective leaders. *NASSP Bulletin*, 39-46. Reston, VA: National Association of Secondary School Principals.
- Thomson, S. D., Hill, T. S., & Conny, B. M. (1993). *Principals for our changing schools: The knowledge and skill base*. Fairfax, VA: National Policy Board for Educational Administration.
- Tirozzi, G. N. (2001, February). The Artistry of Leadership: The Evolving Role of the Secondary School Principal. *Phi Delta Kappan*, 82(6), 434-440.
- Tirozzi, G. N., &. Ferrandino, V. L. (2001, January). How do you reinvent a principal? *Education Week*. Retrieved March 18, 2004 from http://www.nassp.org/news/views/rnvnt_prncpl_edwk101.htm
- Tracy, G., & Weaver, C. (2000, October). Aspiring leaders academy: Responding to the principal shortage. *NASSP Bulletin*, 75-83. Reston, VA: National Association of Secondary School Principals.
- Trail, K. (2000, October). Taking the lead: The role of the principal in school reform. *CSRD Connections*. Austin, TX: Southwest Educational Development Lab.
- Trejos, N. (2001, June 28). A challenging offer for potential principals: Charles recruiting, training leaders. *The Washington Post*.

- U. S. Department of Education. (1984, February). *Selecting principals*. Collected Works Serials (022). Washington, DC: National Institute of Education.
- Whitaker, K. (2001, May). Where are the principal candidates? *NASSP Bulletin*, 85(625), 82. Reston, VA: National Association of Secondary School Principals.
- Wise, W. (2001, November). Principal of Mattawoman middle school. Charles County, Maryland. Personal Interview.
- Yerkes, D. M., & Guaglianone, C. L. (1998, November/December). Where have all the high school administrators gone? *Thrust for Educational Leadership*, 28(2), 10-15. Fresno, California: Association of California School Administrators.